AD-A152 466 UNCLASSIFIED	SPEED-UP SEMICONDU WRIGHT-PF 14 DEC 84	TECHNIQUE JCTOR VERY ITTERSON A AFIT/GE/	S FOR C Larg FB oh S Eng/84d	OMPL EM (U) AI CHOOL -41	ENTARY R FORC OF ENG	' METAL E INST I E	OXID OF T T KEL F/G	CH Ley 10/12	17 NL	کی آ
	·									



MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A



AFIT/GE/ENG/84D-41



SPEED-UP TECHNIQUES

FOR

COMPLEMENTARY METAL OXIDE SEMICONDUCTOR

VERY LARGE SCALE INTEGRATION

THESIS

AFIT/GE/ENG/84D-41

Brian T. Kelley Captain USAF



Approved for public release; distribution unlimited.

AFIT/GE/ENG/84D-41

SPEED-UP TECHNIQUES

FOR

COMPLEMENTARY METAL OXIDE SEMICONDUCTOR

VERY LARGE SCALE INTEGRATION

THESIS

Presented to the Faculty of the school of Engineering

of the Air Force Institute of Technology

Air University

in Partial Fulfillment of the

Requirements for the Degree of

Master of Science

by Brian T. Kelley, B.S. Captain USAF Graduate Electrical Engineering

Approved for public release; distribution unlimited.

Preface

Documentation of techniques for increasing the speed of MOS circuits, evaluation of a specific technique and application of the technique to existing and newly designed CMOS/SOS circuits were performed. A 4x4 bit RAM, 2x4 decoder, two registers and read/write control logic were designed and the precharge speed-up technique was evaluated and simulated using SPICE2, to determine if it actually increased the operating speed of circuits to which it was applied. A methodology was determined for applying precharging to existing circuits through application to a previously designed portion of a CMOS/SOS ALU and to the newly designed circuit elements mentioned above. Extensive SPICE2 simulation indicated that precharging does increase circuit operating speed, but must be selectively applied.

I offer a special thanks to Major W. Sutton, my thesis advisor, for allowing me to take on this thesis project and for his overall assistance and draft reviews. My appreciation also goes to 2nd Lt. M.McConkey for his valuable assistance in understanding the SPICE2 models.

ession for 1. 161 sistribution/ Availability Codes Avail and/or Special Dist

ii

DIIC

Croe+

ECH

Table of Contents

		<u>I</u>	rage
Prefac List o List o List o Abstra	e f Fi f Ta f Sy ct	gures bles mbols	ii v xiii xvi xvi xix
I.	Int	roducton	I-1
	A.	Background	.1-1
	в.	Problem Statement	I-6
	c.	Assumptions	1-7
	D.	Approach	1-7
	E.	Sequence of Presentation	I-8
II.	CMO	S Review	II-1
III.	Ciro	cuit Delay and Timing Analysis	III-1
	A.	General Switching Circuit Timing Analysis	III-1
		1. Bipolar Transistor	III-1
		2. MOS transistor	III-9
		3. CMOS Timing Analysis	III-13
	в.	Propagation Delay Analysis	III-18
IV.	Spee	ed-Up Techniques	IV-1
	A.	Capacitor Pull-Up Circuit	IV-1
	в.	Bootstrap Pull-Up Inverter	IV-4
	с.	Precharging	IV-5
	D.	Propagation Delay Reduction	IV-19
	E.	General Remarks	IV-22
	F.	CMOS VLSI Applications	IV-23
		1. Requirements	IV-23
		2. Selection of Technique	IV-24

· . .

v.	System Desig	nv-1
	A. Introduc	tionV-1
	B. Requirem	entsv-1
	C. Circuit	SelectionV-3
	D. Design R	RulesV-3
	E. ALU Bits	lice ModificationsV-3
	F. Memory/A	LU Bitslice Circuit DesignV-7
VI.	Detailed Des	ign and AnalysisVI-1
VII.	Simulation R	esults AnalysisVII-1
	A. Evaluati	on CriteriaVII-1
	B. Test Cir	cuit SimulationVII-4
	C. Memory/A	LU Bitslice Circuit Element SimulationVII-13
	D. ALU Bits	lice SimulationVII-22
	E. Results	SummaryVII-30
VIII.	Conclusions	and RecommendationsVIII-1
Biblio	raphy	BIB-1
Append	x A: SPICE	Data of Precharge Test CircuitA-1
Append	x B: SPICE Circui	Data of Memory/ALU Bitslice t Precharged ElementsB-1
Append	x C: SPICE	Data of ALU Bitslice CircuitC-1
Append	x D: SPICE	User Guide AddendumD-1
Append	x E: Memory	ALU Bitslice Chip Design

List of Figures

-

Figure	Page
II-1	CMOS InverterII-1
II-2	Transfer CharacteristicII-3
II-3	CMOS Nand CircuitII-4
II-4	CMOS Nor CircuitII-4
II-5	Loading Capacitance versus Feature SizeII-7
III-1	Bipolar TransistorIII-1
III-2a	Bipolar Transistor as an Open SwitchIII-2
III-2b	Bipolar Transistor as a Closed SwitchIII-2
III-3	Active Region Transistor Model
III-4	Input Waveform and Response
III-5	NMOS Transistor with Depletion LoadIII-10
III-6	Turn-on and Turn-off Waveforms
III-7	Ratioless Capacitive Pull-up Inverter CircuitIII-19
IV-1	Ratioless Capacitive Pull-up Inverter CircuitIV-2
IV-2	Bootstrap Pull-Up CircuitIV-4
IV-3	Charge Pump CircuitIV-9
IV-4	Read/Write CircuitIV-10
IV-5	3-Transistor Memory SubsystemIV-12
IV-6	Carry Chain Circuit for OM2 ALUIV-13
IV-7	Domino Logic CircuitIV-14
IV-8	N-type Dynamic CMOS Logic BlockIV-15
IV-9	Precharge Generation and Control CircuitIV-17
IV-10	High-Speed Word DecoderIV-17
IV-11	Sense Amplifier CircuitIV-18

IV-12	Precharge Implementation CircuitIV-21
IV-13	64K Dynamic RAM with RepeatersIV-22
IV-14	Output Precharge Circuit SchemeIV-24
IV-15	Internal Transmission Gate Precharge SchemeIV-25
V-1	CMOS Transmission GateV-4
V-2	Internal Precharge CircuitsV-5
V-3	Output Precharge CircuitV-5
V-4	Unmodified ALU Bitslice Block DiagramV-6
V-5	Modified ALU Bitslice CircuitV-6
V-6	System Circuit DiagramV-9
v- 7	Pad Layout for ChipV-11
VI-1	Delay Analysis CircuitVI-2
VI-2	CLL Layout of Internal Precharge CircuitVI-6
VI-3	CLL Layout of Output Precharge CircuitVI-7
VI-4	ALU Bitslice Schematic DiagramVI-8
VI-5	ALU Bitslice CLL LayoutVI-8
VI-6a	Detailed Design of Modified Bitslice with Ten Nodes PrechargedVI-9
VI-6b	Detailed Design Schematic of Bitslice with Four Nodes PrechargedVI-10
VI-7	Precharged ALU Bitslice CLL LayoutVI-11
VI-8	A Register SchematicVI-13
VI-9	CLL Layout for A RegisterVI-14
VI-10	B Register SchematicVI-15
VI-11	CLL Layout for B Register,,VI-15
VI-12	Read/Write Controller Schematic Diagram,,.VI-16
VI-13	CLL Layout of Read/Write Controller,,VI-17
VI-14	ALU Input Controller Schematic,,,

vi

VI-15 VI-16	ALU Input Controller CLL Layout,VI-19 Six-Transistor Memory Cell,,VI-19
VI-17	Single Bit Memory Cell CLL LayoutVI-20
VI-18	2x4 Decoder Schematic Diagram,VI-21
VI-19	CLL Layout of 2x4 Decoder,VI-21
VI-20	Schematic of Precharged 2x4 Decoder,VI-22
VI-21	2x4 Precharged Decoder Circuit,,VI-23
VI-22	Schematic of Precharged MemoryVI-24
VI-23	CLL Layout of Precharged MemoryVI-24
VI-24	Precharged A Register SchematicVI-25
VI-25	Precharged B Register SchematicVI-25
VI-26	CLL Layout of Precharged A RegisterVI-26
VI-27	CLL Layout of precharged B RegisterVI-27
VI-28	Detailed Memory/ALU Bitslice CircuitVI-28
A-1	Basic Test Cell NodeplotA-3
A-2	SPICE Waveform Plot of Basic Test Circuit with Input 010A-5
A-3	SPICE Waveform Plot of Basic Test Circuit with Input 101
A-4	Node Plot ot Test Circuit Precharged at the Nand Output
A- 5	SPICE Waveform plot of Precharged Nand Output with Input 010A-10
A-6	SPICE Waveform Plot of Precharge Nand output with Input 101
A- 7	Node Plot of Test Circuit Precharged at First Inverter Output
A-8	SPICE Waveform Plot of Precharged First Inverter with Inputs 010A-15
A-9	SPICE Waveform Plot of Precharged First Inverter Output with Inputs 101A-17

1

.

A-10	Node Plot of Test Circuit Precharged at Second Inverter Output
A-11	SPICE Waveform Plot of Precharged SecondInverter with Input 010A-20
A-12	SPICE Waveform Plot of Precharged Second Inverter with Input 101A-22
A-13	Node Plot of Test Circuit Precharged at Nand and First Inverter OutputsA-23
A-14	SPICE Waveform Plot of Precharged Nand and First Inverter with Inputs 010A-25
A-15	SPICE Waveform Plot of Precharged Nand and First Inverter with Inputs 101A-27
A-16	Node Plot of Precharged Nand and Second Inverter OutputA-28
A- 17	SPICE Waveform Plot of Precharged Nand and Second Inverter with Inputs 010A-30
A-18	SPICE Waveform Plot of Precharged Nand and Second Inverter with Inputs 101A-32
B-1	Node Plot of Basic Decoder
B-2	SPICE Waveform Plot of Basic Decoder with Inputs 010B-6
B-3	Node Plot of Decoder with Internal Nodes prechargedB-7
B-4	SPICE Waveform Plot of Internal Node Precharged Decoder with Inputs 010B-9
B-5	SPICE Plot of Internal Node Precharged Decoder with Transmission Gate Widths Twice the Basic WidthsB-13
B-6	SPICE Plot of Internal Node Precharged Decoder with Transmission Gate Widths Three Times the Basic WidthsB-16
в - 7	Node Plot of Decoder Precharged at External NodesB-17
B-8	SPICE Plot of Decoder Outputs Using Basic Width Transmission gatesB-21
B-9	SPICE Plot of Precharged Decoder Outputs Transmission Gate Widths Doubled

B-10	SPICE Plot of Precharged Decoder Outputs Transmission Gate Widths TripledB-27
B-11	Basic B Register Node PlotB-28
B-12	SPICE Plot of Basic B Register OutputB-31
B-13	Precharged B Register Node PlotB-32
B-14	SPICE PLot of Precharged B Register Output Using Basic Transmission GatesB-36
B-15	SPICE Plot of B Register Output Using Transmission Gate Widths Twice the Basic WidthsB-39
B-16	SPICE Plot of Precharged B Register Output Using Transmission Gate Widths Three Times the Basic WidthsB-42
B-17	Basic RAM Cell Node PlotB-43
B-18	SPICE Plot of RAM Cell with Input 010B-45
B-19	SPICE Plot of RAM Cell with Input 101B-46
B-20	Precharged RAM Cell Node PlotB-47
B-21	SPICE Plot of Precharged RAM Output Using Basic Transmission Gate Widths and Input 010B-49
B-22	ST.CE Plot of Precharged RAM Cell Output Using Basic Transmission Gate Widths and Input 101B-51
B-23	SPICE Plot of Precharged RAM Cell Output with Input 010 and Transmission Gate Widths Twice the Basic WidthsB-53
B-24	SPICE Plot of Precharged RAM Cell with Input 101 and Transmission Gate Widths Twice the Basic WidthsB-55
B-25	SPICE Plot of Precharged RAM Cell Output, with Input 010 and Transmission Gate Widths TripledB-57
B-26	SPICE Plot of Precharged RAM Cell Output, with Input 101 and transmission Gate Widths TripledB-59
C-1	Basic Section 1 CLL PlotC-2

•

C-2 SPICE Plot Basic Section 1 with Input 010.....C-6 C-3C-4 C-5 CLL Plot Section 1 with Two Nodes Precharge.....C-10 C-6 Node Plot SEction 1 With Two Nodes Precharged...C-10 C-7 SPICE Plot Section 1 with Input 010 and Two Node Precharge Basic Gate Widths.....C-14 C-8SPICE PLot Section 1 Input 101 and Two Node Precharge with Basic Gate Widths.....C-17 C-9 SPICE PLot Section 1 Input 010 and Two Node Precharge with Basic Widths x2.....C-20 C-10 SPICE Plot Section 1 Input 101 Two Node Precharge with gate Widths x2.....C-23 C-11 SPICE Plot Section 1 Input 010 Two Node Precharge with Gate Widths x3.....C-26 C-12 SPICE Plot Section 1 Input 101 Two Node Precharge with gate Widths x3.....C-29 C-13 CLL Plot Section 1 Four Node Precharge......C-30 C-14 Section 1 Four Node Precharge Node Plot.....C-30 C-15 SPICE Plot Section 1 Input 010 Four Node Precharge with Basic Gate Widths.....C-34 C-16 SPICE Plot Section 1 INput 101 Four Node Precharge with Basic Widths.....C-37 C-17 SPICE Plot Section 1 Input 010 Four Node Precharge with Gate Widths x2.....C-35 C-18 SPICE Plot Section 1 Input 101 Four Node Precharge with gate Widths x2.....C-43 C-19 SPICE Plot Section 1 Input 010 Four Node Precharge with Widths x3.....C-46 C-20 SPICE Plot Section 1 Input 101 Four Node C-21 C-22 Basic Section Node Plot.....C-50

C-23	SPICE Plot Basic Section 2 Input 010C-53
C-24	SPICE Plot Basic Section 2 Input 101C-55
C-25 C-26	Section 2 One Node Precharge CLL PlotC-56 Section 2 One Node Precharge Node PlotC-56
C-27	SPICE Plot Section 2 Input 010 One Node Precharge with Basic Gate WidthsC-59
C-28	SPICE Plot Section 2 INput 101 One Node Precharge with Basic Gate WidthsC-61
C-29	SPICE Plot Section 2 Input 010 One Node Precharge with Widths x2C-63
C-30	SPICE Plot Section 2 Input 101 One Node Precharge with Widths x2C-65
C-31	SPICE Plot Section 2 Input 010 One Node Precharge with Widths x3C-67
C-32	SPICE Plot Section 2 Input 101 One Node Precharge with Widths x3C-69
C-33	Section 2 Precharge Two Node CLL PlotC-70
C-34	Section 2 Two Node Precharge Node PlotC-70
C-35	SPICE Plot Section 2 Input 010 Two Node Precharge Using Basic WidthsC-73
C-36	SPICE Plot Section 2 Input 101 Two Node Precharge Using Basic WidthsC-75
C-37	SPICE Plot Section 2 Input 010 Two Node Precharge with Widths x2C-77
C-38	SPICE PLot Section 2 Input 101 Two Node Precharge with Widths x2C-79
C-39	SPICE Plot Section 2 Input 010 Two Node Precharge with Widths x3C-81
C-40	SPICE plot Section 2 Input 101 Two Node Precharge with Widths x3C-83
C-41	Basic Section 3 CLL PlotC-84
C-42	Basic Section 3 Node PlotC-84
C-43	SPICE Plot Basic Section 3 Input 010C-87

1

xi

C-44	SPICE Plot Basic Section 3 Input 101C-89
C-45	CLL Plot Section 3 One Node PrechargeC-90
C-46	Section 3 One Node Precharge Node PlotC-90
C-47	SPICE Plot Section 3 Input 010 One Node Precharge with Basic WidthsC-93
C-48	SPICE Plot Section 3 Input 101 One Node Precharge with Basic WidthsC-95
C-49	SPICE Plot Section 3 Input 010 One Node Precharge with Widths x2C-97
C-50	SPICE Plot Section 3 Input 101 One Node Precharge with Widths x2C-99
C-51	SPICE Plot Section 3 Input 010 One Node Precharge with Widths x3C-101
C-52	SPICE Plot Section 3 Input 101 One Node Precharge with Widths x3C-103
C-53	Section 3 Four Node Precharge CLL PlotC-104
C-54	Section 3 Four Node Precharge Node PlotC-104
C-55	SPICE Plot Section 3 Input 010 Four Node Precharge with Basic WidthsC-107
C-56	SPICE Plot Section 3 Input 101 Four Node Precharge with Basic WidthsC-109
C-57	SPICE Plot Section 3 Input 010 Four Node Precharge with Widths x2C-111
C-58	SPICE Plot Section 3 Input 101 Four Node Precharge with Widths x2
C-59	SPICE Plot Section 3 Input 010 Four Node Precharge with Widths x3C-115
C-60	SPICE Plot Section 3 Input 101 Four Node Precharge with Widths x3

xii

<u>List of Tables</u>

Ċ

Table	<u>Pa</u>	ge
II-1	CMOS/SOS Versus CMOS BulkII	-6
VI-1	ALU FunctionsVI	-7
VII-1	Basic Test Circuit DelaysVI	I - 8
VII-2	Test Circuit Delays with Precharged Nand OutputVI	I-8
VII-3	Circuit Delays - Precharged First Inverter OutputVI	I-9
VII-4	Circuit Delays - Precharged Second Inverter OutputVI	I-10
VII-5	Circuit Delays - Precharged Nand and First Inverter OutputsVI	I-11
VII-6	Precharged Nand and Second Inverter OutputVI	I-12
VII-7	Total Circuit DelaysVI	I - 13
VII-8	Basic Decoder DelaysVI	1-15
VII-9	Precharged Decoder (Internal Node) DelaysVI	I-16
VII-10	Precharged Decoder (Output Nodes) DelaysVI	I - 17
VII-11	B Register DelaysVI	1-18
VII-12	Precharged B Register DelaysVI	I-18
VII-13	Basic RAM Cell DelaysVI	I-20
VII-14	Precharged RAM Cell DelaysVI	I-21
VII-15	Section 1 Simulation Input ValuesVI	I-23
VII-16	First Bitslice Section DelaysVI	I - 27
VII-17	Section 2 InputsVI	1-27
VII-18	Second Bitslice Section DelaysVI	I-28

VII-19 Bitslice Third Section Inputs......VII-29 Third Section Bitslice Delays.....VII-30 VII-20 A-1 Basic Test Cell Node Cross Reference A-2 Precharge Nand Output Only - Node Reference List.....A-8 A-3 Precharge INV1 Output Only - Node A-4Precharge INV2 Only - Node Reference A-5 Precharge Nand and INV1 Outputs Node A-6 Precharge Nand and INV2 Outputs Node Reference List.....A-28 B-1 B-2 Precharge All Four Outputs of Decoder - Node Reference List.....B-8 B-3 2x4 Decoder All Outputs Precharged Node List.....B-18 B-4 Basic B Register Node List.....B-28 B-5 Precharged B Register Node List.....B-33 B-6 Basic RAM Cell Node List.....B-43 B-7 Precharged RAM Cell Node List......B-47 C-1 Basic Bitslice Section 1 Node Reference List.....C-3 C-2 Bitslice Section 1 Precharged at Two Nodes - Node Reference List.....C-11 C-3 Bitslice First Section with Four Nodes Precharged - Node Reference List.....C-31 C-4 Bitslice Basic Section 2 Node Reference List.....C-51 C~5 Bitslice Section 2 Node Reference List - One Node Precharge.....C-57 C-6 Bitslice Precharged Section 2 Node

xiv

	Reference ListC-71
C-7	Bitslice Section 3 Node ListC-85
C-8	Bitslice Section 3 Node Reference List - One Node PrechargeC-91
C-9	Bitslice Precharged Section 3 Node Reference ListC-105

2

4

. •

<u>.</u>...

٠..

10 10 N 14

۰.

.

· . . .

.

....

List of Symbols

/

v _T N	threshold voltage for n-type transistor
v _t	threshold voltage for p-type transistor
C _{be}	base-emitter bipolar transistor capacitance
Cbc	base-collector bipolar transistor capacitance
Rb	bipolar transistor base resistance
	bipolar transistor barrier voltage
$ au_{ m d}$	time delay of bipolar transistor
ß。	large signal current gain of bipolar transistor
	corner frequency of bipolar transistor frequency response
c _D	bipolar transistor diffusion capacitance
r _{bc}	bipolar base-collector resistance
D	D factor
I _{B2}	initial base current
I _{B3}	final base current
$ au_{ extsf{s}}$	recombination time associated with excess charged stored on the base of a bipolar transistor
$C_{\mathbf{L}}$	load capacitance
т _L	depletion mode transistor
т _D	pull-down transistor
$ au_{ t L}$	time delay of depletion mode transistor
$ au_{ extsf{D}}$	time delay of pull-down transistor
β_{D}	gain of pull-down transistor
$\beta_{\tt L}$	gain of pull-up transistor
μ_n	electron mobility of n-type material

5 S S S S

· · ·

- $\mu_{\rm p}$ hole mobility of p-type material
- insulator capacitance Ci

Vas drain-source voltage of n-type transistor IdsN drain-source current of n-type transistor threshold voltage of n-type transistor

 $v_{threshold_N}$

gate-source voltage of n-type material Vas

n-type transistor constant KN

 $\epsilon_{\rm ox}$ permitivity of oxide layer

- channel width W
- channel length L
- thickness of insulating oxide Tox
- C_{eff} effective gate capacitance
 - Ιp current through the P-type transistor
 - Kp p-type transistor constant

 $v_{threshold_N}$

threshold voltage of p-type transistor

- $\alpha_{\rm N}$ normalized threshold voltage for n-type transistor
- $\alpha_{_{P}}$ normalized threshold voltage for p-type transistor
- T_F signal falltime
- signal risetime TR
- total driver delay ch
- elementary driver delay dr
- C_{T.} load capacitance
- CG gate capacitance of basic NMOS transistor
- $au_{ extsf{L}}$ load line delay
- T_i input circuit delay
- $\tau_{\rm D}$ total delay for large capacitance line drive circuits

xvii

Vss	substrate	e bias	voltage
-----	-----------	--------	---------

- W_{opt} optimum width of crossunder
 - Ct capacitance of crossunder
 - R_{XO} sheet resistance of crossunder
 - R_O output impedance of line driver
 - C_{XO} capacitance per unit area of crossunder

Abstract

Contration Program

Methods of increasing the operating speed of integrated circuits were investigated and a reference to speed-up techniques was generated. Precharging, a specific method, was applied to existing and newly designed CMOS/SOS circuit elements and evaluated. The SPICE was used for transient signal timing analysis.

Precharging was applied to a test circuit, the first bitslice of a previously designed ALU circuit and three newly designed chip circuit elements, to determine its effect on the operating speed of each circuit. Precharge configurations of each circuit were then simulated with SPICE.

The results of this thesis research indicate that precharging can be applied to both existing and newly designed circuits, that it significantly reduces low-to-high signal transition delays, if applied correctly, and, in general, it has a detremental effect on high-to-low signal transitions, and increases the associated delays. In addition, the effectiveness of precharging is dependent on the amount of current applied to the precharged nodes.

xix

I. Introduction

A. Background

1. CMOS Technology

Complementary Metal Oxide Semiconductor (CMOS) technology is characterized by low power consumption and excellent noise immunity, while offering high integration density and rapid circuit switching speeds when compared to other technologies. CMOS originated due to requirements for portable/low power systems for avionics and aerospace applications. Although CMOS has a lower switching speed than some technologies, specifically NMOS and bipolar, and thus a lower maximum frequency of operation, its advantages have made it the emerging technology for both military and civilian applications for Very Large Scale Integration (VLSI).

CMOS circuit performance can be improved as smaller geometries are used. Circuit speed will be improved and power dissipation reduced. However two basic limitations exist; optical printing capabilities, since the wavelength of light provides a lower bound on the dimension sizes used, and the fabrication processes themselves which will call not only for reduced lateral dimensions, but for oxide thicknesses less than 20 nanometers. For these thinner oxides, more "pure" manufacturing elements will be required to maintain a minimum acceptable yield during the fabrication process.

When comparing integrated circuit parameters, layout

I-1

density, speed-power product and gate delay are the primary characteristics of concern to the circuit designer. A small speed-power product can result in low power dissipation and rapid circuit operation (associated with gate delay) and higher circuit densities become increasingly important as VLSI applications and requirements become more advanced.

2. Silicon-On-Sapphire (SOS)

SOS technology uses sapphire as a substrate in place of silicon in the basic MOS fabrication process. SOS has increased in popularity since it offers the highest speed power product of any silicon technology and can compete with any other silicon technology in packing density. However, the cost of the sapphire substrate has limited its use primarily to military systems where its additional advantage of low susceptability to radiation effects is of critical importance. Optimum performance is obtained when SOS is used with CMOS devices, where power consumption is minimal and noise immunity is excellent. Speed can thus be increased since the parasitic capacitance that exists when a silicon substrate is used doesn't exist when sapphire is used. SOS processing requires only a third of the steps required for silicon substrate (bulk CMOS) and is less susceptible to mask and oxide defects.

3. Factors Influencing Circuit Operating Speed

The two factors contributing to the speed of a circuit

I-2

are: capacitance and resistance. Capacitance measures the amount of charge that must be provided to a unit of area before the voltage changes and resistance is a measure of the ease with which current can be supplied to this given area. The values of the resistance and capacitance of certain materials and circuits is dependent on both the size (length and width) of the materials used for a specific application and on the process that is used for fabrication of the circuit or chip. To increase the speed of the circuit, the resistance or capacitance (or both) must be reduced to minimize the resultant RC time constant delay. The following data reviews some of the various approaches used to affect speed changes in integrated circuits.

a. Fabrication Materials

Lower loading capacitances have been achieved using insulators such as sapphire and gallium arsenide in place of silicon for IC substrates. Use of sapphire has resulted in higher clock frequency operation in large scale integrated circuits and is believed to be based on the reduction of long interconnect capacitances in data buses and control lines. Medium scale integrated circuits using gallium arsenide substrates have shown to have overall higher switching speeds. Yuan et al [1], provided test results upon comparison of silicon to both sapphire and gallium arsenide substrates. Their studies indicated that for line widths greater than 2.5 microns, substantial capacitance reduction could be achieved by use of sapphire and gallium arsenide substrates. Propagation over interconnect lines with greater than 2.5 micron widths have shown these positive results but little data is available on line widths below this 2.5 level.

b. Fabrication and Processing

Dennard et al [2], have accumulated data on the optimization of silicon gate MOSFETs for high performance logic applications for both room temperature and liquid nitrogen temperature operating environments. The principal factor affecting their success has been the optimization of ion-implant techniques that have been able to achieve lightly doped substrates, resulting in shallower junctions for minimizing lateral diffusion under the edge of the gate and the field oxide. Strict control of doping has permitted fine adjustment of device and field region thresholds.

New developments in lithographic and etching techniques have resulted in reduced feature size. An expected increase in speed is associated with any reduction in feature size, however this has been found to hold true for only small (low density) circuits. As the size of a circuit increases due to its complexity, the density of the chip per given unit area increases. Since feature size reduction results in less cross sectional area for line widths, hence reduced drive current capability, the effect of reducing the feature size with an increase in the density of the chip, is expected to increase time delays. Device performance is no longer expected to be the primary source of delay since the individual transistors are expected to switch more quickly, however interconnection delays are expected to dominate the delay times. The distances that a signal will need to travel will be greatly increased relative to the feature sizes of the circuits themselves.

c. Circuit Design and Layout

From the previous information, it appears that improvements in the manufacturing processes are going to continue to have a significant effect on the performance of integrated circuits. However, circuit design methods can be a more flexible approach to increase the speed of circuits. The circuit designer can modify an existing design or incorporate specfic techniques into an original design as part of a routine design procedure, to attempt to overcome or at least reduce the RC time delay within the circuit.

"Bootstrapping", "charge pumping" and "precharging" are terms used among circuit designers and rarely refered to within available design literature, to refer to circuit modifications and "hybrid" type circuits used to increase a circuit's operating speed. The distinct lack of data concerning these topics in technical publications and other literature implies a reluctance by circuit designers to disclose their personal "tricks" of the trade due to profes-

I-5

sional (competitive) as well as other reasons.

In some cases, however, even though a particular method may be frequently used and refered to by both individuals and literature, no quantitative performance data in support of this particular speed-up approach is presented or is otherwise made available. As a result, the reader is left with only the author's interpretation of the success of the method and must speculate on its applicability to other circuits.

B. Problem Statement

The purpose of this thesis is to investigate and collect data on various techniques developed and implemented to increase the operating speed of electronic circuits, and to select a specific technique and demonstrate its applicability to CMOS technology for VLSI applications by determining the extent to which the circuit speed is actually affected by the speed-up technique.

At this time, the only CMOS research performed at AFIT was performed by Capt. W. Sommars, GE-83D, in his thesis [3] which developed a CMOS/SOS cell library. Since AFIT interest is primarily in SOS technology, this thesis will involve only CMOS/SOS and will make use of the existing SOS library.

Since time will not permit fabrication of the actual chip and subsequent testing, a simulation will be performed to determine the performance of the circuitry. However, the data package required for fabrication will be completed and sent to MOSIS. Additional thesis work can then be generated to test the chip.

This thesis is intended to provide the VLSI circuit designer with a consolidated source of information on the analysis of circuit signal delays and a reference to specific methods that may be used to increase the operating speed of ciruits under design. Much of the available literature refers to and uses specific speed-up techniques, but the impact of the technique on the circuit's performance is often assumed rather than specifically evaluated. This project will also provide an evaluation of a specific technique to prove or disprove its inherent worth for application.

A secondary objective of this thesis is to actively use the CMOS/SOS library previously designed by Sommars [3] and extend the AFIT knowledge base in the CMOS/SOS area.

C. Assumptions

The following asssumptions were made to limit the scope of this project:

1. The reader has an extensive understanding of the VLSI CAD tools at AFIT

2. The reader has a basic understanding of circuit theory and transient signal analysis

3. The reader is familiar with Metal Oxide

Semiconductor devices

D. Approach

The following approach was used for this thesis:

1. Evaluate transient switching as it pertains to bipolar, MOS and CMOS circuits.

2. Evaluate the existing CMOS/SOS library and become familiar with SPICE2, a circuit simulator

3. Perform a literature search for "speed-up" circuits and techniques

4. Apply precharging to the first bitslice of Sommars' CMOS/SOS ALU

5. Run a SPICE simulation on the modified bitslice and the original ALU bitslice and compare the timing results

6. Design a Memory/ALU Bitslice circuit modified ALU and apply precharging to it where possible

7. Design additional library cells as required to support the Memory/ALU Bitslice circuit design

8. Run a SPICE simulation on the Memory/ALU Bitslice circuit with and without precharging applied and compare results

9. Design chip layout

E. Sequence of Presentation

Chapter II provides a review of basic CMOS circuits.

I-8

Chapter III continues with a review of transient timing analysis for bipolar, MOS and CMOS circuits. A review of bipolar transistor timing is included to facilitate the timing review, since most practicing engineers are more familiar with timing analysis as it applies to the bipolar transistor than to the MOS transistor. Chapter IV provides data on speed-up techniques and circuits that can be applied during the design process and presents the criteria for selection of precharging as the technique evaluated during this thesis. Chapter V presents the system designs for the modified CMOS/SOS ALU Bitslice and for a Memory/ALU Bitslice circuit that demonstrates an application of the modified ALU bitslice and the use of precharging in a larger circuit. Chapter VI presents the detailed design of the circuits described in Chapter V. Chapter VII presents a comparison of the results of a SPICE2 circuit simulation on a test circuit, and the modified and unmodified ALU bitslice circuits. Chapter VIII provides this authors conclusions and recommendations for future projects.

II. <u>CMOS</u> <u>Review</u>

The following sections provide a review of basic CMOS circuits.

A. The basic CMOS element is a series combination of an N-type and a P-type metal oxide semiconductor transistor as shown below in Figure II-1.



Figure II-1. CMOS inverter.

This circuit performs a logical inversion function. A metal oxide semiconductor transistor is a device in which the current in a channel between two electrodes, referred to as the source and drain respectively, is modulated by the voltage applied to a third electrode, referred to as the gate, which is physically separated from the source and drain by an insulating material. In the N-type transistor, the majority carriers are electrons so that positive voltage on the gate will increase the conductivity of the channel. For gate voltages less than an established minimum positive threshold voltage, the channel is cut-off and no appreciable drain current flows. The operation of the P- type transistor is similar to the N-type however, the majority carriers are holes and the operating voltages of the P-type device are negative. The threshold voltages for either type transistor can be controlled by the impurity doping level of the channel during fabrication and is therefore a design parameter of the transistor itself.

The basic operation of the circuit of Figure II-1 is as follows: if V_{in} is at 0 volts, the P-type transistor is biased on by a negative gate-to-drain voltage and the N-type transistor is cut-off by a zero gate-to-drain voltage. The output of the inverter then approaches $+v_{dd}$ (the supply voltage), since the N-type transistor causes an "open circuit" to exist between the output and ground when it is cut-off. If the input voltage is increased, the current in the N-type transistor rises and the current in the P-type transistor falls. The input voltage at which the two transistors are in the constant current region at the same time marks the active operating region of the inverter circuit. As the input voltage increases beyond this point, the output voltage falls to zero. Therefore for an input voltage at or near the value of the supply voltage, the Ntype transistor will be turned on, bringing the output of the inverter to ground potential and the P-type transistor will be cut-off.

For either output state, the total current flow results from the equivalent of a cut-off transistor since either the N-type or the P-type transistor will be in cut-off

11-2

permitting near zero current flow. This is the principal advantage of using CMOS technology, since low current flow results in low power dissipation. Current flows in the CMOS circuit only during high to low or low to high transitions as the circuit passes through the active operating region of the transistors.

In order for one or the other of these transistors to be turned off when the input is either high or zero, the threshold voltages for the N-type and P-type transistors must be positive and negative respectively so both transistors must be enhancement mode types for correct circuit operation. A transfer characteristic for the basic CMOS inverter is provided in Figure II-2. Where v_{T_p} and v_{T_N} are the respective threshold voltages of the p-N type and n-type materials.



Figure II-2. Transfer characteristic for CMOS transistor inverter [4].

II-3

The single CMOS inverter may be extended to provide NAND and NOR applications as illustrated in Figure II-3 and Figure II-4.



Figure II-3. CMOS NAND circuit.



Figure II-4. CMOS NOR circuit.

In the NAND configuration, when all inputs are high, all Ntype transistors in series are in a conducting mode while all P-type transistors are in cut-off. The line to ground is closed and the output is low. In the NOR configuration,
if any one of the inputs is high, the output will be shorted to ground.

B. Four characteristics must be considered if a specific transistor technology is to be used for design:

- 1. Speed of operation
- 2. Power dissipation/consumption
- 3. Ease of designability
- 4. Ease of integration

CMOS dissipates very little power as already stated, but is not the fastest transistor technology. However, measures can be taken to speed up its operation. This is the primary effort of this thesis.

Ease of designability and integration then become primary driving factors in the choice of a specific technology. Conflicting opinions exist over the integration capability of CMOS circuits since proponents of other technologies contend that twice as much area is required for CMOS applications since both a P-type and N-type transistor is required to implement even the most simple inversion function, as compared to NMOS technology for instance, where a smaller transistor/load transistor pair can be used for the same function. However, Krambeck et al [5] have developed a new approach in CMOS requiring the use of only a single P-type device with mulitple N-type devices in their domino logic cirucit. This approach has considerably reduced the requirement for chip area and threatens the design community's universal acceptance of the factor of two increase in chip real estate when using CMOS technology. From the data available, CMOS is on an equal ranking with other technologies as far as ease and flexibility of design are concerned. CMOS offers standard cells for design that leave the problem of compactness and density of design as a function of the state-of-art as in the case of Very Large Scale Integration (VLSI) applications, and as a function of If CMOS is selected as a transistor technology to be used for design, two basic types exist, CMOS bulk and CMOS Silicon-On-Sapphire (SOS). Table I-1 provides comparison data on the advantages and disadvantges of bulk versus SOS [1].

TABLE I-1

CMOS/SOS versus CMOS Bulk

Advantages of CMOS/SOS	Disadvantages of CMOS/SOS
Lower parasitic junction capacitance Lower power dissipation Lower propagation delay Higher packing density Higher packing density Higher punch through voltage Reduced junction shorts Radiation hardened Reduced "soft" errors Reduced Latch-up Good scaling Tolerant to Parameter variations Simplicity of design Mixed technologies	Parasitic edge leakage Back channel leakage Floating substrate - ChargeStorageeffect Reduced mobility - N channel Sapphire variability Thermal conductivity Wafer Breakage - Thermal shock Commercial viability
	1

II-6

CMOS/SOS is characterized by better ease of design, better control of threshold and punchthrough voltages, radiation hardness and according to the available data has a factor of two better power dissipation and propagation delay than its bulk counterpart. The fabrication process however causes many of the problems associated with CMOS/SOS. SOS is subject to variable starting material characteristics and planar substrates during processing, floating substrate charging effects and "back-channel" leakage [1].

There is cause for concern among proponents of SOS technology when CMOS is considered for VLSI application. Present predictions and preliminary test results indicate that as feature sizes decrease below the present state-ofthe-art (2 micron) that loading capacitances are expected to increase in SOS to the point where bulk and SOS become comparable in performance. Figure II-5 provides preliminary data on loading capacitance as a function of feature size for both SOS and bulk [1].





II-7

The applications of CMOS in VLSI design is an emerging field, however whether SOS or bulk technology is the most advantageous appears to be a matter of preference at this point in time since present efforts by industry seem to be making bulk and SOS technologies more comparable and the field is a competitive one.

III. Circuit Delay and Timing Analysis

The following sections provide a review of timing analysis with respect to bipolar, MOS and CMOS circuits.

For the reader who is not familiar with timing analysis for basic circuits, this chapter begins with a review of bipolar transistor timing (the most easily understood analysis) and concludes by expanding the basic bipolar concepts to MOS and CMOS devices switching theory and reviews circuit propagation delay.

There are two types of delays that must be evaluated for signals in circuits, switching delays that are characteristic of individual transistors, and propagation delays associated with the transfer of a signal over an interconnection line.

The section on NMOS circuits is intended to introduce the reader to the switching delays associated with MOS devices. A more detailed review is provided in the section on CMOS devices.

1. Bipolar Transistor

The transistor can be used to closely approximate a switch. A simple bipolar transistor is used here for the purpose of simplifying the analysis and is illustrated in Figure III-1.



Figure III-1. Bipolar transistor [6].

Figure III-2 represents the two ideal cases of transistor operation, as an ideal open switch and as an ideal closed switch.



Figure III-2. a. Bipolar transistor as an open switch b. Bipolar transistor as a closed switch [6].

a.

In Figure III-2a, when the transistor acts as an open switch, no current flows. When the transistor acts as a closed switch, shown in Figure III-2b, a high value of current flows through the transistor to ground. In Figure III-1, when the voltage on the base of the transistor is

zero, the transistor is cut-off and no current flows, similar to the ideal open switch case described above, and the output voltage approaches the supply voltage, Vcc. If a large enough voltage supplies enough current to the base to drive the transistor into saturation, then the transistor turns on and approximates the case of the ideal closed switch. Current flows from collector to emitter and the output voltage drops to an effective ground level. During the time that the transistor switches from cut-off to saturation, the transistor passes through its linear operating range. If high speed switching is to take place, reactive effects of the transistor in this region must be taken into consideration. Three transistor equivalent circuits are used when evaluating the switching characteristics of the transistor switch:

- a. Open switch in the cut-off region.
- b. An active region model as illustrated in Figure III-3.
- c. A closed switch in the saturation region

The open and closed switch models have already been refered to.



Figure III-3. Active region transistor model [6].

There are three factors that affect the transition switching times (from high to low or from low to high) of a transistor:

- Depletion region capacitances causing a delay time
- Diffusion capacitances and the Miller effect capacitances causing rise and fall times to exist
- 3. A storage time constant during which time excess base charge is removed before the transistor is permitted to drop from its saturation region into its active operating region

The response of a common-emitter bipolar switch to a rectangular pulse is shown in Figure III-4.



Figure III-4. Input waveform and response [6].

The waveform is composed of a delay time, a rise time, an on time, a storage time, and a fall time. The delay time is defined as the amount of time required for the waveform to rise to 10% of its peak value after the input changes. The rise time is the time taken for the waveform signal to rise from 10% to 90% of its final value. The turn-on time is the sum of the delay time and rise time. The storage time is defined as the sum of the saturation storage time and the amount of time required for the waveform to drop to 90% of its peak value after the input changes. The falltime is the time required for the waveform to drop from 90% to 10% of its peak achieved value. The turn-off time is the sum of the storage time and the fall time.

If the input to the base of the transistor is less than the required minimum turn-on voltage of the transistor, then the circuit will be cut-off since no current will flow. If terminal is driven with a negative voltage, the base Voff, for a specified period of time, then it will charge the base-emitter and base-collector depletion region capacitances C_{be} and C_{bc} respectively. If the input voltage is then changed to a positive value, the voltage of the base will not be able to change immediately and will approach the turn-on voltage of the transistor at a rate controlled by the capacitance values Cbe, Cbc and the base resistance R_b. The transistor will not turn on until the minimum turn-on voltage has been achieved on the base. The time lapse between the application of the positive input to

the base and the time when the transistor actually turns on is the passive delay time. As the base voltage changes, the voltage across the depletion-region capacitances changes in a non-linear manner. For high frequency transistors the depletion region capacitance is defined as the average value of capacitance determined from the total change in charge on the capacitance as the voltage varies from one value to another, divided by the total voltage change, and is approximated by the equation [6]:

$$C = \frac{1.5k}{(V'-V)} [(\phi - V') - (\phi - V)] [1]$$

where ϕ is the barrier voltage, k is a constant, and V and V' are two different values of voltage. The time constant of the circuit is:

$$T_{d} = (C_{BE} + C_{BC})R_{B}$$
 [2]

If the initial voltage on the base is $-V_{\text{Off}}$ and must increase to V_{On} for the transistor to turn on, then the time required to reach V_{On} can be determined from a general charging equation:

$$v(t) = v_i + (v_t - v_i)(1 - e^{t/T})$$
 [3]

where v_i is the initial voltage and v_t is the target

voltage.

The passive delay time can then be solved for:

$$-t_d / T_d$$

V = -V_{OFF} + (V_{ON} + V_{OFF}(1 - e) [4]

where V_{ON} and $-V_{OFF}$ are the transistor maximum and minimum input voltages respectively as seen by the base then the passive delay is:

$$(V_{ON} + V_{OFF})$$

t'd = $T_d ln[$ -----] [5]
 $(V_{ON} - V_{BE}(ON))$

To determine the rise time of the switch, the analysis must include the transition of the waveform through the active region of operation. To accurately perform this analysis the D factor of the transistor must be used. The D factor is given by the equation:

$$D = (1 + \omega_{tR_{c}C_{BC}})$$
 [6]

and is a standard form for aiding in representing the total capacitance from base to emitter when a load resistance is applied to a high frequency transistor circuit and is related to the Miller effect capacitance. For more detailed information the reader is referred to Modern Electronic Circuit Design, by Comer [6].

The minimum base current required to saturate the transistor is:

$$V_{cc}$$

$$I = ---- [7]$$

where β_0 is the large signal current gain of the transistor. If the transistor enters the active region, the collector current will achieve its final value exponentially with a time constant based on the D factor of:

$$\tau = -\frac{D}{\omega_{\beta}}$$
 [8]

where is the corner frequency of the transistor frequency response and is defined as:

$$\omega_{\beta} = \frac{1}{r_{bc}C_{D}}$$
 (beta cut-off frequency) [9]

where C_D is the diffusion capacitance which will vary with the emitter current and r_{bc} is the base to collector resistance.

Turn-off time analysis must consider two cases. The first case is for the fall-time associated with a nonsaturated switch when the base current is greater than or equal to zero and the second case is for a base current less than zero. If the base current is lowered the collector current reacts to this decrease with its own exponential decrease with a time constant equal to:

$$t = ---- [10]$$

as given earlier for turn-on delay analysis review.

If the input voltage changes to a negative value, the base-emitter junction will become reverse biased. However, the base voltage will not change instantaneously and a negative base current exists. The negative current will be present until the capacitance is totally discharged, corresponding to the point in time when the collector current reaches zero. The negative base current removes charge much faster and the collector current will reach a zero value much faster than in the previous case. The time required for the collector current to reach zero is the turn-off time of the transistor. The collector current can be represented by the equation:

 $I_{C} = I_{C2} + (\beta_{o} I_{B3} - I_{C2})(1 - e^{-t}\omega_{\beta}/D)$ [11]

and the turn-off time can be determined by setting I_C to zero and solving for t:



$$t = \frac{D}{\omega_{\beta}} - \ln [1 - ----]$$
 [12]

where I_{B2} is the initial base current and I_{B3} the final value of base current.

If the base current is greater than the minimum level required to put the transistor into saturation, a decrease in the the base current will not have any noticeable effect on the collector current i.e. no instantaneous changes will try to occur. If the base current is suddenly changed from an overdrive condition to a zero value there will be a time delay associated with the recognition of the change in base current and any noticeable effect on the collector current. The time associated with this "recognition" of a change in base current is the storage delay time and is based on the gradient of minority carriers in the base region. The storage delay is determined by the equation:

$$(I_{B2} - I_{B3})$$

t = $T_s ln$ ----- [13]
 $(I_B(sat) - I_{B3})$

where ${\cal T}_{\rm S}$ is the recombination time associated with excess charge stored on the base.

Many of these delays can be controlled by the fabrication process, through manipulation of the operating parameters of the transistor.

2. MOS Transistor Timing

Consider the NMOS transistor circuit shown in Figure III-5.



Figure III-5. NMOS transistor with depletion load [7].

The time to discharge the load capacitance C_L from a high output state to a low state is smaller than the time required to charge the capacitance C_L through the depletion transistor resistance of T_L due to the low resistance of the pull-down device T_D when it is in the on state. The time to charge the load capacitance is longer than the time to discharge since the charging takes place via the high resistance pull-up transistor T_L .

The t_{off} and t_{on} times can be approximated by using the equation:

 dv_{out} $I_{L} = C_{L} ----- = I_{DS} (+ or -) [14]$

Time constants are defined for each of the delay times.

$$T_{\rm D}$$
 is proportional to $\frac{C_{\rm L}}{D}$

and

$${\cal T}_{\rm L}$$
 is proportional to $----$ L

Where \mathcal{T}_{L} is the time required to charge the load capacitance C_{L} and \mathcal{T}_{D} is the time required to discharge the same load capacitance where β_{D} and β_{L} are the respective gains of the pull-down and pull-up devices. The time constant for the pull-down will be much less than the pullup since β_{D} is much greater than β_{L} .

The gain factor (either $_{\rm L}$ or $_{\rm D}$) is given by:

$$=\mu_n c_i ---$$
[15]

where C_i is the insulator capacitance, μ_n is the electron mobility (μ_n for n-channel devices) and the hole mobility (μ_p for p-channel devices) and W and L are the dimensions of the transistor. The greater mobility of electrons than holes in silicon is one reason for the higher gain obtained for n-channel devices. The effective mobility can also be affected by temperature, electric field strength and the fabrication process used. However, the major

contributor to the gain of a transistor is the proper design of the "aspect ratio", W/L, where the gains of the transistors can be individually determined.

In Figure III-6, the input and switching waveforms for the NMOS circuit are shown. As indicated an increase in the gain of the load device would correspondingly reduce the charging time but would use more power since power dissipated by the transistor is proportional to the gain factor β .



Figure III-6. Turn-on and turn-off waveforms [7].

The transient waveforms are similar to the bipolar waveforms previously presented, however the primary difference is the lack of storage time delay in the MOS transistor. In operation a current path is established from source to drain and a voltage applied to the gate controls the source-to-drain current similar to the operation of a valve. A positive voltage applied to the gate with respect to the source, will tend to attract free electrons from the substrate into a layer or channel adjacent to the oxide. This effect is large enough to cause the channel to have an excess of electrons. The drain-to-source voltage will then cause a considerable current to flow. The gate characteristics differ with the bipolar base characteristics and result in zero storage time for a transient signal. The other delays associated with the MOS transistor are common to the bipolar transistor. In the next section Complementary MOS (CMOS) circuits will be evaluated for their switching characteristics. This is an abbreviated, qualitative analysis of NMOS switching. A more extensive analysis is provided in the following section concerning both NMOS and PMOS devices since both are used in the complementary structure of CMOS.

3. CMOS Timing Analysis

The drain-current characteristic of an N-type transistor is a function of the drain and gate voltages:

$$I_{ds_{N}} = K_{N} [2V_{ds_{N}} (v_{gs_{N}} - v_{threshold_{N}}) - v_{ds_{N}}]^{2} [16]$$
for $V_{ds_{N}} < V_{gs_{N}} - V_{threshold_{N}}$

$$I_{ds_{N}} K_{N} [v_{gs_{N}} - V_{threshold_{N}}]^{2} [17]$$

for $v_{ds_N} \geq v_{gs_N} - v_{threshold_N}$

$$I_{d} = 0 \text{ for } V_{gs} \leq V_{threshold}$$
[18]

where K_N is a constant defined by:

$$\kappa_{\rm N} = \mu_{\rm N} \, \epsilon_{\rm ox} W / \, (2 L T_{\rm ox}) \tag{19}$$

where

 $\mu_{\rm N}$ = electron mobility $\epsilon_{\rm ox}$ = permitivity of the oxide layer W = channel width L = channel length $T_{\rm ox}$ = thickness of the insulating oxide

The effective capacitance at the gate of the transistor is:

$$\epsilon_{ox} (L) (W)$$
Ceff= ----- [20]
$$T_{ox}$$

Similar equations for a P-type device can be derived.

For the inverter shown in Figure II-1, the current in the P-type transistor is determined by:

$$I_{P} = K_{P}[2(V_{dd} - V_{out})(V_{dd} - V_{in}) - (V_{dd} - V_{out})^{2}]$$

$$= |V_{threshold}| - (V_{dd} - V_{out})^{2}]$$
[21]

$$I_{P} = K_{P}[V_{dd} - V_{in} - |V_{threshold}|]^{2}$$
for $V_{out} \leq V_{in} + |V_{threshold}|$

$$[22]$$

and

7

$$I_{P}=0$$
 for $V_{in}+|V_{threshold}| \ge V_{dd}$ [23]

The switching response time of the CMOS logic circuit is determined by the amount of current that can be provided by the transistors to charge the load capacitance of the circuit at the input and the output.

If a step input signal is applied to the circuit, the output voltage can be used to solve for an approximate value of switching speed:

$$I_{P} = I_{N} + I_{C}$$
 [24]

$$I_{C} = C_{out} \qquad \begin{array}{c} dV_{out} \\ ----- \\ dt \end{array} = I_{P} - I_{N} \qquad [25]$$

and solving for t:

$$t = C_{out} \int_{V_0}^{V_{out}} \frac{dV_{out}}{[I_P - I_N]} = C_{out} \int_{V_0}^{V_{out}} \frac{dV_{out}}{I_C}$$
 [26]

A minimum value of t is achieved if the current charging the capacitance, $I_{\rm C}$, is made as large as possible.

Based on the inverter circuit, if V_{in} is a positive voltage equal to V_{dd} , then:

$$C_{out} = -I_N$$

$$[27]$$

since the P-type transistor will be an "open" and will not effect the circuit. The NMOS device will be driven into saturation and I_N can be represented by:

$$I_{N} = K_{N}(V_{in} - V_{threshold})$$
 [28]

and

$$C_{out} \xrightarrow{dV_{out}}{=} -K_N(V_{in} - V_{threshold})$$
(29)

since $V_{gs} = V_{in} = V_{dd}$

$$C_{out} \xrightarrow{dV_{out}} = -K_N (V_{dd} - V_{threshold})^2$$

$$= -K_N V_{dd} (1 - \alpha_N)^2$$
[30]

where
$$\dot{\alpha}_{N} = \frac{V_{\text{threshold}}}{V_{\text{dd}}} = \frac{V_{\text{T}}}{V_{\text{dd}}}$$
 (refer to Figure II-2)

solving for V_{out}(t) gives:

Ę

$$V_{out}(t) = V_{dd} [1 - \frac{K_N V_{dd} (1 - \alpha_N)}{C_{out}}]^2$$
 [31]

The fall time can then be calculated as [4]:

$$T_{\rm F} = \mathcal{T}_{\rm N} \left[\frac{\alpha_{\rm N} - 0.1}{(1 - \alpha_{\rm N})^2} + \frac{1 - \alpha_{\rm N}}{(1 - \alpha_{\rm N})} \right]$$
(32)

where
$$T_{N} = \frac{C_{out}}{(K_{N})(V_{dd})}$$
 and $K_{N} = \frac{\mu_{N}C_{IN}}{2L}$

The rise time response can be determined similarly and results in [4]:

where
$$T_{\rm P} = \frac{C_{\rm out}}{(K_{\rm P}) (V_{\rm dd})}$$
 and $K_{\rm P} = \frac{\mu_{\rm P} C_{\rm IN}}{2L}$

Any difference in the rise and fall times results from the difference in the mobility factors of the majority carriers in the P-type and N-type materials, the input (gate) capacitance C_{IN} , and the length of the channel of the respective transistors.

The response times of particular interest for the CMOS inverter circuit are the rise and fall times and the associated pair delay. The pair delay (for the P-type and N-type transistor pair) is approximated by [4]:

$$T = 0.9 T (-----+---)$$

$$(1 - \alpha_{N})^{2} (1 - \alpha_{P})^{2}$$

$$(34)$$

where $T = \frac{C_{out}}{K_N v_{dd}} = \frac{2LT_{Ox}C_{out}}{N V_{dd}} = T_N$

where $\alpha_{\rm N}$ and $\alpha_{\rm P}$ are the normalized threshold voltages for the inverter.

The pair delay is the delay that a signal experiences after it has propagated through two cascaded stages of inversion thus representing the effect of one rise-time and

one fall-time. It is measured at the 50% points of the signal waveform. This is how signals are standardized for experimental analysis. This approximation results from summing T_F and T_R while:

- 1. Ignoring the tanh term
- 2. Approximating the α 0.1 factor with 0.9 in the numerator of the first term

Experimental results have shown that this approximation is within 10% of the measured values [4] and depends on the response of the two stages since one delay will be determined by the N-type device and the other delay will be determined by the P-type device due to the inversion of the input signal.

B. Propagation Delay Analysis

Propagation delay is the result of transmitting signals from one part of a circuit to another or from one chip to another. This brief analysis is based upon the work of Mohsen and Mead [8].

The propagation path is modeled as a large capacitance. As such, delays are associated with the charging and discharging of this capacitive load when signals are transmitted. A driver stage (composed of a driver chain and output driver) is used at the transmitting end of the line and some form of an input (sensing) circuit stage is used at the receiving end of the line. Figure III-7 shows the system design model:



Figure III-7. Large capacitance design model [8].

Delay times exist for the output driver and driver chain, the time required to charge the line capacitance through the output driver, and an input circuit delay. A load (line) capacitance, C_L , is driven with a signal that originates on the gate of an MOS transistor with gate capacitance C_G .

To drive a large capacitance C_L , an elementary driver circuit is used to drive increasingly larger drivers in cascade until the last (output) driver is large enough to drive the load. If the delay of the elementary driver \mathcal{T}_{DR} is then the delay of a driver f times as large is f \mathcal{T}_{DR} . If N stages are used, then the total driver delay is:

$$T_{\rm ch} = Nf T_{\rm DR}$$
 [35]

The output driver circuit, with input capacitance C_D , charges the capacitance of the line, C_L , with a voltage. There is a delay associatd with the output driver which should be included for in the total driver delay $T_{\rm ch}$ and the capacitive load (line) delay $T_{\rm L}$. The input circuit detects the signal at the other end and generates an output voltage $V_{\rm O}$. A delay $T_{\rm i}$ exists for the input circuit.

Figure III-8 shows the relationship among the three delays considered here.



Figure III-8. Delay relationships chart for large capacitance line [8].

The total delay, τ_D , may be calculated as the sum of the driver delay, the line delay and the input circuit delay:

$$\tau_{\rm D} = \tau_{\rm ch} + \tau_{\rm L} + \tau_{\rm i}$$
^[36]

IV. Speed-up Techniques

This chapter provides information on various speed-up techniques. As previously discussed, these techniques deal with methods of circuit modification that can be implemented by the circuit designer if additional speed is desired within a specific circuit being developed.

This chapter begins with a presentation of capacitor pull-up circuits as they apply to MOS circuits in general and concludes with a review of the "precharge" techniques that have been implemented by various designers. The section on precharge techniques begins with applications to NMOS circuits and concludes with CMOS based applications. The intent of this chapter is to present speed-up techniques that are presently or have been used in circuit applications, to provide reference data for the potential circuit designer.

A. MOS Capacitor Pull-Up Circuits

Simple speed-up capacitors may be used to increase the switching speed of bipolar transistors by reducing the storage time during the switching transient. However, for MOS circuits, there is no storage time delay. Alternative uses have been developed however, for adding capacitors to MOS circuits to try to increase their their operating speed through pull-up circuits.

The data available on this technique dates back to 1969 and addresses only its application to MOS circuits in

general. The circuits are "ratioless" type circuits and are based on a single phase clock. The RC time constant of the circuit will limit its operating speed. In normal ratio type circuits the DC current is limited by the use of high impedance loads. This method implements a capacitor as a load element in place of the normal load resistor to increase the speed of the circuit to the range of 5 - 10 Megahertz.

The pull-up circuit is illustrated in Figure IV-1 below [9].



Figure IV-1. Ratioless capacitive pull-up inverter circuit [9].

The characteristics of the circuit are:

- Operating voltage is obtained from the clock pulse
- b. MOS capacitor C4 blocks all DC current
- c. The rise time at point P1 is as fast as the clock pulserisetime dueto C1-C4



capacitance division

1

Calculations indicated that the circuit
 has the capability to perform at 25-30
 MHz

Referring to the circuit of Figure IV-1, capacitors C1 and C2 are non-linear p-n junction capacitances associated directly with the MOS devices, C4 is the load element, C3 is the input capacitance to transistor Q3 of the next stage and C5 is the overlap capacitance of the coupling device. Q1 and Q3 are driver devices and Q2 is the coupling device.

When transistor Q1 is off, during the rise time of the clock pulse the voltage at P1 rises rapidly. The coupling device begins to conduct when the clock pulse reaches the Q2 turn-on threshold voltage. The charge stored on C1 and C2 is redistributed such that V1 = V2. During the fall time of the clock signal, V1 falls rapidly due to an appropriate C1:C4 ratio while node P2 discharges slightly through Q2 during turn-off.

If Q1 is on, then during the clock rise time the capacitive divider at P1 is in parallel with the resistance of Q1 giving V1 an exponential form. As the clock passes the threshold voltage of Q2 it turns-on putting C1 in parallel with C2 + C3. This results in a lower maximum value of V1 which decays toward zero as the clock pulse decays toward zero. The purpose of this technique is to force the circuit to operate as closely to the shape of the input clock as possible thereby reducing delays.

Measured circuit time constants were determined to be short enough for 10-20 MHz operation, somewhat less than the computer predicted operating speed.

Use of capacitors alone as load elements has not developed as a widespread technique for speeding up circuits. The primary reason being the impracticality of designing capacitors as the scale technology progressed from small, medium and large to very large scale integration. The most widely used load element at this time is the depletion-mode transistor, which provides a high load impedance. However, the parasitic capacitances internal to the load transistor do have an effect on the operating speed of the switching transistor, as discussed above.

B. Bootstrap Pull-Up Inverter

The inverter circuit shown in Figure IV-2 is useful for fast switching operations when large capacitive loads are to be driven.



Figure IV-2. Bootstrap Pull-up circuit for increased drive capability [10].

An additonal transistor, T3, is coupled to the capacitance C1 to drive the load capacitance C2. During initial operation, T1 was operating in the saturation region, the driver, T2, was in the triode region, the output was at a low voltage level and a voltage of V_{dd} - V_0 appears across C1. As V_{in} switches to zero voltage, the output voltage increases as transistor T2 shuts off. The gate bias for T1 is then the voltage across C1.

As T2 begins to shut off due to a zero input voltage, the voltage stored across C1 remains during the switching transient. As the voltage at the output increases, the load transistor is forced into its triode operating region which causes V_0 to rapidly reach its maximum. The gate of the load transistor T1, reaches a maximum of $2V_{dd} - V_0 - V_{SS}$, where V_{SS} is the substrate bias voltage. Transistor T3 simply acts as an additional pull-up device that enhances the speed of the voltage rising from a low to a high level.

C. Precharging

The concept of precharging is based on the premise that the transition of a low level signal to a high level takes a longer period of time than the transition of a high level signal to a low level. This premise is valid assuming the discharge of the circuit occurs through a small resistance compared with the resistances encountered when the circuit is charged to a high level, creating a smaller RC time delay on discharge of the circuit. This situation

exists in the standard MOSFET circuit using a depletion mode load transistor.

The precharge concept can be effective for both switching and propagation applications. The precharging process is easily used when a two phase clock is being implemented within the designed circuit. Temporarily unused lines are precharged to a high level through the relatively low resistance of the wire. The capacitance of the wire holds the line high (similiar to the action of a pull-up transistor). This precharging is accomplished under control of phase 1 of the clock and actual circuit switching occurs during phase 2 of the clock.

In this way if the next signal on the line is high the line will already have the high level available on it and if the next signal is a low level, advantage can be taken of the more rapidly occurring transition fall-time of the line from high to low. The careful design of the circuit to synchronize the two phase clock scheme to accomplish this precharge task is critical to valid operation of the circuit. Coordination of either pulling or setting a point or line high is used to reduce the slow low to high transitions. It has applications in carry-chain circuits for ALUS and is frequently used within RAM memory cells for read operations when increased speed is desired in the slow memory access process. The precharge concept can be used selectively within a circuit to set specific lines high on the first phase of the two phase clock. Due to the

discharge times of the individual gates, the precharged line(s) will remain at the high level and trigger subsequent combinational logic to its functionally correct levels (either high or low), until the actual input signals on gates which electrically precede the precharged node propagate to the precharged lines and the gates that follow. As the input signals arrive, they can either be high and take advantage of the already high (precharged) inputs to subsequent gates and not have to suffer a gate by gate delay for each gate, or be low where the more rapid high to low transition time is the only speed-up that can occur. This is similar to the "domino" effect [4] described by Krambeck et al as the precharge is used to initialize the circuit for operation using a two phase clock. In practice, precharged lines to the same gate could, in the optimum case, propagate the correct signal level to the output of the combinational logic, occur independently of the input, and reduce the gate delay for circuits beyond the precharged node to near zero, since all the gates will be set to the correct high or low level before the actual inputs ever reach the combinational logic themselves. If high levels occur on certain lines more frequently during operation in comparison to other lines, and this can be predicted during the design process, then those lines should be precharged. The advantage of thii is that this charging to a high level is automatically applied to combinational logic gates that follow the precharged layer of gates, thereby "presetting" the inputs and outputs

of the following gates so that "highs" on the "real" inputs that occur later won't have to do it and the only gate delays that will occur happen while the "real" inputs are propagating to the location of the precharged gates.

For switching applications, fabrication techniques and new materials are progressing to the point where equal rise and fall times will be achievable. Even so, this application of precharge will continue to be valuable. However, the designer should keep in mind the limitations of this The maximum frequency of operation of a application. circuit will be determined by the speed of the slowest part of the circuit. Even though some lines may be precharged, and the delays along these portions of the circuit reduced, if other parts of the circuit operate slower, i.e. a large number of "real" input lows occur, then this slower part of the circuit and its associated delays will determine the maximum frequency of operation. This effectively negates the addition of the precharging, since the remainder of the circuit must wait for the slowest part to complete its function.

A complementary alternative to the precharging concept is the selective grounding of certain lines that are expected to have a high occurrence of low signals on them. This produces the same effect as the precharge since the ground, or low signals are propagated through the combinational logic that follows the "grounding" layer and set the gates that follow to the correct level before the

actual input signals arrive. The delays associated with these following gates are again reduced/eliminated since the real input signals will not have to cause the actual switching of the gates since it will have already occurred by the grounding taking place on phase one of the two phase clock. The effective use of precharging or grounding is clearly dependent upon the individual delays of the gates being used. The precharge/ground must have time to occur (and stabilize) before the actual input signals arrive at the precharged layer of gates. Additionally, phase two of the clock, which controls when the signals are passed out of the "black box" circuit, must not occur until the actual input signals have propagated through the circuit itself. This aspect of delay reduction will continue to be a function of the individual gate delays which are controlled primarily by the fabrication techniques used. However, the use of precharge will remain as a good means for reducing the delays associated with long, high capacitance lines within circuits as well as for the selective method discussed above.

There is very little reference material on the concepts of precharging. Most of the data available deals with specific applications in which precharging was used. No established guidelines seem to exist at this time. The following information refers to specific implementations of precharge or precharge-like techniques. Although the first section below is directed at NMOS circuits, the applications

of precharge can be modified to CMOS design. A review of precharge-based circuits follows.

1. NMOS Applications

a. Charge Pumping

This charge pump technique was designed to realize the standard erase-then-write two step write operation of a memory circuit in a single step. The charge pump circuit is shown in Figure IV-3 [11]. The charge pump circuit was applied to the read/write circuit as shown in Figure IV-4. The actual operation of the circuit will not be discussed here. Any reference to actual circuit operation is only intended to aid in evaluating the operation of the precharge circuitry.



To Charge Pump Circuit

Figure IV-3. Charge pump circuit [11].


ļ

ľ

Figure IV-4. Read/Write circuit [11].

When the read operation begins, Q_{CG} , Q_{CD} , Q_E , and Q_D are turned on by the column select (CS) and the row select (RS) respectively. The data is stored on Q_M and is provided at point B for output. If Q_M is ON originally, then node B is low and node A is high after the read operation. The charge pump circuit pulls the level of node A to the value of V_{dd} and since the path to ground through point B is still closed, point B will remain low. After a short time Q_M will invert its state due to tunneling and node B will be pulled high by the charge pump circuit. Therefore by applying CL to the charge pump circuit, the state of Q_M can be inverted. The simultaneous occurrence of Q_M driven to a high or a low, and node B driven to the inverse of the signal stored on the gate of Q_M , turns the normal two step erase-then-write process into a single step.

Although referred to as "charge pumping" this circuit performs a simple precharge function on the memory cell. When the clock pulse occurs, the state of the transistor Q_M will determine if the node A will be charged to a high level and whether or not node B will be high or low.

b. NMOS Dynamic RAM Cell

Newkirk and Matthews [12] implement precharging when their dynamic RAM cell is used in a 3 transistor memory subsystem, with an interface and address cell. The interface cell is precharged during phil, the first clock pulse of a two-phased clock. If sense (read) is high then ExtIO will be high due to the precharge unless a stored

"high" discharges the precharge. If drive (write) is high, then the input value on ExtIO to be written to the memory cell will determine if the precharged line is grounded (value on ExtIO is high) or if it stays at a high level (ExtIO is low). Figure IV-5 shows a diagram of the memory subsystem.





In this implementaion, the precharge scheme was used to facilitate the reading and writing of data to the memory cell. An underlying benefit of using the precharge is an increase in the speed of the read and write operations of the cell. Memories are inherently slow in this regard and precharging is an excellent way to shorten the memory access cycle.

c. <u>CalTech OM2</u> Arithmetic Logic Unit (ALU)

Mead and Conway [13] provide a review of their "Our Machine" (OM) Project at Caltech and emphasize the system requirement for a high performance carry chain within the ALU since they felt the carry chain would have the most limiting effect on the performance of the system. They implemented a Manchester-type carry chain, and since it had a limited capability to propagate high signals, they took advantage of a null period experienced in the processing cycle of the op-code of the computer. They decided to precharge the data paths of the carry chain, to reduce the requirement for high-level signal propagation through relatively slow pass transistors. The carry-out line was precharged and the actual data being supplied to the carryout line determined if the carry-out line would be shorted to ground or not. Figure IV-6 shows the carry chain circuit for the ALU.



Figure IV-6. Carry chain circuit for OM2 ALU [13].

2. CMOS Applications

a. Domino Logic Circuit

Krambeck et al [5] apply the concept of precharging in their design of a CMOS domino circuit. They employ a precharge signal that is used to turn on all the gates in a circuit at one time thereby reducing the requirement for complex timing schemes and reportedly taking advantage of the inherent speed of the CMOS gates. Specific application of their domino circuit to an 8-bit ALU has indicated through simulation to be one and a half to two times as fast as traditional circuits performing similar functions. The domino circuit is shown in Figure IV-7. The values of the inputs (I1 - I5) to the transistors determine the output of the domino circuit cell after the precharge. Individual dynamic cells are isolated from each other by inverter stages.



Figure IV-7. Domino logic circuit [5].

In addition to increasing circuit speed, Krambeck's circuit has eliminated the internal clock race condition by placing the inverter after each logic block for isolation. During the precharge phase, the outputs of all the inverters are driven to a low level by the precharged node and therefore all n-type transistors driven by these inputs are turned off. During phase two of the clock, any internal clock delays cannot incorrectly discharge the storage nodes since the path to ground through the n-type transistors is open. All internal nodes can make at most one transition until the next precharge, therefore a stable propagation state exists. Since this process is similar to the falling of dominos as the signal passes from block to block, this name was applied to the design.

b. Domino Logic Modification

Gonsalves and Man [1] describe a dynamic CMOS

technique making use of the domino circuit previously described. The redundancy of information characteristic of CMOS, low power dissipation and lower capacitances, thus higher speeds, can be achieved if the n-type dynamic CMOS logic block shown in Figure IV-8 is used in lieu of standard CMOS circuitry. Silicon area is also reduced since only N ntype devices are used and 2 p-type devices (total N+2 devices) are used instead of N p-type and N n-type devices as in standard CMOS.



Figure IV-8. n-type dynamic CMOS logic block [14]. The inherent deficiencies of Krambeck's design are determined as:

- The combination of the dynamic block and inverter results in inverted signals which decreases logic design flexibility
- The clock race problem is just controlled, it is not overcome

These deficiencies are resolved by a specific "No Race" (NORA) [1] design approach. However, the precharge scheme

is treated in the same manner as implemented by Krambeck. The primary difference between the two approaches is Krambeck's use of inverters for isolation versus Gonsalves use of latch stages to store information in successive stages, his addition of p-type dynamic blocks as well as the n-type domino logic cell used by Krambeck and a subsequent emphasis on pipelining of data within cirucits.

3. High Speed Precharge Circuit

Stewart and Plus have integrated precharging to provide a high performance, high speed EEPROM [15]. The precharge generation circuit is shown in Figure IV-9. The precharge signal generated in the precharge and control circuit is used to switch on the decoders and sense amplifiers following changes in the memory address inputs.

A high speed parallel decoder and sense amplifier circuit which detect small voltage changes on the lines as data is read out of the memory cells. The actual operation of the circuitry will not be discussed. This circuit simply demonstrates another application of precharging used to drive other circuits.

The parallel decoder is shown in Figure IV-10 and the sense amplifier circuit in Figure IV-11.



PC - Precharge

6

D





Figure IV-10. High-speed word decoder [15].

IV-19



PC - Precharge

Figure IV-11. Sense Amplifier Circuit [15].

When the precharge occurs, the source of P1 remains high and the drain of P3 is set to low and NODE 3 will be precharged high. This sets up the circuit for inputs from A1 through Am which will determine the final value of NODE 3.

4. Bootstrap Circuit

Masuda et al have designed an advanced and accurate 5-V "bootstrap" generator circuit for accessing a MOS structure 64K RAM and achieved 5 nanosecond delays using another form of precharge [16].

The bootstrap circuit is simply a two phase precharge circuit similar to those already mentioned.

D. Propagation Delay Reduction

There is one standard rule to follow when

designing for minimum propagation delay: avoid dominating the chip area with wiring interconnections.

a. Cross Under Effect

In addition, Anami et al [17] have recommended an approach for reducing the "crossunder effect" and its associated delay due to the parasitic effect of the resulting resistance. The designer should avoid crossunders in the critical signal paths of the circuit. The aspect ratio can be minimized however to reduce any loss in speed and the optimum width of the crossunder can be expressed by:

$$W_{opt} = [(C_t R_{xo}) / (R_o C_{xo})]$$
 [1]

where R_0 is the output impedance of the line driver driving the crossunder line, C_t is the capacitance following the crossunder, C_{XO} is the capacitance per unit area of the crossunder and R_{XO} is the sheet resistance of the crossunder. They have verified this equation through experimental use of ring oscillators.

b. Large Capacitance Loads

Mohsen and Mead [8] have addressed driving a large capacitive load as referred to in Chapter III.

Mohsen and Mead have evaluated specific cases for minimization. Delay times for push-pull depletion load driver stages with a single-ended depletion load input stage were analyzed. For this case the following analysis was derived [8]. Total driver chain delay is minimized when each successive driver stage is larger than the previous one by a factor of e (the base of the natural logarithm). Where:

$$Y = \frac{C_L}{----} = f \text{ and } \ln(Y) = N[\ln(f)]$$

$$C_G$$

and

$$= \ln Y \begin{bmatrix} f & C_L \\ --- \end{bmatrix} = e \ln \begin{bmatrix} --- \\ --- \end{bmatrix}$$
[2]
$$\ln f & C_C$$

Minimum total delay is achieved with a fan-out factor f equal to the base of the natural logarithm, e.

Depending on the application, the input stage delay can be optimized to different values.

Their overall conclusions suggest that for minimum delay time, the designer should try to keep the delays of the driver circuit, the high capacitive line and the sense circuit comparable, i.e. $T_{\rm Ch} = T_{\rm L} = T_{\rm i}$. This conclusion follows from the delay relatonships presented in Figure III-8.

c. <u>Repeaters</u>

A repeater is simply a series pair of inverters. Chwang et al [18] have successfully implemented both precharging and line buffers/repeaters in their high density 64K CMOS Dynamic RAM. The repeaters used in series between 8K memory array segments reduced the word-line delay from 30 nanoseconds to 10 nanoseconds. This application uses both precharging and simple inverters to increase circuit operation. Figure IV-12 shows the dynamic RAM arrays with the repeaters located between arrays and Figure IV-13 shows the use of precharging within the circuit.



Figure IV-12. 64K Dynamic RAM with repeaters [18].



Figure IV-13. Precharge implementation circuit [18].

E. General Remarks

This chapter consists of all the data that was found as a result of a continuing literature search for information. Much data was found referring to fabrication methods and improved selection of fabrication materials. These appear to be the primary areas in which most of the effort is being expended to improve the speed of circuitry.

Different terms have been used i.e. charge pumping and bootstraping, to describe, after analysis, what in actuality is a form of precharging. However, the term, "bootstraping" has also been associated with transistor pull-up circuits. The reader is cautioned to disregard the terms used to describe specific techniques and really



determine exactly what the circuit is doing prior to considering it for use in a design, since different names are sometimes associated with the same method or technique.

F. CMOS VLSI Applications

1. Requirements

The major requirements for a specific speed-up technique to be considered useful are (in order of importance):

- a. Simplicity
- b. Ease of Implementation
- c. Applcability to variety of circuits
- d. Frequency of Use

The designer should be able to implement the method with minimum modification if a circuit exists already, the particular technique should use a minimum of chip area, which becomes critical in VLSI applications, and should be a simple enough approach to permit use by even the most basic of designers.

The frequency of use of a certain technique is a measure of the design community's acceptance of the technique for use. This requirement is included as a justification for selection of a specif c technique for evaluation during this thesis.

2. Selection of Technique

From the information provided in section A of this

chapter, and from a review of available literature, the most commonly used speed-up technique is precharging or some modification thereof; i.e. Krambeck et al [5] and the charge pumping scheme [11]. There are two primary methods of precharge implementation. The first method, using precharging on output lines with a two-phase clock, permits an implementation of a precharged line as shown in Figure IV-14 below. The line is charged high in phil and the value of the data and the phi2 clock determine if the line will be grounded or not. This circuit is useful for reducing delays over large capacitance lines and will reduce the charging delay time of the lines. This method was not considered for simulation during this thesis.





The second method precharges selected nodes within a circuit using individual transmission gates that pass Vdd to the selected node when a pass signal is applied to the transmission gate. This normally occurs on phase one of a two phase clock. This scheme is shown in Figure IV-15. Any number of transmission gates can be used for this appication, however a single transmission gate is required for each line that is to be precharged to maintain isolation of the individual signals.



Figure IV-15. Internal transmission gate precharge scheme.

Implementation of transmission gate circuits can be accomplished during the design of the circuit with no additional effort and with minimal use of circuit area. The precharge scheme can be used anywhere within the circuit where additional speed is required and can be applied to any circuit design. Any line that can be charged high and then pulled low or left high, based on circuit data manipulation, can be subjected to precharging. Transmission gates will be discussed in more detail in chapter V.

These characteristics make the precharging concept a reasonable choice for evaluation.

From review of the literature, precharging was the most popular manual design technique used to speed-up circuit operation. Although it was frequently used, little data was available on exactly how good an approach it really was. Many designers had used it, with reported increases in speed, however, a consolidated source of information was not available. Since precharging had the highest frequency of use, an analysis of its capabilities was performed.

Therefore, the remainder of this thesis was concerned with applying precharging to an existing ALU bitslice and to newly designed circuit elements using CMOS/SOS technology. The circuits were extensively simulated using SPICE transient analysis to determine the success of precharge application.

Chapters V and VI present the precharge design of the ALU bitslice and the design and precharge of additional circuit elements to be used for simulation.

The results of the simulations are presented and evaluated in Chapter VII.

V. System Design

A. Introduction

The design portion of this project was divided into two main areas:

1. CMOS/SOS ALU Bitslice modification using precharge techniques

2. Implementation of a circuit using the modified ALU bitslice as a circuit element built around a 4x4 bit Random Access Memory (RAM) referred to as the Memory/ALU Bitslice circuit

In this chapter, the modified ALU bitslice using precharging, and the Memory/ALU Bitslice circuit design, incorporating the modified ALU bitslice, are presented. The more elaborate circuit was used to demonstrate the applications of precharging techniques to elements within a more complex circuit. The applications presented here demonstrated the flexibility of precharging as a design tool which can be used in new designs as well as being added to existing designs.

B. Requirements

1. The following design requirements applied to the modification of the ALU bitslice and the design of the Memory/ALU Bitslice circuit:

a. The bitslice precharge circuitry must be:

1. Capable of speeding up circuit operation

2. Be easy to implement

3. Use as little area as possible

4. Apply to a wide variety of circuits

b. The Memory/ALU Bitslice circuit must:

1. Demonstrate the precharging concept applied at a chip level, i.e. a large circuit application.

2. Demonstrate uses of precharging applied to elements commonly used by the circuit designer

2. To meet these requirements, SPICE2 was used for simulation. To determine the performance of the precharge speed-up technique, the following steps were followed:

a. Design a test circuit

2

- b. Simulate the test cirucit with SPICE2
- c. Use a variety of precharge configurations and simulate circuit operation for each configuration
- d. Compare test circuit results
- e. Simulate the 2x4 decoder circuit using SPICE2
- f. Apply precharging to the decoder, RAM and registers
- g. Simulate the operation of the circuit using SPICE2
- h. Compare the results of the two simulations
- i. Simulate basic bitslice operation using SPICE2
- j. Select bitslice nodes to apply precharge
- k. Design of the precharge circuit itself
- 1. Implement the precharge modifications to the circuit

- m. Simulate operation of modified bitslice with SPICE2
- n. Compare the results from each analysis
- o. Incorporate the ALU bitslice into a circuit using a 4x4 memory circuit
- p. Evaluate results and present conclusions

C. Circuit Selection

For the evaluation, a test circuit was designed for basic precharge and timing evaluation and the first bitslice of the CMOS/SOS ALU designed and developed by Sommars [3], was selected as a more complex circuit to which the precharge technique was applied and evaluated. At the time this thesis took place, the ALU and the CMOS/SOS library were the only existing CMOS/SOS based VLSI components available at AFIT. The precharge technique was applied to only the first bitslice of the ALU to limit the scope of the simulation required and to eliminate redundancy of effort for application/evaluation since the subsequent bitslices of the ALU differed only slightly from the first one and thererfore could be precharged in the same way.

D. Design Rules

This design effort made use of the same design rules used by Sommars [3] in support of his design of the CMOS/SOS library cells based on the unpublished notes of Seitz [18].

E. ALU Bitslice Modifications

The output lines of the ALU were precharged on phase

one of a two phase clock and the data will be read out of the ALU on phase two of the clock. The capacitance of the output lines will be charged on phase one and the value of the data (high or low) will determine if the line is discharged to ground through the low resistance of the pulldown transmission gate or will remain high at the output of the circuit. The CMOS transmission gate is used for the precharge pass element. This gate is shown in Figure V-1.

4



Figure V-1. CMOS transmission gate [7].

This gate ideally has zero resistance when closed and infinite resistance when open. It is formed by joining the sources and drains of a pair of MOS transistors of opposite polarity. Control signals on the gates of each transistor determine whether or not the signal will be transmitted. If each device has a threshold voltage of 2 volts, and the control voltages to the gate of the p and n type devices are

0 and 10 volts respectively, the n type device will remain "on" until the input to the transmission gate reaches 10 - 2 = 8 volts, when it will turn "off". However, the p type device will continue to conduct since the gate to source voltage of 8 volts is greater than the p type device threshold voltage of -2 volts. In this manner the entire supply voltage, Vdd, can be transmitted by the gate and a "good" precharge can take place. Two precharge circuits were used; an internal precharge circuit and an output precharge circuit. The internal precharge circuit was used to selectively precharge nodes within the bitslice circuit and the output precharge circuit precharged only the output lines of the bitslice. The basic block diagrams for the internal and output precharge circuits are shown in Figure V-2 and Figure V-3 respectively. Block diagrams of the unmodified and the modified ALU bitslices are also provided in Figures V-4 and V-5 for comparison.



Figure V-2. Internal precharge circuit.



1.1

Figure V-3. Output precharge circuit.



Figure V-4. Unmodified ALU design [3].





F. Memory/ALU Bitslice Circuit Design

To complete the second part of this system design, a circuit was designed that incorporates the modified ALU, makes active use of other CMOS/SOS library cells and requires the development and implementation of additional circuit components. Precharging is applied to this expanded circuit (in addition to the ALU) to further demonstrate the use of precharging techniques in basic design.

In addition, design of the Memory/ALU circuit

demonstrates the utility of the CMOS/SOS library for practical design and sets the stage for future design efforts in the CMOS/SOS area.

The Memory/ALU circuit was designed to perform in conjunction with a microprocessor controller. The circuit is composed of the basic ALU, a 4 bit by 4 bit static Random Access Memory (RAM), a row decoder and the necessary control logic required to manipulate the memory and the ALU.

The transmission gate is used in this design to clock the signals within the circuit, since the use of transmission gates is presently the only effective way to control and isolate clocked signals in CMOS circuits.

1. <u>Circuit</u> Function

The function of the circuit is as follows:

Data will be written into memory in 4-bit words in accordance with a decoded 2-bit row address, with both data and address supplied by the microprocessor. Data processing can be performed for two operator functions and for single operator functions. To perform arithmetic or logical operations between different operators, two registers will be implemented to hold the two respective 4-bit words that have been read from memory. The registers will be selected by a register select line. Data is first read into the A register. As data is subsequently read into the B-register, the register select signal will pass the ALU command on the

AD-A1	SPE Sen Nri	SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL Semiconductor Yery Larg. (U) air force inst Wright-Patterson AFB oh School of Engi. B							OXIDE OF TECH T KELLEY		2/	215	
UNCLA	SSIFIE	0 14	DEC 84	AFIT.	/GE/EN	G/84D-	-41				F/G 20/12		
L - 2		aka e aka e			2.48 -45 -47 -7								
					2 66					an a			
						; 					а 15-15		



MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A SO-S3 input lines into the ALU, and the function selected by SO-S3 will be performed. If only a single operator is required, i.e. A + 1 (an increment function), the register select line will still select register B, after data is entered into register A, but the contents of register B will not be used. For either case, the data from memory is applied to the ALU for manipulation based upon the command input to the ALU generated by the microprocessor controller. The output of the ALU can be monitored to determine if the circuit performs the desired functions correctly.

For single chip implementation, the following signal lines will be required:

a. Two address li	ines
-------------------	------

- b. Four data input lines
- c. One read/write line
- d. Four ALU command lines
- e. Two ALU bitslice output lines
- f. One register select line to the ALU
- g. One external clock line.
- h. One chip ground line
- i. One Vdd line
- j. One substrate connection

Total: 18 chip pins

The system circuit diagram is shown in Figure V-6.



Figure V-6. System circuit diagram.

2. Additional Components Required

In addition to the existing ALU, the following additional circuit components that will be required are:

- RAM cell static (no refresh required) from which a 4x4 bit RAM was realized
- 2. 2x4 Decoder
- 3. Two four-bit registers
- 4. Read/Write Control
- 5. ALU input control

The basic decoder, 4x4 bit RAM and the register

circuit elements were initially designed, and then precharged. This was done to demonstrate the flexibility of precharging during the circuit design phase.

F. Memory/ALU Circuit Chip Layout

The pad layout for the expanded circuit chip is provided in Figure V-7. The choice of chip pads is limited for CMOS/SOS applications. Pads will be selected from the following list:

- 1. padVdd
- 2. padground
- 3. padblank
- 4. padout (composed of paddriver and padamp)
- 5. padin

At this time no capability exists to generate a two phase clock on chip. Therefore, these circuits were designed with the intention of driving them with an off-chip two phase clock source. Pad types are represented by their respective numbers listed above.



Figure V-7. Pad layout for chip.

VI. Detailed Design and Analysis

This chapter begins with a brief analysis of the predicted advantage of precharge when used with the CMOS/SOS library cells in circuit design.

The chapter concludes with a presentation of the detailed designs of the modified ALU bitslice and the Memory/ALU Bitslice circuit. The precharge circuitry which is applied to the ALU bitslice and the elements of the Memory/ALU Bitslice circuits are shown in their schematic and CLL plot layouts.

A. Test Circuit Delay Analysis

This analysis is intended to provide the reader with a better understanding of the expected advantage of using precharge as a circuit speed-up technique. The timing values presented are those results achieved by Sommars from his SPICE2 simulations of the individual CMOS/SOS library cells that were designed [3]. SPICE2 circuit simulations were performed on this test circuit to determine the validity of this analysis and demonstrate the precharge concept. The results of the simulations are discussed in more detail in Chapter VII.

This is a rough approximation analysis of the delays associated with logic gates to be used for this design and is only intended to provide the reader with some basic reference information to more easily understand the nature of the delays being evaluated. In Figure VI-1, a series logic circuit is presented composed of one nand gate and two inverters.

· ·



Figure VI-1. Delay analysis circuit.

The delays referred to here are taken from SPICE2 analyses performed by Sommars [3].

The delays in the nand gate, as the inputs change, are:

inputs: 0 volts to 5 volts - 1.0 nsec

(output goes from high to low)

inputs: 5 volts to 0 volts - 1.0 nsec

(output goes from low to high)

When inputs A and B are both driven high, OUT1 is driven low and OUT2 is driven high. OUT1 experiences a 1.0 nsec delay before it reaches its final low value.

The delay times for the inverter are:

(output goes form low to high)

If A and B are driven low then OUT1 will be driven to a low value after 1.0 nsec, OUT2 will be driven high after a

VI-2

0.25 nsec delay and OUT3 is driven low after an additional delay of 0.15 nsec. The total delay time is calculated in equation [1]:

$$1.0 + 0.25 + 0.15 = 1.4$$
 nsec [1]

If A and B are driven high the same delay time is achieved due to the equal rise and fall times for the respective gates. Therefore the maximum operating frequency of the test circuit is:

$$1/1.4 \text{ nsec} = 714 \text{ MHz}$$
 [2]

If precharge was implemented, and applied to OUT1, then OUT2 and OUT3 would be driven low and high respectively. If nand inputs A and B were subsequently driven low, driving the output of the nand gate high, maximum advantage could be taken of the high precharge at OUT1 and the only contribution to the circuit delay time would be the nand transition delay (which would, theoretically, be reduced since the nand wouldn't necessarily need the entire 1.0 nsec to drive the OUT1 node high when it is already precharged high), since the precharge signal would have already "set up" the circuit. Therefore the overall circuit delay would be reduced by 0.4 nsec and the operating frequency would be:

1/1.0 = 1000 MHz [3]

Therefore, with the nand output driven high by inputs A and B, a 40% increase in operating speed could be achieved by

VI-3

precharging the OUT1 node. However, if the nand inputs drive the output low after it has been precharged, then OUT1 will be driven from a high precharged level to a low level, OUT2 will be driven high from a low level and OUT3 will be driven low from a high level. The only delay reduction that occurs for this scenario is the reduction in transition delay in the second inverter, since OUT3 will be driven low from a high level and in an inverter, a high to low transition takes 0.15 nsec and the low-to-high transition takes 0.25 nsec in an inverter. The total delay is reduced by only 0.1 nsec, from 1.4 nsec to 1.3 nsec, and a maximum operating frequency of 769 MHz, a 7% increase in operating frequency.

If each output, OUT1 through OUT3 were precharged on phase one of the clock, even if the output of the nand were driven low, a gain in speed would be achieved since OUT2 would already be high and OUT2 would not have to wait for the inverter delay. The delay would then be reduced by the 0.25 nsec savings incurred by the precharge. The maximum operating frequency would then be:

$$1/1.15 \text{ nsec} = 869 \text{ Mhz}$$
 [4]

Therefore, a 21% increase in operating frequency could be achieved with the addition of the precharge. This method is only valid for alternating inversion of signals within the combinational logic. This situation occurs frequently within common combinational circuits and can be taken

VI-4
advantage of by the circuit designer by applying precharge techniques. This is the method by which the nodes within the bitslice were selectively precharged. If all of the nodes, OUT1, OUT2 and OUT3 are precharged, then no matter what the output of the nand gate (either high or low), there will always be at least a minimum time savings of one transition from low to high across an inverter, either INV1 or INV2.

This delay approximation can be performed for each possible precharge configuration that could be applied to the test circuit.

This analysis provided the baseline from which the results of the SPICE simulations of the test circuit, Memory/ALU bitslice circuit elements and the bitslice were interpreted.

B. Modified ALU Bitslice

The detailed design of the precharge implementation was performed in three steps. First, the lines to be precharged were selected. For simulation, two precharge configurations were used, the bitslice with ten internal nodes precharged and the bitslice with four nodes precharged. The two configurations were chosen to determine the best approach to precharging the bitslice based on the results of the test circuit simulation. The bitslice was divided into three sections and each section was simulated independently. The simulations and results are presented in more detail in Chapter VII. Secondly, the layout of the circuit was determined to minimize wasted space, although this is not critical for this application and simulation, and remain within the established design rules. The third step was the active implementation of the circuit using CMOS/SOS library cells developed by Sommars [3]. The CLL layouts of the internal precharge circuit for each bitslice precharge configuration are shown in Figure VI-2. Some redesign of Sommars' ALU was required to permit SPICE simulation and to correct deficiencies in design.



a.



Figure VI-2. CLL layout of internal precharge circuit. a. Ten node precharge b. Four node precharge Ten transmission gates are required for the internal precharge circuit of the first approach and four transmission gates are required for the second precharge approach. A single transmission gate is required for each precharge_line. Therefore, to precharge more lines the designer need only add more transmission gates.

The output precharge circuit is more complex since it is based on the output line being pulled down to ground or left at a high level depending on the value of the data on the output line. This precharge scheme causes the output of the ALU to be inverted for each line that is precharged. If the output from the ALU is high, the precharged output line will be grounded. If it is low, the output line will remain high. The advantage of the precharge applied here is that, although they are inverted, the outputs from the ALU are realized more quickly, since all the output lines are charged high and only the high data outputs will cause any type ~ 5 delay when the precharged line is grounded through the low impedance transmission gates to ground. This precharge approach is typically used for high capacitance lines and is recommended by Ullman [24]. A CLL layout of the output precharge circuit is shown in Figure VI-3.

This circuit was not used in the circuit simulations, since analysis of delays in long high capacitance lines was not the primary purpose of this thesis.



Figure VI-3. Output precharge circuit CLL layout.

If the output precharge circuit had been used, the first bitslice of the ALU would have used only one such circuit since the bitslice has only output. However, the output precharge circuit could easily be expanded to handle more precharge lines by addition of more output precharge circuits.

The ALU performs the basic functions of the Motorola ALU after which it was modeled, which are listed in Table VI-1.

			_	Logic Functions	Arithmetic Operation
- Fu	netio	n Sei	ect 👘	l Nins High	Mis Low or LSE must be High
\$3	52	51	50	F	F
		_	L	F - X	F + A minus 1
Ē	Ē	Ē	н	#+A-B	F + A price (A + E)
L.	ι.		6	F • I • 6	F + A gius (A + E)
L	6	Ħ	м	F + Lagrest "0"	F + A times 2
٤.	H	L		F + A + 8	# + (A + 8) minus 1
6	н	L	N	* - 8	F + (A + 8) alus (A + 8)
ι.	-	H	L	F - AQ1	F = A plus B
ι.	H	H	H	F + A + 5	. F + A plus (A + B)
H	L	6	6	F+I-8	F = (A + 8) minus 1
- 10	۰.	L.	м	[F+A38	F * A minus & minus 1
H	6	H	ι.	F • C	F + (A + E) plus (A + S)
14	L.	м	Ħ	F +A+ R	F = (A + B) give A
14	H	Ŀ	6	F + Logical "1"	F = minus 1 (two's complement)
24	10	μ.	H	F+A+B	F + (A + E) prus 0
н	м	н	ι.	P+A+8	F = (A + 8) plus 0
		-			F = A prus G

Τa	able	VI-1.	ALU	Func	tions
----	------	-------	-----	------	-------

The first bitslice schematic diagram is shown in Figure VI-4 and the associated CLL layout is shown in Figure VI-5.



Figure VI-4. ALU Bitslice Schematic Diagram [3].



Figure VI-5. ALU Bitslice CLL Layout.

The ALU bitslice schematic diagrams with the precharge circuit implemented for the ten node and four node precharge configurations are presented in Figure VI-6.

VI-9

.





• 1



Figure VI-6b. Detailed Design Schematic of Bitslice with Four nodes precharged.

A CLL layout of the two precharged bitslice configurations is provided in Figure VI-7.

-



Figure VI-7a. CLL layout of Ten node precharge bitslice.





C. Memory/ALU Bitslice Circuit Design

The Memory/ALU circuit design was accomplished in three steps. First, the schematic designs were generated for each circuit element within the limits imposed by the availability of logic circuit cells in the CMOS/SOS library. The following gates were available:

- Inverter Gate (Slow and Fast types)
- 2 Input NAND Gate
- 3 Input NAND Gate
- 4 Input NAND Gate
- Transmission Gate

- 2 Input NOR Gate
- Double Buffer

Second, the elements were layed out to minimize the area used and remain within the imposed design rules. Third, the circuits were implemented using the available NMOS CAD design tools and required CMOS/SOS library cells.

The following circuit elements were designed:

- Two 4 Bit Registers
- Read/Write Controller
- ALU Bitslice Input Controller
- 2x4 Decoder
- 4x4 Bit Static Random Access Memory (RAM)

1. Register Design

Two different size registers were required. The A register received data first when data is read from the memory to be processed. Data is read from memory and stored in the A regsister on phi 1 of a two phase clock. On phi 2 of the same system clock cycle the data is retained in the A register and, on the next phi 1, data is read into the B register. On the next phi 2 pulse the data from both registers is read into the ALU for manipulation. To properly time this process the A register is designed twice as large as the B register so the data it contains may be retained throughout another read cycle as data is read into the B register. Both registers are composed of transmission gates to isolate the signals and inverters on which the signal is stored temporarily as it is passed between register layers. A register select signal controls the direction to where the read data is stored. The schematic diagram for the A register is shown in Figure VI-8 and the associated CLL layout in Figure VI-9. The third and fourth layers of the register reverse the signal connections required to pass the signal between layers since during this time data is being read into the B register and the register select line will be selecting register B. A schematic diagram and CLL layout of the B register are shown in Figures VI-10 and VI-11 respectively.

TO ALU

From Memory

Figure VI-8. A register schematic.

1		┢	+	+	+			+	+	+	┣—	-	⊢	┣	⊢	-		-	+		-	$ \rightarrow \rightarrow $	+	+ +	-	+ +	<u> </u>	+		-+	-+-	+-+
L			L																											_1	L.	
Г		Γ	Ι	Γ					Γ	Γ		[_		Γ	Γ	Г		П					Т		П			П				
⊢	⊢	t	+		1	t				-	-	t	ł		t	+	t	IF 1		-			+	1 1	++-	+		H	1 1	-+	+	+
┝	_	╆	┢	┢	⊢		9	-	ł	┢	⊢		┢─	┣	1					-					н.,	+				-+	_	+
			1		L		1			<u> </u>		L			2	1	Ĩ							Ĩ st			18			_1		
[~	L	L.,	-	Į		4	L	Į	_	-	ι –			И	T							T					72	ΠI	Т	Т	Τ-
┝━	H	τ.			<u> </u>	÷.,		÷	÷	È.,	Εt	r	-	t-	11	t	16	T					+	i la	111		16	÷16		-+	+	+
⊢		€	⊢	╇	←	-	44	4			н.	-	-	⊢	ш	╘	44	4	1				_			4	_ i.i			-+	_	+
		1					6								И	1 M			ĽĽ	μ4.						- C			6 4			
			T	Г	Γ		-		Г		Г	Г			И	И	16	11		Г		5 5	T	1.	1.8			18				
		-	Ł	İ.,	<u>t -</u>		H		⊢	┢─	IL.	┢─	-		Ľ	Ť.		2		-	-			1-7	141	1		5		-+	-+-	+-+
-								5.4	ė-	⊢	4		2		12	-	les i				-		+		-	+		-		_	_	+
		5	381	H									1	a la		И	III.		L					ΙĦ			17	T	1	- 1		11
Г	E	L to					5			T		131	•	ί.,		Æ	12 3		-			1 10 1000			-	<u>е т</u>		1 2.7		7.1		1-1
⊢		-6		72	1	-			- H	+	۲					#1					-		-	#			- 4	10.011		44	<u> </u>	╈╪
<u> </u>			<u>84</u>	÷	- 1		ببا	H	-	-			Ц.	2	- 1	11					_			旧形			H	-		1	_	111
77	1	2	24	100	21					100						Ű.	11	Ι			88	1-1-18	-	118	- 18		311	11		ы	1	111
	I	—		1	Г				Г	Г		—				1	12					11.11	17			5	말날	119		開		+++
⊢	-	 	┝	+	+			-	+	+	Η.	┢─	-	Н		<u>+</u>	-	R		F-6	-	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	- 11	h f	-1		<u> </u>	-16		-1-11	-+-	+++
⊢		⊢	1	⊢	\square			1	1						4	-	-			L)	Ľ.		#	4	_ p #	(i f	14	4.1	2	Ыľ	1	ЦЦ.
I T		1	1	E	17				1	1				1		Í	11	-		1 10			М	1 T	4					Ы	T	
—	1	1	t	t	1				1	T	1		-		1	t		Ú÷.	-	-		199 ⁴	-	1		1	-	B	1	- - - -	-	+#
┣		⊢	+	+	1	1	H	+	+	+	H	-	-	Н			1 74						-		20			<u>.</u>	-	-+	+	+++
		L	L				<u> </u>	L	L						1		LË										E					LH
			Γ	Γ					Γ						1	Γ	LE.	C.	Ľ					H	मि			10	IT	T	Т	TT
F	177	t	t	+	+	Η	H	+	t	H	-	-		Н	17	t	i -	H		-	f				Tip	H-i	-fi-	10		-+-	+	+H
┣	÷.	-	⊢	₊	 		4		+	⊢	H -			\vdash	2	-	L.	4,	H.		, H	開口		ЦĦ	117	4.4		ŧļ₽	r,	_+		111
L		L	L						L						ź	ET.	T	Ú	3	Ľ	ΞŤ.	= LI	1	ПÉ		<u>t</u>	1E	11 P	5	_1		111
			17				-		1							И	1	11				15		E	LLC.					-		TH
-	115	-								-	H-		Н.		2	112	F			\vdash		-	+	H	-	++	F	14	F	-+-	+-	+++
L	5	4.5	2	É,	1,	٩		\square	.	Ļ.,	۲.	17				벣	ł	ų,	-			الأبيب	+	l ant	<u></u>	+		11	Ц	_	1	+11
	Ε.	12	11	\mathbb{R}^{2}			\square	ł	2		22	13		131	÷.	114	A.			1 1				I FI			- 81			- 1	1	11
	1.7		**	17	. 7	-		П			.	152	A,			И	1.7		22	-		11 12 1	2.1	112.15		-1	1.15	Hal-		7 1	1.1	Ħ
_			1		H		-	Н					-	Ľ.		#7					-	an san	- 44.	1.0	1.1.1					Щ4-	4	+++
15		42	44	22					2	2	1	-		Π.	144			1							- 644		P #	1				
		i i	1						ŧ.								ĽН			1			11		100		- 11	1 12		-		TH
	Γ.		—	1					1		Г	-				1	1.			58	- 1	-	ЪŤ	雔	-	17 1	퀴븙		H.	in the		++
-		┣	┢━	+				+	⊢	┝	+	-	-	Η	4	-				Ηř			-				ار با	14	μ÷	++	┝┼─	+#
		L														-	÷							<u>"-</u> †				1.1	- I	-Fil-		
		Г	Г	T			Π	1		Ι					Π	Γ		7	-	Ē			1			L ii	П			ш		тн
		⊢	┢╌	+	t				+						H	t-			<u> </u>	"			-	H	-	+-9	┖╼┿╍	P	H	-	+	+++
_		┡	┝	┢			-	Ч	ΥĽ.	22	- 693	22	77		1	100	1	2578	20	100	4	2010027	2		t in	1	_	-	-1	-+	_	14
			L	L_				1.1	8	2	Ŀ							ć i		łł						1 1	니콜	1	唐 /	.1	1	111
			Γ					Π	1								H	5.					П	Ħ			Ŧ					тп
	Ľ.	-	⊢	+				Н	-						1		19	н			- j		+ -		; E			Ξī		-+	+	+#
-	÷.,			+				Ц	4		L	_			-						- 1									_+		111
77	. 11	22	12		t d	ل و						L	9-	h 1		И	II.	4			I		T		11-1	1	1	11.		_1	1	1 11
		2.0	#	H I					2						1			-			Ľ						19					TH
		T C	17				늘	+ +	1	+	1		Ŧ	4		+	-	-	1	+	-				- 111	+ +	+	17	⊢	-+	-+-	+++
┝─┥	1	L.	U.	- 5	- 93	.		+	8	┢╌┥	1.5	121		ŧ.		4	i geri		í.				+		<u></u>	+	_	Ш		_		44
L		83			9			L				3	Ŧ				Ю		Ľ	Lł				Ш	T		- 11	L.		_ [THE
22		μ.	F	10	7		1		1				μ.	1	П	T		-	25	- 1		17 1 2 17	Ш.	- FR	0	1.11	T P	100	-	rtt	11	TH
		-	F	f		F		ť	F -	F	1	F	=	F	֠	←	1			1	-			1		£4			19	44		++
-	.	-	1				1	_	I	⊢					14	⊢	1		-	ι.	Ļļ	LEGT 1	- 15	L C		- 1	<u>₩</u>	4.4		цĻ		111
L			L	L				L	L							1	1					- Dest	1164	ЦŤ	1.00		si (r	12	11	μť	1	111
		1		Γ				T	Γ										22		-	-	3111	圕	1.	1	구 별	1.0) II	-	
H		-	+	+-	t	\vdash	H	+	+	H	t	h			1	1		-1	1		-				10	1	11	+.	77		+	++++
	سل	┣	-	1		Ц	.	1	1	\vdash					1	<u> </u>				12			11	F	Ē			+		±#	•	111
	\mathbf{r}	L	L						L	L	L.	L	L		LI	1				L.U				LI		+			H	الير	1	1117
		Г	1	1			1		L.		U,					L	L	1					T		H			P			+-	$\uparrow \uparrow \uparrow$
	-	-	+	+	+	\vdash	H	Η	1	1	H ^S	14	72	22	-	Ŧ-	12		е		÷		4	-I	E T			-		-+	+	+++
		Ļ	1	1			11		1	⊢	LL_					⊢	1	•				1.1	1		(15	7.2		_		111
		1	E		1]	1	r T			11	HF.		- "		ΓT	1	E	h	10		E C			lr∰	ųШ		10	4		T		ITT
_		-	ļ.,	1.			-		1	Г	Ē					1	T.	1	-	H	H			10			12	1	Ð İ	-+	+	+++
							51		-	⊢	Ŀ.	-		η,	1	Ì.		÷		22				1		1	-	- 15	i a c	-+	-+	+++
		12		H	4				1				14			Ľ	1	1					1					i e				
	£	10	17	. : :)			57	ſ٦	A			st		175	-		μ u	-			Ľ						1.4	1	- 1	T	T	TT
-		H.	÷	14	-			H	8	t	i î					t	⊢	1	~	+ +		111		+	- 11	+ +	-+-	11	+	-+	+	+++
		4	1	+-'	- 1	4	أبنا		2	1			цĽ	E	4	I	+	H	1			⊢⊮∔⊢	+	\vdash	- #1	+		44	┝─┥	-+		111
72		1		1					2		1				L1	1		LF	1	1			1	l ľ	11	1	1	ШT	ΙĪ	- F		117
	1			Г			1									Г		F				TET T	T		Ħ	+-1	-	11		-+	1	\uparrow
-	+	┢	+	+-	\vdash	\vdash	μų	+	┢	+	-		<u> </u>	Η	H	+	+	H	┢~	\vdash		<u>⊢ #</u> #+	+	┝─┼	-##	+	-+-	+++	⊢∔	-+	+	+++
		į.i.	L I	p.					t -	1					TT.	↓		L.	1				+	\square	18			11				111
	1	<u> </u>	۲ŧ	Ħ.	17		1	+	Ť					1	ш	1	17	1	1	1		ИТ		iΓ	H.	T T	i	AT	IT	T	T	111
	1	r -	Ē	П					Ι.							Ĺ		Ľ					1		11			#1			<u> </u>	\pm
	+		ΗĒ	H	E			E	F	-	-		-	•••	÷	-	<u></u>	H	-				+		- HI			H		-		F#F
	u i i i		í "	_		-	_	-	-		_	_	_	-		_	_	- 7	_	_	_	1041	-									_

•

-

.

.

Figure VI-9. CLL layout for A register.

VI-14



TO ALU





Figure VI-11. CLL Layout of B Register.

2. Read/Write Controller

The Read/Write Controller determines in which direction data is transferred to or from the 4x4 bit RAM.

Data is written to the memory on phi 2 and when the read/write line is high. Data must be available on the data input lines D0 - D3. Both the inverted and non-inverted data values are written into the memory cells. Data is read from the memory on phi 2 when the read/write line is low. Data is read only from the non-inverted data line of the memory cell. The schematic diagram for this circuit is shown in Figure VI-12.



Figure VI-12. Read/Write Controller schematic diagram.

A CLL layout of the Read/Write Controller is shown in Figure VI-13.



Figure VI-13. CLL layout of Read/Write Controller.

3. ALU Input Controller

The commands to the ALU, SO - S3 must be applied only after the data has been read from memory into the A and B registers (or just the A register - depending on the function to be performed). The input commands SO - S3 are passed into this circuit on phi 2 and when register select goes low, indicating that data is being read into the B register, passed along in the A register and eventually out of both registers, the command is passed into the ALU. The data that was input from the A and B registers (or just the A register - depending upon the command input) is processed. This circuit insures that only valid data is processed. The ALU has a built in level of transmission gates that clock data in on phi 1, therefore this circuit is buffered on its output to insure the charge signal is held until the following phi 1 clock pulse when the SO-S3 inputs are applied to the ALU. A schematic diagram of the ALU Input Controller is shown in Figure VI-14 and a CLL layout is shown in Figure VI-15.



Figure VI-14. ALU Input Controller schematic.

1		1	ł	1	1	Ł	1		1	1	1		1									1			ł	1
_					\mathbf{T}	t							T			T	T	-		1			Π		1	T
		-	┢──	╆╌	╆	┢				┢	┥		+	-		H	╢	_	-	╉			Н		┼──	╀
					1		_	****			73	277.	-				ł									
	T	1	1	1	1	ſ	<u>.</u>	:	the second	1			7 .			H	1		E	T			Π	-	Г	T
		4		+	_	⊢				┢	-	-	#			\mathbf{H}	#	_	1	謹	1.2	 		<u>. u</u>	<u></u>	ᡶ
		1	CT2		(T								11				1	. (ΪÌ	T	<u>v -</u>		Η	-7	<u> </u>	f
			H	ĒΥ		Ł		1.22	<u> </u>	144			t				Ŧ	Ĩ.	₿₿	1	F		Ħ	t	<u>+</u>	╈
	200		ΗE						Ι				1				#				E7				j –	
	2	100	1			Γ							11				Ŧ	4		Ŧ	H	1	Π	E		Г
_		1.041	1	811	1 442	∔	_			11-1	4		1			- 44	뫢	10	₽.#	+	4		Ц	-f	4	╋
		H.		ŦĦ						IF I			4				Ŧ	Ħ			Ħ			- E	1	
_			<u>.</u>	邗		╋	-		-	H			Ŧ				Ľ	F	F	-	=		H	[-	<u>+</u>	╈
110		L#		¥₩						H.		Мул	+	-	· ·		Ē	÷ .				जिन्द		_	4	
		ľ	1	Г		1		1. I	<u>i.</u>				6.5	- 1	- 23	<u></u>	F	: • ;	· .	Ŧ			۲		-	Г
			1	1	1				-					- {					.	L					1	Ł
		Ē	24.	27.27	224		.8		8 X	H-			Ŧ				F		-	-1	Ì	H			⊞_	+
· · ·					-	+1	R	艶,	F	щ			H	-			÷	44	-	F		щ	H	H	E-	÷
				T	[17	L.		10.00	12	i,	11	Г	- 1			1		÷ 8				H	95		
				1		Ø	щ										E	112		E		1				T
_				<u> </u>		Ø		1		1	I	1					Ę			-						
			. -	+		н	16	b .	1.1		_		Ħ	_			k	Н		č	<u> </u>	114	4		再-	\mathbf{t}
			-	<u>(</u>	<u> </u>	61					-	<u> </u>	μĽ	4			£			÷		í.			μ£÷	Ŧ
						14	Н				ž						ŀ,	H	٠,			Ħ	ľ	e i i	1	
		-	†	+	1	Ľ.	1.6	- m		ι×.	1		1				Ľ	1.0		Ŧ					1	+
						Ħ	_6		XX			2.5					É	LE			. X	H:				
				1	<u> </u>	h	τĹ	È.	1000	1	Н		Ŧ.	~			7	Ш	b,	, P		Hť I			۳b	tr
			-			- [4	ŧ٨					14) 1	₽		1	-	ŧ.	Ħř		f	ving		1		<u>لنا</u>	+
		1	1	1	1	Ø	Ť		8	ŧ.	ľ		1				P	lt			3	1	ł		H	
		t –	+	+	+	巤		Π			-	-	1-	- 1			Ŕ	E		Ŧ			-		╋──	╋
									ğ X								E				N.				122-	+
				1	1	11		勧		1			П			·	T.	H		ľ				Ħ	TF	F
_		I				Ħ	tH						₽		فتضه		۴	Ш	Ľ	⊧					₽_	┢
				1		Ø	1	- 1	11	4	H	÷.	8				Ł	I C		4		4	i.		tl.	

Figure VI-15. ALU Input Controller CLL layout.

4. 4x4 Bit Random Access Memory (RAM)

The memory is composed of the vertical and horizontal iterations of a basic six-transistor static memory cell that is shown in Figure VI-16.



Figure VI-16. Six-transistor memory cell.

The operation of the cell is based on back-to-back inverters connected in a flip-flop configuration. Whenever data is to be written, the row select is set high and the data on the data line and the data bar line is applied to their respective inputs of the flip-flop. This data remains available on the flip-flop indefinitely, may be read at any time and does not require refresh (i.e. a static memory cell).

The CLL layout for the static memory cell is shown in Figure VI-17. This cell is iterated sixteen times to generate the 4x4 bit memory.



Figure VI-17. Single Bit Memory Cell CLL layout.

5. 2x4 Decoder

1

The 2x4 memory row decoder provides the row select signal to the row of memory cells that is read from or written to. The schematic diagram of the circuit is shown in Figure VI-18 and a CLL layout in Figure VI-19.



-





Figure VI-19. CLL layout of 2x4 Decoder.

C. ALU/Memory Bitslice Precharge

The ALU/Memory was precharged in accordance with the previously presented precharge schemes. For this project, the chip elements that were precharged were the one bitslice of the ALU, the 2x4 row decoder, the A and B registers and the 4x4 bit RAM.

To effectively isolate the interconnections between chip elements and to precharge selected lines efficiently, transmission gates were used on the inputs and outputs of each chip element. Signals are passed on alternating phases of a two phase clock.

The bitslice was precharged as outlined in Secton B and the decoder was precharged as shown in Figure VI-20.



Figure VI-20. Schematic of precharged 2x4 Decoder

	A 17MAG N W C A A N MONTO Y A A A N IN A NOVA NA A A A A A A A A A A A A A A A A A
	2 (24-2 § (26 64-2) (26
「「「「「「「」」」「「「」」」「「「」」「「」」「「」」「「」」「「」」「	
444 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	

Figure VI-21 shows the CLL layout of the decoder circuit.

Figure VI-21. 2x4 Precharged Decoder Circuit.

The only change to the basic RAM, in addition to one transmission gate for each of the four data lines, was the addition of inverters on the data output lines to the registers to isolate the precharge on the output lines. If this isolation did not occur, the data lines, of the

individual flip-flop memory cells, would change to all high states if the data lines were directly precharged. The A and B registers were designed to compensate for the additional inverter on the output lines from the memory.

The corresponding schematic diagram and CLL layout for the precharged memory are provided in Figures VI-22 and VI-23 respectively.







The registers were precharged on their outputs only using one transmisiion gate for each output line. The schematic diagrams and CLL layouts for each register are presented in Figures VI-24 through VI-26.



To ALU



Figure VI-24. Precharged A & B register schematic.



Figure 25. CLL layout of Precharged B register.

VI-25

┿╋╬┿┿┿┿╬ ╈╋╋┿┥┿╫ ╡┼╣┼ TT ┼┼┠┼ ┼┼┨╢╴ T H 111 TT 面 TIT ĿШ ┶┽┼╆╫ TH H ╧<mark>╣╴┨╷╢╴┇╴╴╴╬┑╖╗╴╌╕╕╷┍╴</mark>╢╽╡╴╣ ╶╴┵╺┵┅╴╵╴┥╶╢┟╴┤╶╴╴╻┥┥╴┙ 1111/11 ┝┽╏╫┾┾╫╬┽┼╫╬╪╤╗ ┽┼┠╫┾┵╫╬┽┼╶┨╣╪╤╗ TH iff Ш Ī.III 圉 111 I TH t tt † † † † Ħ THE . ₩ҬҬШ 抽畫.

Figure VI-26. CLL layout of precharged A register.

VI-26

The Memory/ALU Bitslice design presented here demonstrated a practical application to at least a part of Sommars' ALU, offered a larger circuit in which to demonstrate the technique of precharging, and demonstrated the application of precharging to an existing circuit (ALU bitslice) and newly designed circuits (2x4 decoder and RAM). The decoder, RAM and the registers were selected as candidates for precharge since these elements are traditonally slow devices. Memory and other storage elements have slow access times and are in most need of speed-up technique application. The ALU input controller and read/write controller were composed primarily of transmission gates used as pass elements, and were not good candidates for precharge application.

To conclude this chapter a detailed schematic diagram of the Memory/ALU Bitslice chip circuit is shown in Figure VI-27.



7

. .

Figure VI-27 Detailed Memory/ALU bitslice circuit schematic.

VII. Simulation Results Analysis

i V V This chapter reviews the results of the simulations performed on the test circuit presented in Chapter VI, the newly designed circuit elements of the Memory/ALU Bitslice circuit and the first bitslice of the ALU previously designed by Sommars [3].

The intent of this simulation was to determine the validity of the precharging concept as a legitimate method of speeding up circuit operation, to demonstrate the use of precharging in both newly designed and already existing circuit elements, and, given a specific circuit to speed-up, how to most effectively precharge the circuit.

Experience was gained on an incremental basis as more simulations were performed on the different circuits.

Since the design of many of the circuit elements was in process and modifications to the bitslice circuit were being considered in conjunction with the performance of the initial test circuit simulations, the precharged circuit element designs and the precharged bitslice implement precharging in accordance with the results of those initial simulations.

Specific constraints were established for the simulations to evaluate circuit precharge performance and for standardizing the inputs and outputs of the circuits. These criteria are discussed in the following section.

A. Simulation Scenario Evaluation Criteria

Simulations were performed for each test circuit configuration, each newly designed Memory/ALU bitslice circuit element except the A shift register, and each bitslice section. Each circuit's inputs were selected to generate an output high-to-low transition for one simulation and an output low-to-high transition for the other simulation. This illustrated the effect of precharging on both high and low transitioning signal levels. Transitioning input waveforms were used to enhance delay analysis.

A 2 nsec precharge pulse width was selected since simulations on the individual gates demonstrated that they switched on the order of 2 nsec, and thus this was considered a valid precharge pulse width based on the expected operating speed of the circuits under simulation. The precharge pulse was applied to selected nodes without isolating the nodes from the rest of the circuit. There is an advantage gained by applying precharge in this fashion to the active circuit. Since the precharge occurs when the circuit becomes active, the requirement to isolate the individual gates/circuit elements when nodes are precharged is reduced. In this way, most transmission gates required for isolation could be eliminated, thus saving space on the chip.

For purposes of simulation, the precharge pulse was delayed (from the start of the input signal sequences) by 4

nsec and "straddled" the 5 nsec transition point of the input waveforms. This insured that the precharge was available when the circuit became active. It was not intended to simulate an actual phase 1 signal of a two-phase clock.

The varying input levels insured a varying output that illustrated the effects of high-to-low and low-to-high transition gate delays for purposes of analysis. The input signals sequences are shown in Figure VII-1. The first signal level (1 or 0) began at 0 nsec and lasted for 5 nsec where it transitioned to either high or low, lasted 10 nsec and then transitioned back to its original value for 25 nsec, for a total signal time of 40 nsec. The same time frames for the precharge and input pulses were used for each simulated circuit configuration to permit direct comparison of results.



Figure VII-1. Test circuit input waveforms.

All input waveforms were "ideal" with no delays (instanteous level transitions). This simplified the delay analysis. Circuit delays for each simulated circuit were determined with respect to the ideal input waveforms as they transition from high-to-low or low-to-high at the 5 nsec point. For this analysis, all the circuit delays presented are measured with respect to the 5 nsec ideal transition point on the time scale.

For purposes of delay evaluation, "valid" high and low signal level thresholds were established. For a signal that was transitioning from low-to-high, the delay was determined as the time required for the signal to reach 3.75 volts DC measured from the 5 nsec ideal time point on the time axis of the SPICE waveform plot (0 to 75% V output signal max swing). For a signal that was transitioning from low-tohigh, the delay was determined as the time required for the signal to reach 1.25 volts DC measured from the 5 nsec point on the time axis of the SPICE waveform plot (100% to 25% V output signal swing). max

SPICE simulation data are provided in Appendix A, B and C for the test circuit configurations, the Memory/ALU Bitslice circuit elements and the sections of the bitslice respectively.

Also included in the appendices, for each simulated circuit configuration, is a numbered node cifplot and SPICE/mextra node cross reference provided to aid the reader

in following this analysis and for future duplication of the simulations presented in this thesis.

B. General Observations

Some general observations can be made from review of the SPICE data.

The 2 nsec precharge pulse is clearly visible on many of the waveform plots as a high transitioning spike around the 5 nsec point on the time axis. In many cases the pulse fails to cause the circuit to switch because the circuit was active (beginning to perform its function, based on the input), at the time the pulse was applied. The output waveforms display the effect of the precharge, both when it is suppressed (circuit trying to reach 0 volts) and also when the waveform is transitioning high after the precharge has been applied.

In order to provide an adequate precharge pulse to the selected precharge nodes, oversized transmission gates were used to provide enough current to precharge the nodes. This was accomplished by increasing the gate widths of the transmission gate PMOS and NMOS devices.

Simulations were performed using different widths for the transmission gates. The cases where the widths were increased shows a considerable increase in the strength of the precharge pulse as it was applied to the different nodes. Simulations were performed using gate widths two, three and four times the width of the basic transmission gates designed by Sommars.

The test circuit simulations were performed with transmission gates four times the basic gate width, and the chip elements and bitslice sections were simulated using gate widths of one, two and three times the basic gate widths. These simulations were performed to demonstrate that when a sufficiently large current pulse could be provided, the precharge method did indeed precharge the specific nodes as desired. However, due to the capacitance of interconnect wiring, addition of precharge circuitry can slow the circuit down. This effect can be eliminated if the circuitry is laid out with short wiring runs and consideration is given to insuring sufficiently large transistor sizes within the precharge circuit.

C. Test Circuit Simulation

•

The test circuit described in Chapter VI was composed of a single nand gate and two inverters in series. This test circuit was implemented with the CMOS/SOS gates available in Sommars' library. The test circuit simulations were performed to demonstrate the basic technique of precharging, to determine its impact on positive and negative signal transitions, and from these results determine a methodology for applying precharging to circuits in general. The results of these initial simulations were applied to the Memory/ALU Bitslice circuit element and Bitslice section simulations. Twelve separate simulations were performed on the test circuit. Two initial simulations

(one for a high-tolow signal transiton and the other for a low-to-high signal transition) were run on the circuit, before any precharging techniques were applied, to establish a baseline to which the other simulation results could be compared. Three nodes were available for precharge application, the output of the nand gate, the output of the first inverter and the output of the second inverter. The remaining ten simulations selectively precharged these outputs individually or in combinations.

For the test circuit, the inputs to the nand gate were selected to provide a low-going pulse, 101, at the output (identical nand inputs of 010) for the first simulation and to provide a high going pulse 010 at the output (identical nand inputs of 101) for the second simulation. The outputs from the nand and both inverters were monitored and plotted for each SPICE simulation. On each waveform plot, the nand output is depicted by the equal sign (=) waveform, the first inverter output by the plus sign (+) waveform and the second inverter output by the asterisk (*) waveform. Of primary interest is the second inverter output waveform. Since this is the final circuit output it suffers from the total circuit delay generated by the circuit and is the signal provided to subsequent circuits or circuit elements in a more complex circuit configuration.

1. Basic Test Circuit

The SPICE simulation data is provided on pages A-3 through A-7 of Appendix A.

The delays associated with the input signals' first transition for this circuit simulation are presented in Table VII-1.

Table VII-1

Basic Test Circuit Delays

Nand Inputs	<u>Nand</u> Output (nsec)	<u>First Inverter</u> <u>Output</u> (nsec)	<u>Second</u> <u>Inverter</u> <u>Output</u> (nsec)
010	2.5	3.0	4.0
101	1.5	2.5	3.0

2. Precharging Nand Output

The delaytimes presented in Table VII-2 were determined from the SPICE waveform data in Appendix A, pages A-8 through A-12.

Table VII-2

	<u>Test</u>	<u>Circui</u>	<u>t</u> <u>Delays</u>	<u>with</u>	Precharged	Nand Out	tput
Nand	Inpu	<u>uts</u> 0 7	<u>Nand</u> utput nsec)	First Ou (nse	t <u>Inverter</u> itput c)	<u>Second</u> Out (ne	Inverter tput sec)
	010		5.5	(6.0	7.	.0
	101		0.5		1.5	2	.0

Results

The nand output high-to-low transition delay is increased, which substantially delays the transition for each subsequent gate output. The nand output low-to-high transition delay is reduced by 1.0 nsec, which causes an earlier transition in each subsequent gate, when compared to the basic test circuit.

3. Precharging First Inverter Output

The SPICE data for this circuit simulation is provided in Appendix A, pages A-13 through A-17. The delay times of interest to this analysis are presented in Table VII-3.

Table VII-3

	Circuit	<u>Delays</u> -	Precharged First	Inverter Output
<u>Nand</u>	Inputs	<u>Nand</u> Output (nsec)	<u>First Inverter</u> Output (nsec)	<u>Second</u> <u>Inverter</u> <u>Output</u> (nsec)
	010	3.0	1.5	2.0
	101	2.0	3.5	4.5

Results

The nand output high-to-low transition delay is increased while the transition times are reduced for both inverter outputs. The nand output low-to-high transition delay and each subsequent gate output transition times are all increased.

4. Precharging Second Inverter Output

The SPICE data for this circuit simulation is provided in Appendix A, pages A-18 through A-22. The delay times for the circuit are presented in Table VII-4.

Table VII-4

<u>Circuit</u> <u>De</u>	<u>lays - Pr</u>	echarged Second	Inverter Output
Nand Inputs	Nand Output (nsec)	<u>First Inverter</u> <u>Output</u> (nsec)	<u>Second Inverter</u> Output (nsec)
010	2.5	3.5	9.0
101	1.5	2.5	5.5

Results

For the nand output, high-to-low transition, the first inverter output shows a 0.5 nsec delay increase, while the second inverter output delay increases by 5.0 nsec, as compared to the basic test circuit.

For the nand output low-to-high transition delay, the nand and first inverter outputs occur at the same time as for the basic test circuit, however the second inverter output transition time is increased by 2.5 nsec.

Examination of the extracted circuit parameters, used in the SPICE simulation, shows that the addition of the precharge circuitry to the output of the second inverter increased the load capacitance by a factor of seven, thus slowing down the transitions greatly.

5. Precharging Nand and First Inverter Outputs

The SPICE data for this simulation is provided in Appendix A, pages A-23 through A-27. The circuit delays are presented in Table VII-5.

Table VII-5

Circuit Delays	- Precha	rged Nand and I	First Inverter Outputs
Nand Inputs	Nand Output (nsec)	<u>First Inverter</u> Output (nsec)	<u>Second</u> <u>Inverter</u> <u>Output</u> (nsec)
010	5.5	6.5	7.5
101	0.5	3.0	4.0

Results

The precharge increases the delay for the nand output high-to-low transitioning signal which then degrades the transition time for both subsequent inverters. Although a 1.0 nsec delay reduction is achieved at the nand output for its low-to-high transitioning signal, both the first and second inverter output delays increase by 0.5 and 1.0 nsec respectively, due to the detrimental effect of the precharge on the first inverter output high-to-low transition.

6. Precharging Nand and Second Inverter Outputs

The SPICE data for this circuit simulation is provided in Appendix A, pages A-28 through A-32. The circuit delays are presented in Table VII-6.

<u>Table VII-6</u>

Precha	arged Nand	and Second Inve	rter Output
Nand Inputs	<u>Nand</u> Output (nsec)	First Inverter Output (nsec)	Second Inverter Output (nsec)
010	5.5	6.0	11.5
101	0.5	1.5	3.5

Results

The nand output high-to-low transition delay and hence the subsequent inverter transition times are significantly increased by the application of precharge. The nand output low-to-high transition delay is reduced and hence the first inverter outputs also. However, in spite of the precharge signal, a 0.5 nsec delay increase is evidenced at the output of the second inverter due to the previously mentioned large capacitance load created at the circuit output by the particular precharge interconnection.

Overall Results

As already discussed, the output of the second inverter is of primary interest. If its delay can be reduced, by reducing the delays of the gates affecting that node, then the overall speed of the circuit can be increased.

The second inverter output delays are presented again together in Table VII-7. The respective increases or decreases in delay times are shown as percentages.
Table VII-7

Total Circuit Delays						
Nand Input	Total <u>Circuit</u> <u>Delay</u> (nsec)	Node Precharged	Delay Change			
010	4.0	none	baseline			
101	3.0	none	(no change) baseline (no cahnge)			
010	7.0	NAND	+75.0%			
101	2.0	NAND	-33.3%			
010	2.0	INV1	-50.0%			
101	4.5	INV1	+50.0%			
010	9.0	INV2	+125.0%			
101	5.5	INV2	+83.3%			
010	7.5	NAND, INV1	+87.5%			
101	4.0	NAND, INV1	+33.3%			
010	11.5	NAND, INV2	+187.5%			
101	3.5	NAND, INV2	+16.7%			

D. <u>Memory/ALU</u> Bitslice Circuit Element Simulation

Three elements of the Memory/ALU Bitslice circuit were simulated; the decoder, the B register, and a single static RAM cell. The data from the SPICE simulations for each of these elements is provided in Appendix B.

1. 2x4 Decoder Simulation

Three decoder configurations were simulated, the basic decoder to establish a baseline for comparison, the decoder with precharge of each nand-inverter interconnect line, and the decoder with precharge of each output line. Three output lines were monitored for each

simulation. On the SPICE waveforms, the 00 selected line is represented by the asterisk (*), the 11 selected output line is represented by the plus (+) sign, and the 01 selected output line is represented by the equal (=) sign.

The inputs to the decoder are identical 010 pulses. Onlyone simulation was performed for each configuration, since the 00 and 11 select lines are monitiored and the 010 inputs cause a high-to-low and a low-to-high transiton on the same SPICE output plot.

The 010 input waveform is of the same format as presented in the sample waveforms in Figure VII-1.

A 010 input to each decoder input corresponds to a selection of the 00 output line, then the 11 output line, and then the 00 output line again.

For each simulation, the 01 and 10 output line should remain constantly low for all the simulations since the input never causes either one to be selected. For the decoder simulations it can be seen that the 01 output line was constantly low, indicating that the decoder operated correctly for non-selected lines, based on the decoder input, since the line was never selected (never went high).

For the precharge configurations, simulations were performed for precharge transmission gate device widths using the basic widths, and two and three times the basic widths. The intent is to show the effect of a deficiency of current (narrow width transmission gate) and increasingly larger current precharge pulses (wider gate transistors) on

the effectiveness of the precharge technique. In the tables, results from circuits using transmission gates transistors with these dimensions are represented by x1 (times 1), x2 (times 2) and x3 (times three) respectively.

a. <u>Basic Decoder Simulation</u> (Refer to Figures B-6)

To determine decoder delay times, the analysis was performed by first observing the transition of the output lines as the two decoder inputs changed from 00 to 11 (based on an input sequence of 010 at both inputs). After the decoder input changes to 11 at the 5 nsec point, the 11 selected output line reaches the 3.75 volt minimum value at the 9.0 nsec point. The 00 output line achieves the 1.25 volt, non-selected condition at the 9.0 nsec point. This data is tabulated in Table VII-8.

Table VII-8

Basic Decoder Delays

Inputs	<u>High-to-Low</u> Delay	Low-to-High Delay		
010	4.5 nsec	4.0 nsec		

b. <u>Precharged</u> <u>Decoder</u> (Refer to Figures B-4 through B-10)

Two different precharge configurations were simulated to determine the best approach to precharging the decoder. These simulation results demonstrate the need to carefuly select the nodes to be precharged within individual circuits and circuit elements.

1. <u>Precharged Decoder (Internal nodes)</u> (Refer to Figures B-4 through B-6)

The application of precharge at the internal node between the nand gates and the inverters causes the outputs of all the select lines to be driven low after the precharge. This accounts for the output node high-to-low transition delay reduction for the x2 and x3 gate widths data presented in Table VII-9. The original transmission gate transistor size does not produce an appreciable reduction in transition delay. The low-to-high output transition is degraded due to the precharge causing the outputs to be pulled low during the first part of the input signal.

Table VII-9

Precharged	<u>Deco</u>	oder (Interna.	l <u>node)</u> I	Delays	
Input	Deco	oder O	utput Ti	ransition	n <u>Dela</u> y	Z
	High-to-Low			Lot	w-to-H	igh
	<u>x1</u>	<u>x2</u>	<u>x3</u>	<u>x1</u>	<u>x2</u>	<u>x3</u>
010	4.0	3.5	2.5	7.5	7.5	8.0

2. <u>Precharged Decoder (Output Nodes)</u> (Refer to Figures B-8 through B-10)

This precharge configuration degrades the performance of the high-to-low transitioning waveform, however, this is not a serious drawback, since it is a "high" select decoder. The low-to-high transition delay is reduced by 1.5 nsec in the best case, and by 0.5 nsec in the worst case, when compared to the basic decoder delay

results. The results are presented in Table VII-10.

Table VII-10

Prech	arged De	ecoder ((Jutput 1	Nodes)	Delay	
Input	High	-to-Low Decoder	<u>Output</u>	Low Transi	-to-Hic tion <u>D</u> e	gh elay
	<u>x1</u>	<u>x2</u>	<u>x3</u>	<u>x1</u>	<u>x2</u>	<u>x3</u>
010	5.0	5.0	5.0	4.0	3.5	3.0

2.<u>B</u> <u>Register</u> <u>Simulation</u> (Refer to Figures B-11 through B-16)

a. <u>Basic B</u> <u>Register</u> (Refer to Figure B-12)

For the basic B register, the output lines float until 5.5 nsec after the input signal transitions (input finally propagates to output), then gradually approach their final values. However, the monitored high signal output line fails to reach the minimum 3.75 volt level and the low signal output line fails to reach the 1.25 volt level. Instead, the high driven output reaches a maximum value of 3.6 volts at 11.5 nsec after the input signal transition and the low driven output reaches a minimum of 1.35 volts at 11.5 nsec after the input signal transitions. The B register fails to operate correctly, but the effective signal delays (time to reach maximum/minimum values) will be used for comparison to the precharged register configuration delays. These effective

delays are tabulated in Table VII-11.

Table VII-11

<u>B</u> Register Delays

Input	<u>High-to-Low</u> Delay	<u>Low-to-High</u> <u>Delay</u>		
1	11.5 nsec	~~~~~		
0		11.5 nsec		

b. <u>Precharged B Register</u> (Refer to Figures B-14 through B16)

The circuit fails to operate correctly after the precharge is applied, however the effecs of the precharge are clearly visible on the waveform plot. the precharge favors the low-to-high transitioning signal. The delay (actual and effective) results are presented in Table VII-12.

Table VII-12

Input	<u>Precharged</u> <u>B</u> <u>Regi</u>	<u>Delays</u> Delay	
	<u>x1</u>	<u>x2</u>	<u>x3</u>
1	12.0	11.5	7.5
0	12.0	12.0	12.0

3. <u>RAM Cell Simulation</u> (Refer to Figures B-17 through B-26)

The ram cell was not simulated directly, since simulation of the flip-flop itself would have been necessary and would have unnecessarily complicated the precharge simulation. The precharge of the RAM cell was set up as a precharge of the "data" line and was designed to take place only during a read of the data, when data is read from the "data" line of the memory cell. With cascaded cells, making up the entire 4x4 memory, the data lines are common to all the RAM cells. To precharge the data line on a read of the memory, isolation of the precharged node is required to prevent the indescriminate writing of a high data signal into the flip-flop memory cells, due to the precharge, when a specific row of the memory was selected. Therefore isolation inverters were added to the data lines of the RAM. These inverters limit the precharge to only the data line outputs.

The simulation of the memory was limited to a single cell with an isolation inverter attached to its data line output. The simulation of the cell entailed application of a signal to the inverter input and a precharge of the inverter output. This represented a selection of the cell on a memory read, the flip-flop data being output to the data line andthedata signal propagating to the isolation inverter output. Therefore, the only delay being simulated was that across the isolation inverter.

The 4x4 memory supplies data to the A and B registers. Since the isolation inverters inverted the data signal from the flip-flop, memory cells, the registers were designed to account for this inversion.

For the precharge configurations, the precharge

simulations were performed for the basic device width transmission gate, and widths two and three times the basic width. In the tables, these widths are represented by x1, x2 and x3 respectively.

a. <u>Basic RAM Cell</u> (Refer to Figure B-17 through B-19)

A 010 waveform, as presented in Figure VII-1, was applied to the data line of the RAM cell driving a 101 output waveform at the output of the inverter. Then a 101 input was applied driving a 010 at the output of the inverter. The delays are presented in Table VII-13.

TABLE VII-13

Bas: Input	ic RAM	<u>Cell</u>	<u>Delays</u> Dela	ay
010			5.5	nsec
101			5.5	nsec

b. <u>Precharged RAM Cell</u> (Refer to Figures B-20 and B-26)

When the output transitions from high-to-low, a delay increase results when the precharge is applied. When the output transitions from low-to-high, the transition delay is reduced by 0.5, 1.5 and 2.5 nsec respectively for the original, 2x and 3x widths of the transmission gate devices. Again the effectiveness of precharging is increased when the transmission gate transistors are large enough to supply an appreciable amount of current.

	Precharged RA	<u>M Cell Delays</u> Delay	
<u>Input</u>	<u>x1</u>	<u>x2</u>	<u>x3</u>
010	6.0	6.0	7.0
101	5.0	4.0	3.5

TABLE VII-14

These simulations indicate the following:

- Precharging can enhance the operation of a circuit, i.e. more clearly define high and low signal levels

- The results of precharging must be carefully analyzed to determine if it has any effect on the circuit to which it was applied, i.e. the basic RAM cell circuit displayed no positive or negative effects of precharging for the original size transmission gate, but did show an effect as the transmission gate transistors were increased in size, allowing a greater precharge current pulse.

- Selection of the nodes to which precharge is applied is critical, i.e. as displayed by the two different decoder precharge configurations. One circuit displayed a marked enhancement in operation while the other precharged circuit displyed degraded operation due to the difference in the nodes that were precharged.

D. ALU Bitslice Simulation

Simulation of the entire bitslice was attempted but failed. The simulator produced error statements. The reason for the failure was not determined. However, incrementally increased circuit size simulations that were performed by adding portions of the bitslice circuit (individual gates), indicated that SPICE was not capable of simulating a circuit that was comprised of more than thirteen gates. As an alternative simulation approach, the bitslice was divided into three separate sections. The bitslice design provided three independent sections that were tested for high and low transitioning output waveforms.

Two different precharge configurations were simulated on the bitslice sections. In the first, all the nodes selected were internal to the individual sections. this included four nodes in the first section, two nodes in the second section and four nodes in the third section. Τn the second simulation, the precharged nodes were the respective outputs of each section. Thus only two nodes in the first section, one node in the second section, and one node in the third section were precharged. Simulations using the basic transmission gate transistor width, a transmission gate with a 2x transistor width and a 3xtransistor width were perfoormedThe SPICE simulation data are provided in Appendix C. Numbered node cifplots and node cross reference list are also provided in the appendix. For the bitslice section simulations, the input waveforms are of

the same format as presented in Figure VII-1 and the 3.75 volt and 1.25 volt levels apply to high and low transitioning output signals.

Simulation of the sections of the bitslice completes the basic testing of the precharge concept, since the test circuit was a series combination of gates and the bitslice sections offer parallel combinations of the gates.

To clearly follow this analysis, the reader will need to refer to the appendix.

1. First Section Simulation

The first section of the bitslice requires eight inputs to operate:

A -	• operator
в -	- operator
PHI1 -	 Phase 1 Transmission gate pass signal
PHI1BAR -	- Inverse of PHI1
so -> ss -	- ALU Control Inputs

For these simulations these inputs had the following values:

Table VII-15

Section 1 Simu:	lat	ion	Input	Values
A	-	010/	101	
В	-	000		
PHI1	-	111		
PHI1BAR	-	000		
S0	-	000		

S1 - 000
S2 - 111
S3 - 111

Two simulations were performed, one for A = 101 and the other with A = 010. The ALU control signals SO-S3 caused these inputs to produce outputs of 010 and 101 (the inverse of whatever A was) respectively. The same outputs existed at each of the two section 1 output nodes. Although the outputs from each node were logically the same, they were not identical since one output came from a two-input nand gate and one came from a three-input nand gate. The two different input sequences for A insured both high and low transitioning waveform output, the two-input nand output is represented by the asterisk (*) waveform and the three-input nand output is represented by the plus sign (+) waveform.

The A input did not conform exactly to the waveforms of Figure VII-1 for these simulations. Instead of waiting for 5 nsec for a change from high-to-low or low-to-high, the signal transitions at 0 nsec on the time axis. The input was modified in this manner since initial SPICE waveform outputs indicated that the delays associated with the first section caused an approximate 5 nsec delay before a section one output waveform transition and the output waveform exceeded the 40 nsec total simulation time before reaching equilibrium when the standard 5 nsec delay waveform was used

as input. To encompass the entire waveform the inputs had to be provided at an earlier timeframe during the simulation. The precharge pulse was maintained at the same time frame (4 to 6 nsec). The standard precharge pulse was maintained since the 5 nsec internal circuit delay of the first section permitted the precharge pulse to coincide directly with the output waveform transition when the 5 nsec delay on the inputs was eliminated.

The result of this earlier application of the A input is that the transitions from low-to-high or high-to-low occur slightly earlier than the 5 nsec time point which had been characteristic of the previous simulation waveforms.

For the precharge configurations, simulations were performed with the basic transmission gate, and the transmission gate with its device widths two and three times the widths of the basic transmission gate device widths. These are represented by x1, x2 and x3 respectively in the Data Table VII-16.

a. <u>Basic First Section</u> (Refer to Figures C-3 and C-4)

For a 010 input for A, the expected output is a 101 at both monitored output nodes. The 2-input nand gate output is high, then begins its transition to a low threshold. It achieves the 1.25 volt threshold after11.0 nsec. The 3-input nand output achieves the 1.25 volt threshold after 7.5 nsec.

For a 101 A input, a 010 output was expected. The 2-

input nand output reaches the 3.75 volt threshold after10.5 nsec and the 3-input nand output fails to reach the threshold and reaches a maximum of 2.55 volts after11.5 nsec.

b. First Section Precharged at Two Output Nodes (Refer to Figures C-7 through C-12)

For an A input of 010, both the two and the three input nand outputs achieve the 1.25 volt threshold for the simulations.

For a 101 A input, the two-input nand output achieves the 3.75 volt threshold for each simulation, however the three-input nand output only achieves the threshold for the simulation with transmission gate device widths three times the width of the basic transmission gate.

> c. <u>First Section Precharged at Four Internal Nodes</u> (Refer to Figures C-15 through C-20)

For an A input of 010, both outputs achieve the established 1.25 volt threshold.

For a 101 A input, the two-input nand achieves the 3.75 volt threshold, but the three-input nand still fails to achieve the threshold for all the simulations.

Table VII-16 summarizes the delays for the first section of the bitslice.

<u>Table VII-16</u>

<u>First Bitslic</u> <u>A Input Pre-</u>	<u>e Section Delays</u> Output Delays 2-Input Nand/3-Input Nand			
charge	<u>x1</u>	<u>x2</u>	<u>x3</u>	
010 none 6.0/2.5 101 none 5.5/6.5		·		
010 2 nodes 101 2 nodes	12.5/9.5 10.5/11.5	13.0/10.0 8.5/11.5	13.0/10.0 6.5/7.0	
010 4 nodes 101 4 nodes	10.5/7.5 13.0/11.5	9.0/6.5 13.0/11.5	8.5/7.5 13.5/11.5	

2. <u>Second Section Simulation</u>

The second bitslice section had only two inputs and was the least complex of the sections. One input was held at a constant high for each simulation while the other was varied to obtain a high and a low transitioning output waveform. The values are presented in Table VII-17. Refer to Figures C-21 and C-22 in Appendix C.

> <u>Table VII-17</u> <u>Section 2 Inputs</u> VIN1 - 111 VIN2 - 010/101

These simulation inputs conform to those specified in Figure VII-1.

a. <u>Basic Second Section</u> (Refer to Figures C-23 and C-24)

With input 010, the expected output was 101. The output reaches the 1.25 volt threshold after 6.0 nsec.

For input 101, the waveform reaches the 3.75 volt threshold after 6.5 nsec.

b. Second Section Precharged at One Output Node (Refer to Figures C-27 through C-32)

For inputs of 010 and 101 the outputs achieve the established 1.25 and 3.75 volt thresholds. The low-to-high delays are reduced and the high-to-low transition delays are increased.

Second Section Precharged at Two Internal Nodes (Refer to Figures C-35 through C-40) c.

For 010 and 101 inputs, the output waveforms also achieve the threshold voltages. However, the precharge reduces the delay of the high-to-low transition in this precharge configuration, since the precharge of both inputs to the two-input nand gate drives the output low after the precharge, thereby favoring the high-to-low transition and hindering the low-to-high transition.

Table VII-18 tabulates the delays each of these circuit configurations.

Second Bitslice Section Delays							
Input	Precharge		Outp	ut Delay	Z		
			<u>x1</u>	<u>x2</u>	<u>x3</u>		
010 101	none none	6.0 6.5		<u> </u>			
010 101	1 node 1 node		7.5 6.0	7.5 5.5	7.5 4.0		
010 101	2 nodes 2 nodes		6.0 9.5	4.5 9.5	3.5 9.5		

3. Third Section Simulation

The third bitslice section had three inputs; a carry bit, an ALU select signal and the input from the second bitslice section. The carry signal was kept constant at zero, the select signal was kept at a constant high and the third input was varied to produce a transitioning waveform at the circuit output. Table VII-19 summarizes the inputs. Refer to Figures C-41 and C-42 in Appendix C.

Table VII-19 Bitslice Third Section Inputs

Carry - 000 Select - 111 VIN1 - 010/101

The expected output is 101 for a VIN1 = 010 and 010 for VIN1 = 101. VIN1 conforms to the waveforms presented in Figure VII-1.

> a. <u>Basic Third Section</u> (Refer to Figures C-43 and C-44)

For a 010 input, the output reaches the 1.25 volt threshold after 7.0 nsec.

For a 101 input, the output reaches the 3.75 volt threshold after 6.5 nsec.

b. <u>Third Section Precharged at One Output</u> Node (Refer to Figures C-47 through C-52)

For a 010 input, the output delays are all increased by at least 1.5 nsec.

For a 101 input, the output delay is initially increased for the x1 case, but is reduced by 0.5 nsec in the

x3 case.

c. <u>Third Section Precharged at Four Internal Nodes</u> (Refer to Figures C-55 and C-60)

For the 010 input, all cases reflect an output delay increase of 3.5 nsec.

For an input of 101, the output delay is initially increased by 2.5 nsec for the x1 case, but is reduced by 1.5 nsec for the x3 case when enough current is supplied to the node during precharge.

Input	Ta Third Sect Precharge	<u>I-20</u> tslice Delays Output Delay			
			<u>x1</u>	<u>x2</u>	<u>x3</u>
010 101	none none	7.0 6.5			
010 101	1 node 1 node		8.5 8.5	8.5 7.5	9.0 6.0
010 101	4 nodes 4 nodes		10.0 9.0	10.5 7.0	10.5

E. Results Summary

The results of these simulations indicate the following:

- Precharge reduces the circuit delay when applied to nodes that are transitioning from a low to a high level.

- Precharge has a negative effect when applied to

nodes where the signal level is transitioning from high-tolow.

- Precharge can be applied to both series and parallel combinational circuits.

- In circuits, such as the logic high select decoder, the detremental effect of the precharge on the high-to-low transition is not a serious drawback, since the decoder operates only on high select (low-to-high) transitions. Circuits such as the decoder, and other "high only" type circuits, would benefit from precharge application.

- The effect of the precharge at a node, is dtermined by the amount of current that is applied to the node during the precharge.

- Nodes considered for precharge application must be carefully selected to prevent degradation of circuit operation.

VIII. Conclusions and Reccommendations

A. Conclusions

The intent of this thesis was to investigate and generate a single reference document of techniques to speedup electronic circuit operation, select a specific technique that could be applied to CMOS/SOS technology, apply the technique to existing and new CMOS/SOS circuit designs, and through simulation, determine if the technique actually speeds up the operation of the circuit to which it was applied, and subsequently determine a methodology for applying the technique to circuits in general.

When reviewing the circuit simulations, it should be emphasized that, overlapping the precharge pulse with the input signal transition is not an accurate reflection of the <u>standard</u> two-phase clocking used in CMOS circuit design at AFIT. This experimental approach insured that the precharge pulse would be available at the time of input signal transition and eased analysis and evaluation of the precharge technique.

1. Activities Performed

The following activities were performed during this thesis:

- Data was collected on known circuit speed-up techniques.

- Precharging (a specific speed-up technique) was

evaluated to determine if it applied to CMOS/SOS circuits and if it increased circuit operating speed, since it was the most commonly used method in the literature for speeding up circuits.

- To expand the AFIT CMOS/SOS technology base, a chip circuit was designed, using Sommars' CMOS/SOS library, that made use of the first bitslice of his ALU and required the design and development of additional circuit elements.

- Precharging was applied to the bitslice, and some of the newly designed circuit elements.

- SPICE circuit simulation was performed on a test circuit, the bitslice and the newly designed circuit elements.

- Through the simulations, precharging was applied to indivdual circuits and the effects of the precharge were determined.

2. Circuit Simulation Conclusions

The following conclusions were drawn from the circuit simulations performed:

- Precharging reduced the delays of signals that were transitioning from low to high.

- Increased output delay occurs when consecutive levels of inversion, e.g. the outputs of two series

VIII-2

inverters, are both precharged.

Ì

- Precharge can increase the signal delay at a node when the node is subsequently driven low by circuit function.

- Poor selection of candidate precharge nodes can have a detremental effect on circuit operating speed.

- It is difficult to determine which nodes within a circuit to precharge to obtain the maximum benefits of precharge application.

- The results of precharging must be carefully analyzed to determine the effect on the circuit to which it was applied. The addition of precharge circuitry increases the capacitance of the nodes to which the precharge is applied due to the capacitance of the interconnect wiring. To insure the effectiveness of the precharge, a sufficient amount of current must be applied to the selected nodes to overcome this increase in capacitance.

- Since it is difficult to determine which precharge approach to use, either precharging internal nodes or by charging only circuit outputs, then it is essential that circuits under consideration for precharge application be simulated using SPICE or some other method to determine the optimum approach.

VIII-3

3. Suggested Precharge Approach

The following procedure is recommended if a circuit is being considered as a subject for precharge application:

- Select candidate precharge nodes.

- Through signal analysis, determine the signal levels expected at the candidate nodes during normal circuit operation.

- If signals are predominantly high at specific nodes, then these nodes should be considered for precharge. An excellent example of this is the high select 2x4 decoder designed during this thesis.

- If neither high nor low levels dominate, then perform a simulation to determine the effect of precharge and apply precharge in accordance with the simulation results.

- Implement one transmission gate for each node to be precharged and determine the best place to locate the gates, to save maximum chip area.

4. Special Problems Encountered

In addition to these precharge results, several problems were encountered with some of the tools used to prepare the SPICE simulation files. - SPICE fails to simulate a circuit consisting of more than thirteen gates.

- Available SPICE reference documentation is not adequate, especially in the preparation of input files.

- The extractor, Mextra, fails to correctly calculate the capacitance of output nodes that are "floating poly", and some other node capacitances calculated by Mextra are not obvious when the circuit layout is examined.

B. Recommendations

The following are recommendations for future research on topics related to this thesis:

- Determine why the extractor, Mextra, fails to calculate correct capacitances on outputs when a "floating" polysilicon line is connected and research the algorithm for other node capacitances to verify correct parameters for SPICE analyses.

- Investigate the operation of SPICE and determine its limitations for circuit simulation, specifically, why it simulates only a limited number of CMOS gates

- Since only a limited analysis was performed during this thesis, expand the analysis and apply precharge to more advanced/complex circuits

- Investigate delays in large capacitance lines

- Investigate the impact of implementing the output precharge circuit described in chapter 4, which applies to delay reduction in high capacitance lines

- Simulate the entire bitslice circuit, then simulate Sommars' entire ALU verifying its logical correctness

- Complete the chip design and submit the Memory/ALU Bitslice chip fabrication specifications, and test the operation of the chip after manufacture

- Fabricate portions of the bitslice that were enhanced by precharge and determine the impact of precharge on an actual simple circuit

- Fabricate the output precharged decoder circuit and the basic circuit and determine the effect of precharging on the circuit through physical performance evaluation of the circuit

- Precharge the ALU, then fabricate both the basic and precharged configurations, test each version and compare the results

<u>Bibliography</u>

- 1. White, Marvin H. "Characterization of CMOS Devices for VLSI," <u>IEEE Journal of Solid State Circuits</u>, SC-17: 208-214 (April 1982).
- Dennard, R. H. <u>et al.</u> "1 micron MOSFET VLSI Technology: Part II-Device Designs and Characteristics for High-Performance Logic Applications," <u>IEEE Journal of Solid</u> <u>State Circuits</u>, SC-14: 247-255 (April 1979).
- 3. Sommars, W. E. <u>The Design and Layout of a Complementary</u> <u>Metal Oxide Semiconductor Silicon on Sapphire Cell</u> <u>Library</u>. MS thesis. Wright-Patterson AFB, Ohio: School of Engineering, Air Force Institute of Technology, Dec 1983.
- Burns, J. "Switching Response of Complementary-Symmetry MOS Transistor Logic Circuits," <u>RCA</u> <u>Review</u>, 25: 627-661 (December 1964).
- Krambeck, R. H., et al. "High-Speed Compact Circuits with CMOS,"<u>IEEE Journal of Solid State Circuits</u>, SC-17: 614-619 (June 1982).
- 6. Comer, D. <u>Modern Electronic Circuit Design</u>. Massachusetts: Addison-Wesley; 1976.
- 7. Milne, A. D. <u>MOS</u> <u>Devices</u> <u>Design</u> <u>and</u> <u>Manufacture</u>. New York: John Wiley & Sons, 1983.
- Mohsen, A. M. and C. A. Mead. "Delay-Time Optimization for Driving and Sensing of Signals on High-Capacitance Paths of VLSI Systems," <u>IEEE Journal of Solid State</u> <u>Circuits</u>, SC-14 462-470 (April 1979).
- 9. Crawford, R. H. and B. Bazin. "Theory and Design of MOS Capacitor Pull-Up Circuits," <u>IEEE Journal of Solid</u> <u>State Circuits</u>, SC-4: 145-162 (June 1969).
- 10. Can, W. and J. Mize, <u>MOS/LSI</u> <u>Design</u> and <u>Application</u>. New York: McGraw-Hill, 1972.
- 11. Fang, S. "A Novel NMOS E² PROM Scheme," <u>IEEE</u> <u>Journal</u> of <u>Solid</u> <u>State</u> <u>Circuits</u>, SC-18: 610-612 (October 1983).
- 12. Newkirk, J. and Mathews, R. <u>The VLSI Designer's</u> <u>Library</u>. Addison-Wesley, 1983.
- 13. Mead, C. and L. Conway. <u>Introduction to VLSI</u> <u>Systems</u>. Massachusetts: Addison-Wesley, October 1980.

- 14. Gonsalves, N. and H. De Man. "NORA: A Racefree Dynamic CMOS Technique for Pipelined Logic Structures," <u>IEEE Journal of Solid State Circuits</u>, SC-18: 261-266 (June 1983).
- 15. Stewart, R. and D. Plus. "A 40 ns CMOS E² PROM," <u>IEEE</u> <u>Journal of Solid State Circuits</u>, SC-17: 841-846 (October 1982).
- 16. Masuda, H., <u>et al</u>. "Single 5-V, 64K RAM with Scaled-Down MOS Structure," <u>IEEE Journal of Solid State</u> <u>Circuits</u>, SC-15: 672-677 (August 1980).
- 17. Anami, K., <u>et al.</u> "A 35 ns 16K NMOS Static RAM," <u>IEEE Journal of Solid State Circuits</u>, SC-17: 815 (October 1982).
- 18. Chwang, R., et al. "A 70 ns High Density 64K CMOS Dynamic RAM," <u>IEEE Journal of Solid State Circuits</u>, SC-18: 457-462 (October 1983).
- 19. Cobbold, R. <u>Theory</u> and <u>Applications</u> of <u>Field</u> <u>Effect</u> Transistors, New York: John Wiley & Sons, 1970.
- 20. Delhom, L. <u>Design and Application of Transistor</u> <u>Switching Circuits</u>. New York: McGraw-Hill, 1968.
- 21. Crawford, R. <u>MOSFET in</u> <u>Circuit</u> <u>Design</u>, New York: McGraw-Hill, 1967.
- 22. Benedict, R. <u>Electronics</u> for <u>Scientists</u> and <u>Engineers</u>. New Jersey: Prentice-Hall, 1976.
- 23. Kabaservice, T. <u>Applied</u> <u>Microelectronics</u>, New York: West, 1978.
- 24. Ullman, J. <u>Computational Aspects</u> of <u>VLSI</u> Maryland: Computer Science Press, 1984.
- 25. Barna, A. <u>VHSIC Technologies</u> and <u>Tradeoffs</u>. New York: Wiley and Sons, 1981.
- 26. Eimbinder, J. <u>FET</u> <u>Applications</u> <u>Handbook</u>. Pennsylvania: Tab Books, 1967.
- 27. Glasser, L. and W. Song. Introductory CMOS Techniques, unpublished notes. Department of Electrical Engineering and Computer Science, MIT, Cambridge, MA 02139.
- 28. Seitz, C. Unpublished notes on CMOS/SOS. Caltech, August 1982.

BIB-1

Appendix A

SPICE analyses data are presented for the test circuit discussed in Chapter VI. Five different precharge configurations were used to determine the basic function of precharging:

- a. Precharge only the Nand output
- b. Precharge only the first inverter output
- c. Precharge only the second inverter output
- d. Precharge the nand output and the first inverter output at the same time
- e. Precharge the nand output and the second inverter output

A simulation was run on the basic test circuit to act as a baseline for comparison to the precharged configurations.

There were two simulations run for each test circuit configuration. The first was with identical nand input signals pulsed high from 0 - 5NS, low from 5 - 15NS and high from 15 - 40NS. The second simulation inverted the identical inputs to the nand (low - high - low) for the same time frames. A standard precharge pulse was used for each configuration; a high pulse from 4NS to 6NS and low for the remainder of the simulation.

To evaluate the SPICE simulation data, a mextra/SPICE node number cross reference list and a node cifplot are provided for each test circuit configuration. The SPICE

A-1

input data refers to SPICE nodes and the cifplot refers to mextra nodes. In the reference list, the mextra nodes are in the first column and the corresponding SPICE nodes are presented in the second column.

The waveform plots are also based on SPICE reference nodes. To determine which node is being monitored for output, look at the ".PLOT" line of the input listing, find the node number in the "V()" expression (there may be more than one V() expression, since multiple node output waveforms can be plotted) refer to the node list, look at the right column and find the number you retrieved from the input listing. Then find the corresponding mextra node number and locate this node on the circuit node cifplot. This can be repeated for any node on the input listing or the circuit cifplot.



Figure A-1. Basic Test Circuit Node Plot.

Table A-1

BASIC TEST CELL NODE CROSS REFERENCE LIST

.

.

A-3

T

.

r -----

CHOS/SOS TEST CELL CONFIGURATION TRANSIENT ANALYSIS TEMPERATURE - 27.888 DEG C #LEGEND: V(8) V(9) V(2) TIME 1.25#d+## 2.5##d+## 3.75#d+## 5.###d+##

 d+BB
 5, BBBd+BB

 ABBd-BB
 5, BBBd+BB

 ABBd-BB
 5, BBBd+BB

 SBBd-BB
 2, BBd-BB

 SBBd-BB
 2, SBd+BB

 lх ******* Ĵх :

ß

ļ

Figure A-2. SPICE Waveform Plot of Basic Test Circuit with Input 010.

A-5

Ï

. .

A-6

• . • . •

1*******11/27/84 ******** SPICE 26.1 (150CT8#) *******18:#2:47* CHOS/SOS TEST CELL CONFIGURATION . TRANSIENT ANALYSIS TEMPERATURE - 27.888 DEG C ******** SLEGEND: +: V(8) +: V(9) =: V(2) TIME V(8) X(*+=)----- #. d+## 1.25#d+## 2.5##d+## 3.75#d+## 5.###d+## 2.259d-#8 X 2.259d-8 X 2.250d-8 X 2.250 Зx Зx - 3 9 8 2 4 9 8 - 3 9 8 2 4 9 8 - 4 6 8 4 4 9 7 - 4 6 8 4 4 9 7 - 5 7 8 4 4 9 7 - 5 7 8 4 4 9 7 - 5 7 8 4 9 9 - 7 1 9 8 4 9 9 - 7 1 9 8 4 9 9 - 7 1 9 8 4 9 9 - 7 1 9 8 4 9 9 - 7 1 9 8 4 9 9 - 7 1 9 8 4 9 9 - 7 1 9 8 4 9 9 - 7 1 9 8 4 9 9 - 7 1 9 1 9 9 - 7 1 9 1 9 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 1 9 - 7 1 9 -~~~~~~ 333 333334

.

Figure A-3. SPICE Waveform Plot Of Basic Test Circuit with Inputs 101.

A-7



Figure A-4. Node Plot of Test CIrcuit Precharged at the Nand Output.

Table A-2

PRECHARGE NAND OUTPUT ONLY - NODE REFERENCE LIST

GiD	,c
Veid	1
L DS	Ĭ
PROS	1
33	2
45	3
1	ĩ
3	Ś
10	š
16	7
. G	<i>.</i> ,
-	., .,
<u>.</u>	. n
	0
1.5	11
J.	12

-

-

.WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 0 DC 5V *M1 1 3 2 0 NMOS L=5.00 W=20.00 *M3 2 6 1 1 PMOS L=5.00 W=20.00 M4 5 6 1 1 PMOS L=5.00 W=40.00 M4 5 6 1 1 PMOS L=5.00 W=10.00 M6 10 9 0 0 NMOS L=5.00 W=15.00 M7 10 7 5 0 NMOS L=5.00 W=15.00 M8 0 12 11 0 NMOS L=5.00 W=15.00 M10 1 12 11 1 PMOS L=5.00 W=15.00 M11 1 5 12 1 PMOS L=5.00 W=30.00 M12 1 9 5 1 PMOS L=5.00 W=30.00 M12 1 9 5 1 PMOS L=5.00 W=10.00 M12 1 9 0 PULSE (0V SV SNS 0NS 0NS 10NS) VIN2 9 0 PULSE (0V SV SNS 0NS 0NS 10NS) VIN2 9 0 PULSE (0V SV 4NS 0NS 0NS 2NS) VP1 3 0 PULSE (0V SV 4NS 0NS 0NS 2NS) VP2 6 0 PULSE (V11) V(12) V(5) (0V,5V) .END

A-9


Ĺ

Figure A-5. SPICE Waveform Plot of Precharged Nand Output with input 010.

CMOS/SOS TEST CELL PRECHARGING ONLY THE NAND OUTPUT ø ø INPUT LISTING TEMPERATURE = 27.000 DEG C .WIDTH OUT=8ø .OPTIONS ITL1=540 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM)

SPICE 2G.1 (150CT8Ø) *******23:57:23*****

11/27/84 ****

.END

Į.

+LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=2& +LEVEL=1 VDD 1 Ø DC 5V *M1 1 3 2 Ø NMOS L=5.ØU W=2Ø.ØU M3 2 6 1 1 PMOS L=5.ØU W=2Ø.ØU M4 5 6 1 1 PMOS L=5.ØU W=1Ø.ØU M4 5 6 1 1 PMOS L=5.ØU W=1Ø.ØU M5 1 7 5 1 PMOS L=5.ØU W=15.ØU M6 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M7 1Ø 7 5 Ø NMOS L=5.ØU W=15.ØU M8 Ø 12 11 Ø NMOS L=5.ØU W=15.ØU M1Ø 1 12 11 1 PMOS L=5.ØU W=3Ø.ØU M10 1 12 11 1 PMOS L=5.ØU W=3Ø.ØU M11 1 5 12 1 PMOS L=5.ØU W=3Ø.ØU M12 1 9 5 1 PMOS L=5.ØU W=10.ØU M12 1 9 5 1 PMOS L=5.ØU W=3Ø.ØU M12 1 9 5 1 PMOS L=5.ØU W=3Ø.ØU M12 1 9 5 1 PMOS L=5.ØU W=3Ø.ØU N12 1 9 5 1 PMOS L=5.ØU W=1Ø.ØU C13 1 Ø Ø.425PF C14 Ø Ø.3Ø5PF C15 5 Ø Ø.243PF C16 12 Ø Ø.1Ø9PF C17 11 Ø Ø.1PF *C18 2 Ø Ø.68PF VIN1 7 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP2 6 Ø PULSE (£V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(11) V(12) V(5) (ØV,5V) .END

.

- '

he i -



.

Figure A-6. SPICE Waveform Plot of Precharged Nand Output with Inputs 101.

٠.

A-12

.

۰.





Table A-3

 PRECHARGE INV1 OUTPUT ONLY - NODE REFERENCE LIST

 GFD
 Ø

 VC 1

 VC-</td

*****11/27/84 ******* SPICE 2G.1 (150CT8Ø) *******23:59:47*****

Z

.

11



Figure A-8. SPICE Waveform Plot of Precharged First Inverter with Inputs 010.

.

•

A-15

.

٠.•

INPUT LISTING TEMPERATURE = ********* .WIDTH OUT=8Ø .007110NS ITL1=5## ITL5=# .Model NMOS NMOS (VTO=1V TOX=75NM UO=4## NSUB=2.5E16 LD=#.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=2Ø.ØU M2 2 5 1 1 PMOS L=5.ØU W=4Ø.ØU M3 1 7 6 1 PMOS L=5.ØU W=10.ØU M4 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M5 1Ø 7 6 9 NMOS L=5.ØU W=15.ØU M6 Ø 2 11 Ø NMOS L=5.ØU W=15.ØU M7 Ø 6 2 Ø NMOS L=5.ØU W=15.ØU M8 1 2 11 1 PMOS L=5.ØU W=3Ø.ØU M10 1 9 6 1 PMOS L=5.ØU W=3Ø.ØU M10 1 9 6 1 PMOS L=5.ØU W=10.ØU C11 1 Ø Ø.4Ø5PF C12 Ø Ø Ø.3Ø5PF

Ø Ø Ø.3Ø5PF

C12 Ø Ø Ø.3Ø5PF C13 6 Ø Ø.129PF C14 2 Ø Ø.248PF C15 11 Ø Ø.1PF VIN1 7 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (£V ØV 4NS ØNS ØNS 2NS) VP2 5 Ø PULSE (£V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(11) V(2) V(6) (ØV,5V) .END

C12

.END

1*******11/27/84 ******* SPICE 2G.1 (150CT8Ø) *******23:59:34***** CMOS/SOS - PRECHARGE INV1 ONLY - INPUT 101 - STANDARD D4,P2 27.ØØØ DEG C



•

Figure A-9. SPICE Waveform of Precharged First Inverter with Inputs 101.

المتواد والمتراد والمراد والمراد والمراد

A-17





Table A-4

PRECHARGE INV2 ONLY - NODE REFERENCE LIST

GND Vdd	Ø 1
NHOS PMOS	Ø 1
15 41	234
37	÷ 5 6
16 2	7
7 24	9 ເ ສ
9	i 1

F

 g^{****} INPUTLISTING TEMPERATURE = 27.888 g^{*****} .WIDTH OUT=88 .OPTIONS ITL1=560 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 .MODEL PMOS 100 SV M1 1 3 2 0 NMOS L=5.00 W=20.00 M2 2 5 1 1 PMOS L=5.00 W=10.00 M3 1 7 6 1 PMOS L=5.00 W=10.00 M4 10 9 0 0 NMOS L=5.00 W=15.00 M4 10 9 0 0 NMOS L=5.00 W=15.00 M6 0 11 2 0 NMOS L=5.00 W=15.00 M7 0 6 11 0 NMOS L=5.00 W=20.00 M8 1 11 2 1 PMOS L=5.00 W=20.00 M9 1 6 11 1 PMOS L=5.00 W=30.00 M9 1 6 11 1 PMOS L=5.00 W=30.00 M10 1 9 6 1 PMOS L=5.00 W=30.00 M10 1 9 6 1 PMOS L=5.00 W=10.00 C11 1 0 0.109PF C12 0 0 0.305PF C13 6 0 0.129PF C14 11 0 0.109PF C15 2 0 0.77PF VIN1 7 0 PULSE (0V 5V 5NS 0NS 0NS 10NS) VIN2 9 0 PULSE (0V 5V 5NS 0NS 0NS 10NS) VIN2 9 0 PULSE (0V 5V 4NS 0NS 0NS 10NS) VIN2 9 0 PULSE (0V 5V 4NS 0NS 0NS 2NS) .TRAN 0.5NS 40NS .PLOT TRAN V(2) V(11) V(6) (0V,5V) END

1*******11/28/84 ******** SPICE 2G.1 (150CT8Ø) ********Ø0:Ø0:55***** Ø CMOS/SOS PRECHARGE INV2 ONLY - INPUT Ø1Ø - STANDARD D4.P2 Ø**** INPUT LISTING TEMPERATURE = 27.ØØØ DEG C Ø*****

A-19



1

Figure A-11. SPICE Waveform Plot of Precharged Second Inverter with Input 010.

.WIDTH OUT=8Ø .OPTIONS ITL1=5#Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=2ØØ NSUB=3E15 LD=Ø.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=2Ø.ØU M2 2 5 1 1 PMOS L=5.ØU W=2Ø.ØU M3 1 7 6 1 PMOS L=5.ØU W=15.ØU M4 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M5 10 7 6 Ø NMOS L=5.ØU W=15.ØU M6 Ø 11 2 Ø NMOS L=5.ØU W=15.ØU M7 Ø 6 11 Ø NMOS L=5.ØU W=15.ØU M8 1 12 1 PMOS L=5.ØU W=15.ØU M1Ø 1 9 6 1 PMOS L=5.ØU W=3Ø.ØU M1Ø 1 9 6 1 PMOS L=5.ØU W=3Ø.ØU M1Ø 1 9 6 1 PMOS L=5.ØU W=10.ØU C11 1 Ø Ø.4Ø5PF C12 Ø Ø.3Ø5PF C13 6 Ø Ø.129PF C14 11 Ø Ø.1Ø9PF C14 11 Ø Ø.1Ø9PF C15 2 Ø Ø.77FF VIN1 7 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VPI 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) PLOT TRAN V(2) V(11) V(6) (ØV,5V) .END



ľ

Figure A-12. SPICE Waveform Plot of Precharged Second Inverter with Input 101.

A-22





Table A-5

PRECHARGE NAND AND INVI OUTPUTS NODE CROSS REFERENCE LIST

Ø
ĝ
1
2
4
5
6
7
ğ
ĩø
11

1******11/28/84 ******* SPICE 2G.1 (150CT8Ø) *******ØØ:Ø2:11***** CMOS/SOS - PRECHARGE NAND AND INV1 - INPUT Ø1Ø ~ STANDARD D4,P2 ø a**** INPUT LISTING TEMPERATURE = 27.ØØØ DEG C .WIDTH OUT=8ø .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL = 1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) $\begin{array}{c} \text{Hose this triangle tr$ +LEVEL≠1 M12 C13 1 Ø Ø.452PF C14 Ø Ø Ø.3Ø5PF C15 5 Ø Ø.243PF C16 2 Ø Ø.248PF C17 11 Ø Ø.1PF VIN1 7 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP2 6 Ø PULSE (EV ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 40NS .PLOT TRAN V(11) V(2) V(5) (ØV.5V)

•

. .

. END

C



<u>.</u>

t.-

•

.

Figure A-14. SPICE Waveform Plot of Precharged Nand and First Inverter with Inputs 010.

والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع وال

A-25

٠.-

AD-A1	52 466 551F1E	SPE SEN WRI	SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SEMICONDUCTOR VERY LARG. (U) AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGI. B T KELLEY 14 DEC 84 AFIT/GE/ENG/840-41 F/G 20/12										31.5 NL		
				ë ≽`⊷⇒											
							۰۴ د ⁴ .								
	AD-A1	AD-A152 466 UNCLASSIFIE	AD-A152 466 SPE SEM UNCLASSIFIED 14	AD-A152 466 SPEED-UP SEMICOND WRIGHT-PP UNCLASSIFIED 14 DEC 84	AD-A152 466 SPEED-UP TECHN SEMICONDUCTOR WRIGHT-PATTERS UNCLASSIFIED 14 DEC 84 AFIT	AD-A152 466 SPEED-UP TECHNIQUES SENICONDUCTOR VERY L WRIGHT-PATTERSON AFB UNCLASSIFIED 14 DEC 84 AFIT/GE/EN	AD-A152 466 SPEED-UP TECHNIQUES FOR CO SEMICONDUCTOR VERY LARG. C URCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-	AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEM SEMICONDUCTOR VERY LARG. (U) AII UNCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-41	AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY SEMICONDUCTOR VERY LARG. CU) AIR FORC WRIGHT-PATTERSON AFB OH SCHOOL OF ENG UNCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-41	AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL SENICONDUCTOR VERY LARG. (U) AIR FORCE INST WRIGHT-PARTIERSON AFB OH SCHOOL OF ENGL. B UNCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-41 1 1	AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SENICOMOUCTOR VERY LARG. (U) AIR FORCE INST OF TE WRIGHT-PARTTERSON AFB OH SCHOOL OF ENGL.) B T KEI UNCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-41 F/G 2 </td <td>AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SENICONDUCTOR VERY LARG. (U) AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGL. B T KELLEY F/G 20/12 Image: Complete the second</td> <td>AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SEMICONDUCTOR VERY LARG. CU) AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGL. B T KELLEY UNCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-41 5/6 20/12 NL Image: I</td> <td>AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SEMICONDUCTOR VERY LARG. (U) AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGL. BI KELLEY UNCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-41 F/G 20/12 NL Image: Im</td>	AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SENICONDUCTOR VERY LARG. (U) AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGL. B T KELLEY F/G 20/12 Image: Complete the second	AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SEMICONDUCTOR VERY LARG. CU) AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGL. B T KELLEY UNCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-41 5/6 20/12 NL Image: I	AD-A152 466 SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SEMICONDUCTOR VERY LARG. (U) AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGL. BI KELLEY UNCLASSIFIED 14 DEC 84 AFIT/GE/ENG/84D-41 F/G 20/12 NL Image: Im	



MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A

.WIDTH OUT=8Ø .OPTIONS ITL1=5#Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL= .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=20.ØU M2 1 3 5 Ø NMOS L=5.ØU W=20.ØU M3 2 6 1 1 PMOS L=5.ØU W=40.ØU M4 5 6 1 1 PMOS L=5.ØU W=40.ØU M5 1 7 5 1 PMOS L=5.ØU W=10.ØU 1 7 5 1 PMOS L=5.00 W=10.00 10 9 0 0 NMOS L=5.00 W=15.00 10 7 5 0 NMOS L=5.00 W=15.00 0 2 11 0 NMOS L=5.00 W=15.00 0 5 2 0 NMOS L=5.00 W=15.00 0 1 2 11 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 1 5 2 1 PMOS L=5.00 W=30.00 1 5 2 0 PMOS L=5.00 W MG M7 **M**8 2 11 0 MMOS L=5.00 W=15.00 5 2 0 MMOS L=5.00 W=15.00 2 11 1 PMOS L=5.00 W=30.00 5 2 1 PMOS L=5.00 W=30.00 9 5 1 PMOS L=5.00 W=10.00 M9 M1Ø 1 M11 1 M12 1 C13 1 C14 Ø 1 Ø Ø.452PF Ø Ø Ø.3Ø5PF 5 Ø Ø.243PF 2 Ø Ø.248PF C15 C15 5 Ø Ø.243PF C16 2 Ø Ø.243PF C17 11 Ø Ø.1PF VIN1 7 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (£V 5V 4NS ØNS ØNS 2NS) VP2 6 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(11) V(2) V(5) (ØV.5V) END . END

A-26



••••

. . .



Figure A-16. Node Plot of Precharged Nand and Second Inverter Output.

Table A-6

PRECHARGE NAND AND INV2 OUTPUTS - NODE REFERENCE LIST

GMD Volu Imos	Ø 1 Ø
PHOS 15	12
46 1	3 4
8	5
16	7
7	9
24 9	107

/

1*******11/28/84 ******** SPICE 2G.1 (150CT80) ********************************** CMOS/SOS TEST CELL PRECHARGE INV2 AND NAND - INPUT \$18 - STANDARD ø INPUT LISTING TEMPERATURE = 27.000 DEG C ****************** .WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=2Ø.ØU M2 1 3 5 Ø NMOS L=5.ØU W=2Ø.ØU M3 2 6 1 1 PMOS L=5.ØU W=4Ø.ØU M3 2 6 1 1 PMOS L=5.&0U W=4&0.&0UM4 5 6 1 1 PMOS L=5.&0U W=4&0.&0UM5 1 7 5 1 PMOS L=5.&0U W=1&0.&0UM6 1&0.&0.&0M7 1&0.&7.&0.&0M8 &0.&11.&2.&0.&0&U W=15.&0UM9 &0.&5.&11.&0.&0&U W=15.&0UM1&0.&11.&2.&0&U W=15.&0UM1&0.&11.&2.&0&U W=15.&0UM1&1.&1.&2.&1&0&U W=3&0.&0UM11 1 5 11 1 PMOS L=5.&0U W=3&0.&0UM12 1 9 5 1 PMOS L=5.&0U W=1&0.&0UC13 1 &0.&0.&452PFC14 &0.&0&0.&0&0&UC13 1 Ø Ø.452PF C14 Ø Ø Ø.3Ø5PF C15 5 Ø Ø.243PF C16 11 Ø Ø.1Ø9PF C17 2 Ø Ø.77PF VIN1 7 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP2 6 Ø PULSE (EV ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(2) V(11) V(5) (ØV,5V) FND

.END



Figure A-17. SPICE Waveform Plot of Precharged Nand and Second Inverter with Inputs 010.

1*******11/28/84 ******* SPICE 2G.1 (150CT8Ø) ********ØØ:Ø3:Ø8***** CMOS/SOS TEST CELL PRECHARGE INV2 AND NAND - INPUT 101 - STANDARD a g**** TEMPERATURE = INPUT LISTING 27.000 DEG C ******* .WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=2Ø.ØU M2 1 3 5 Ø NMOS L=5.ØU W=2Ø.ØU M3 2 6 1 1 PMOS L=5.ØU W=4Ø.ØU M4 5 6 1 1 PMOS L=5.ØU W=4Ø.ØU M5 1 7 5 1 PMOS L=5.ØU W=10.ØU M6 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M7 1Ø 7 5 Ø NMOS L=5.ØU W=15.ØU M8 Ø 11 2 Ø NMOS L=5.ØU W=15.ØU M1Ø 1 11 2 1 PMOS L=5.ØU W=3Ø.ØU M11 1 5 11 1 PMOS L=5.ØU W=3Ø.ØU M12 1 9 5 1 PMOS L=5.ØU W=3Ø.ØU M12 1 9 5 1 PMOS L=5.ØU W=3Ø.ØU M12 1 9 5 1 PMOS L=5.ØU W=1Ø.ØU C13 1 Ø Ø.452PF +LEVEL=1 C14 Ø Ø Ø.3Ø5PF C14 Ø Ø Ø Ø.305PF C15 5 Ø Ø.243PF C16 11 Ø Ø.109PF C17 2 Ø Ø.77PF VIN1 7 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 9 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (¢V 5V 4NS ØNS ØNS 2NS) VP2 6 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) TPAU Ø FNS 40NS

.TRAN 0.5NS 40NS PLOT TRAN V(2) V(11) V(5) (0V.5V) END

10.10



Figure A-18. SPICE Waveform Plot of Precharged Nand and Second Inverter with Inputs 101.

Appendix B

SPICE analyses of the Memory/ALU Bitslice circuit elements selected for precharge are presented.

The elements selected for precharge were:

- a. 2x4 Memory Row Decoder
- b. 1-Bit RAM Cell
- c. B Register

Two simulations were run on each element. The first input was a high-low-high (5NS - 10NS - 25NS) varying pulse waveform. The second inverted the pulse waveform for the same time periods.

In addition to the two basic simulations, the precharged RAM cell and the B register were simulated for transmission gate device widths of two and three times the basic transmission gate device width. This required four more simulations for these circuits each.

To facilitate evaluation of the output waveforms, the node reference list is provided for each circuit element. The left column provides mextra node numbers and the right column provides the corresponding SPICE node numbers. The SPICE input data uses SPICE reference node numbers and the cifplot of the circuit references mextra nodes. To determine the nodes to which input voltages or ground are applied, identify a node on the cifplot circuit, find the node number on the reference list and look at the right

B-1

column and find the corresponding SPICE node. In this way, the SPICE input data can be directly related to the circuit node cifplot.



.

đ



B-3

<u>Table</u> <u>B-1</u>

BASIC DECODER - NODE REFERENCE LIST

ND /dd MOS MOS	Ø 1 Ø 1
54	2
3	4
Ø	5
2 3 1	6 7
31	8
72	9
44 7	11
59	12
48 26	14
36	15
28	16
18	18
12	10

CMOS/SOS BASIC 2X4 DECODER SPICE TRANSIENT ANALYSIS a**** INPUT LISTING TEMPERATURE = 27.ØØØ DEG C ******* ***** .WIDTH OUT=8Ø .OPTIONS ITL1=5&Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 Ø 3 2 Ø M2 1 2 5 1 NMOS L=5.ØU W=15.ØU PMOS L=5.ØU W=10.0U M3 3 2 PMOS L=5.0U W=30.0U 1 1 8 7 8 8 NMOS L=5.00 W=15.00 Μ4

 M5
 B
 2
 5
 Ø
 NMOS
 L=5.00
 W=15.00

 M6
 Ø
 5
 9
 Ø
 NMOS
 L=5.00
 W=15.00

 M7
 1
 7
 5
 1
 PMOS
 L=5.00
 W=10.00

 M7
 1
 7
 5
 1
 PMOS
 L=5.00
 W=10.00

 M10
 0
 11
 7
 0
 NMOS
 L=5.00
 W=15.00

 M11
 12
 11
 0
 MMOS
 L=5.00
 W=15.00

 M11
 12
 11
 0
 MMOS
 L=5.00
 W=15.00

 M11
 12
 11
 0
 MMOS
 L=5.00
 W=15.00

 M13
 1
 1
 7
 1
 PMOS
 L=5.00
 W=15.00

 M14
 0
 10
 13
 0
 NMOS
 L=5.00
 W=13.00

 M15
 1
 11
 10
 1
 PMOS
 L=5.00
 W=13.00

 M14
 10
 13
 1
 PMOS
 L=5.00
 W=15.00

 M18
 15 M5 8 2 5 Ø M6 Ø 5 9 Ø NMOS L=5.00 W=15.00 M6 Ø NMOS L=5.00 W=15.00 C36 C37 C36 16 Ø Ø.18ØPF C37 7 Ø Ø.132PF C38 10 Ø Ø.128PF C39 13 Ø Ø.128PF C40 2 Ø Ø.171PF C41 5 Ø Ø.128PF C42 9 Ø Ø.130PF VIN1 11 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 3 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) TPAN Ø 5NS 4ØNS TRAN Ø.5NS 4ØNS .PLOT TRAN V(9) V(19) V(16) (ØV.5V)

1******12/Ø1/84 ******** SPICE 2G.1 (150CT8Ø) *******Ø6:45:47*****

.END

. . .

, ,

, . , .

...

a

÷.,

٠. **#**.'~ 6. cm

	CHOS	S/SOS BASIC	2X4 D	ECOP	ER S	301 P	TRAN	SIENT	ANALY	515				
g****	TR	ANGIENT ANA	LVSIS				TEM	PERATU	RE -	27		DEG C	:	
- g						****	****		*****	****	****	*****	*	
•														
ØLEGEND														
*: V(9)													
i va	6)													
х ті	HE	V(9)												
x(++=)-		····· 8 .	d+##		1.25	8d+84		2.584	1d+88		3.75	8d+88	5.8	88d+1
ø.	d+88	5.868d+88	x											
5.880 1.888	1d-18 1d-89	5.0f#d+0# 5.0f#d+0#	X			:			•			:		
1.580	11-89	5.888d+89 5.888d+89	×			:						:		
2.500	1-89	5.8694+88	х Х			•			•			:		1
3.586	11-89	5.8£8d+08	ŝ			:						•		•
4.086	1d-09 1d-09	5.0£8d+08 5.0£8d+08	x			:			:			:		
5.088	11-89	5.8£84+88 5.8£84+88	x			:						:		
6.888	10-09	5.968d+88	¥.			•			:			:		
7.000	10-199	4.985d+08	••			:						•		•
7.506	3d-09 3d-09	4.065d+08 4.454d+88	:		•	:	•		:			:	•	-
8.58	14-89 14-89	3.578d+08 2.3654+##	:			:			•		•	:		
9.500	14-09	1.3524+00	-						•			•	•	
1.000	14-00 94-00	2.745d-01	1. •	-		:			:			:		•
1.100	9 0-0 8	1.136d-01 4.090d-02				:			:			:		•
1.20	00-10	1.9614-82	Ϋ́			•			•			:		
1.380	8.1-88	3.5584-03	Ŷ.			:						•		
1,350	0.1-08 0.1-08	1.4240-03 3.874d-04	x			:			:			:		
1.450	81-00 84-08	1.9#9d-#4 5.754d-#5	×			:			:			:		
1.55	01-08	1.8490-05	x			:			:			:		
1.65	67-60	3.446d-0G	÷.			•			•			•		•
1.75	0-]-08 0-]-08	7.315d-47	Ŷ			:			:			÷		
1.85	8-1-08 8-1-08	1.116d-92	×.			:	•		:•			:		
1.90	80-08 04-08	3.595d-01 8.772d-01	: :		•	:			:			:		
2.09	64-68	1.728d+98	••			•	•		: .			÷		
2.18	8.1-68	3.7754+00	x			:						•		
2.15	01-00 01-00	4.7124+00	x			:			:			:		٠
2.25	80-68	4.879d+##	X			:			•			•		
2.35	84-68	4.9634+88	X			:						:		
2.45	81-88	4.9974+88	Ŷ			:			:			:		
2.50	0d-08 0d-08	4.999d+88 5.868d+88	x			:			:			:		
2.60	0.1-98	5.000d+00	x			•			•			•		
2.79	8-1-00	5.848d+88	X			:			:			:		
2.00	84-80 84-88	5.000d+00 5.000d+00	x			:			:			:		
2.05	6 1-6 8	5.000d+00 5.000d+00	x			:			:			:		
2.95	04-08	5.008d+00	x			•			•			-		
3.05	0.1-00	5.8004+00	ŝ			:			:			:		
3.10	0.1-08 0.1-08	5.800d+08	x			:			,			:		
3.20	0.j-#0 0.j-#0	5.000d+00 5.000d+00	×			:			:			:		
3.00	0.1-03	5.000d+00	X			•			•			·		
3.40	6.1-00	5.0000+00	x			:			:			:		
3.45	កេរ-៩ស កេរ-៩ស	5.000d+00 5.000d+00	X			:			:			:		
3.55	0-1-00 0-1-00	5.008d+08 5.008d+88	X			:	•		:			:		
3.65	9.1-00	5.888d+08	X			•			•			•		
3.75	5-1-00	5.0000+00	Ŷ			:			:			:		
3.00 3.05	11-08 1-08	5.000d+00 5.000d+00	X			:			:			:		
3.90	01-08	5.000d+00 5.000d+00	X			:			:			:		
									-					

. .

Figure B-2. SPICE Waveform Plot of Basic Decoder with Inputs 010.

B-6





Table B-2

PRECHARGE ALL FOUR OUTPUTS OF DECODER - NODE REFERENCE LIST

••••

. . . .

1******12/01/84 ******* CMOS/SOS DECODER FOUR OUTPUT PRECHARGE a INPUT LISTING TEMPERATURE = 27.000 DEG C *************** .WIDTH OUT=8Ø .OPTIONS ITL1=5## ITL5=# MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=5.ØU M2 1 3 5 Ø NMOS L=5.ØU W=5.ØU M3 1 3 6 Ø NMOS L=5.ØU W=5.ØU M4 1 3 7 Ø NMOS L=5.ØU W=5.ØU M5 2 8 1 1 PMOS L=5.ØU W=1Ø.ØU M5 2 8 1 1 PMOS L=5.00 W=10.00 M6 5 8 1 1 PMOS L=5.00 W=10.00 M7 6 8 1 1 PMOS L=5.00 W=10.00 M8 7 8 1 1 PMOS L=5.00 W=10.00 M9 0 10 9 0 NMOS L=5.00 W=15.00 M10 1 9 7 1 PMOS L=5.00 W=10.00 M11 1 10 9 1 PMOS L=5.00 W=30.00 M12 13 12 0 0 NMOS L=5.00 W=15.00 M13 13 9 7 0 NMOS L=5.00 W=15.00 M14 0 7 14 0 NMOS L=5.00 W=15.00 M14 0 7 14 0 NMOS L=5.00 W=10.00 M15 1 12 7 1 PMOS L=5.00 W=10.00 M16 1 7 14 1 PMOS L=5.00 W=30.00 1 7 14 1 PMOS L=5.00 W=10.00 1 7 14 1 PMOS L=5.00 W=30.00 9 6 1 PMOS L=5.00 W=10.00 0 15 12 0 NMOS L=5.00 W=15.00 16 15 0 0 NMOS L=5.00 W=15.00 16 15 0 0 NMOS L=5.00 W=15.00 M16 M17 M18 M19 M2Ø 16 9 6 Ø NHOS L=5.ØU W=15.ØU M21 1 15 12 1 PMOS L=5.ØU W=3Ø.ØU M22 Ø 6 17 Ø NHOS L=5.ØU W=15.ØU M23 1 15 6 1 PHOS L=5.ØU W=1Ø.ØU M24 1 G 17 1 PMOS L=5.00 W≈30.00 M25 1 12 5 1 PMOS L=5.00 W=10.00 M26 18 18 8 8 NMOS L=5.80 W=15.80 M27 18 12 5 8 NMOS L=5.80 W=15.80 M28 8 5 19 8 NMOS L=5.80 W=15.80 M29 1 18 5 1 PMOS L=5.80 W=18.80 M29 1 18 5 1 PMOS L=5.80 W=18.80 1 15 2 1 PMOS L=5.00 W=10.00 M36 1 Ø Ø.1145PF C37 C38 Ø Ø Ø.561PF C39 15 Ø Ø.157PF C4Ø 2 Ø Ø.312PF C41 21 Ø Ø.18ØPF C42 1Ø Ø Ø.151PF C43 5 Ø Ø.3Ø2PF C44 19 Ø Ø.180PF C45 12 Ø Ø.182PF C46 6 Ø Ø.291PF C47 17 Ø Ø.18ØPF C48 9 Ø Ø.171PF 7 Ø Ø.229PF 14 Ø Ø.180PF C49 C SØ C51 3 Ø Ø.5ØPF VINI 15 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 10 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP2 8 Ø PULSE (EV ØV 4NS ØNS ØNS 2NS) TRAN Ø ENS 4000 .TRAN Ø.5NS 4ØNS .PLOT TRAN V(14) V(21) V(17) (ØV,5V) . END

B-9

1******12/71/04 ******** SPICE 26.1 (150CT8#) *********************************** . CHOS/SOS DECODER FOUR OUTPUT PRECHARGE #1 TRANSIENT ANALYSIS TEMPERATURE - 27.888 DEG C g * * * * ********** #LEGEND: •: V(14) •: V(21) •: V(17) x TIME V(14) .)----.... . 1.25#d+## 5.*888*d+88 2.588d+88 d+88 3.75#d+##
 d+80

 5.000

 1.500

 2.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500

 3.500</t

. .

-.

• .

Figure B-4. SPICE Waveform Plot of Internal Node Precharged Decoder with Inputs 010.

B-10

CMOS/SOS DECODER FOUR OUTPUT PRECHARGE a INPUT LISTING TEMPERATURE = 27.000 DEG C .WIDTH OUT*80 .OPTIONS 1TL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 & DC 5V M1 1 3 2 & NMOS L=5.80 W=18.80 M2 1 3 5 & NMOS L=5.80 W=18.80 1 3 6 Ø NMOS L=5.ØU W=10.ØU 1 3 7 Ø NMOS L=5.ØU W=10.ØU M3 M3 1 3 6 0 MMOS L=5.00 W=10.00 M4 1 3 7 0 MMOS L=5.00 W=10.00 M5 2 8 1 1 PMOS L=5.00 W=20.00 M6 5 8 1 1 PMOS L=5.00 W=20.00 M7 6 8 1 1 PMOS L=5.00 W=20.00 M8 7 8 1 1 PMOS L=5.00 W=20.00 M10 1 9 7 1 PMOS L=5.00 W=15.00 M10 1 9 7 1 PMOS L=5.00 W=10.00 M11 1 10 9 1 PMOS L=5.00 W=15.00 M12 13 12 0 0 MMOS L=5.00 W=15.00 M13 13 9 7 0 MMOS L=5.00 W=15.00 M14 0 7 14 0 MMOS L=5.00 W=15.00 M15 1 12 7 1 PMOS L=5.00 W=18.00 M16 1 7 14 1 PMOS L=5.00 W=10.00 M17 1 9 6 1 PMOS L=5.00 W=10.00 M18 0 15 12 0 NMOS L=5.00 W=10.00 M19 16 15 0 0 NMOS L=5.00 W=15.00 M10 16 1 7 14 1 PMOS L=5.00 W=15.00 M19 16 15 0 0 NMOS L=5.00 W=15.00 M19 16 15 0 0 NMOS L=5.00 W=15.00 M20 16 9 6 0 NMOS L=5.00 W=15.00 M21 1 15 12 1 PMOS L=5.00 W=15.00 M22 0 6 17 0 NMOS L=5.00 W=15.00 M24 1 6 17 1 PMOS L=5.00 W=10.00 M25 15 0 1 PMOS L=5.00 W=10.00 M24 1 6 17 1 PMOS L=5.00 W=10.00 M30 L=5.0 M4 M24 1 G 17 1 PMOS L=5.00 W=30.00 M25 1 12 5 1 PMOS L=5.00 W=30.00 M26 18 10 0 0 NMOS L=5.00 W=10.00 M27 18 12 5 0 NMOS L=5.00 W=15.00 M27 18 12 5 0 NMOS L=5.00 W=15.00 M27 18 12 5 Ø NMOS L=5.ØU W=15.ØU M28 Ø 5 19 Ø NMOS L=5.ØU W=15.ØU M29 1 1Ø 5 1 PMOS L=5.ØU W=18.ØU M30 1 5 19 1 PMOS L=5.ØU W=20.ØU M31 1 1Ø 2 1 PMOS L=5.ØU W=19.ØU M32 2Ø 15 Ø Ø NMOS L=5.ØU W=15.ØU M33 2Ø 1Ø 2 Ø NMOS L=5.ØU W=15.ØU M34 Ø 2 21 Ø NMOS L=5.ØU W=15.ØU M35 1 2 21 1 PMOS L=5.ØU W=15.ØU M36 1 15 2 1 PMOS L=5.ØU W=18.ØU C37 1 Ø Ø.1145PF C38 Ø Ø Ø.561PF C38 Ø Ø Ø.561PF C39 15 Ø Ø.157PF C4Ø 2 Ø Ø.312PF C41 21 Ø Ø.18ØPF C42 10 0 0.151PF C43 5 0 Ø.151PF C43 5 0 Ø.302PF C44 19 Ø Ø.180PF C45 12 Ø Ø.182PF C46 6 0 Ø.291PF C47 17 Ø Ø.180PF

- A CARLENSING

B-11
C48 9 Ø Ø.171PF C.9 7 Ø Ø.229PF C.3 14 Ø Ø.180PF C31 3 Ø Ø.50PF VIN1 15 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 1Ø Ø PULSE (ØV 5V 4NS ØNS ØNS 10NS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP2 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(14) V(21) V(17) (ØV,5V) .END

1*****12/#	1/84 ******	• SPICE	26.1 (15	OCT8#) **	*******	47:51****	
в сно	S/SOS DECODE	R FOUR OL	TPUT PRE	CHARGE J2			
#**** TR	ANSIENT ANAL	VS I S		TEMPERAT	URE = 3	27.888 DEG C	
.			*******	*********		*********	•
ØLEGEND:							
•1 V(14)							
-1 V(17)							
TINE	V(14)						
x(*+=)	#.	d+##	1.25#d+#	# 2.58	1d+88	3.75#d+##	5. <i>888</i> 4+ <i>88</i>
8. d•88	- 5. <i>828</i> 4+88 X					• • • • • • •	• • • • •
5.888d-18 1.888d-89	5.889d+88 X 5.889d+88 X		:		•	:	:
1.588d-89 2.888d-89	5.888d+88 X 5.888d+88 X		:			:	:
2,5884-89	5.8884+88 X				•		:
3.5880-89	5.888d+88 X		:		•	:	:
4.5884-89	5.888d+88 X		:		•	:	•
5.5084-89	4.959d+## X		:		•	:	
6,50#d-#9	4.468d+## X		:		•		•
7.5880-89	2.734d+## X		:		•		:
8, <i>888</i> d-89 8,588d-89	1.719d+## X 9.437d-#1 =	•	• :	•	•	:	:
9,808d-89 9,588d-89	4.625d-#1 = 2.113d-#1 =	:.*	:		•	:	•
1.0884-08	8.795d-#2 =	• •	•		•		:
1.1844-88	1.591d-#2 X		•	•		:	:
1.2884-88	2.3314-#3 X		:	•	: •	•	:
1.3004-08	3.588d-04 X		:		•		• . :
1.400d-#8	5.#52d-#5 X		:		•	:	••:
1.45#d-#8	6.248d-86 X		:		•	:	•:
1.550d-88 1.688d-88	1.823d-06 X 7.909d-07 X		:		•	:	:
1.650d-00 1.700J-08	3.563d-07 X 1.667d-07 X		:		•	:	:
1.750J-#0 1.800J-00	8.968d-88 X 6.513d-84 X		:		•	:	• •:
1.05 <i>84-8</i> 8 1.9884-88	4.5#9d-#3 X 3.857d-#2 X		:		:	• :	• :
1.950d-88 2.888d-88	1.255d-#1 = 2.679d-#1 =	• •	•	•	•	•	•
2.058d-08	6.3#6d-#1 =	•	•		•	•	:
2.150d-00	1.826d+88 =	· • `	:	•		:	:
2.2584-88	3.556d+## -	•	:		•	• :	
2.350d-#8 2.350d-#8	4.591d+08 X		:		•		• :
2.4 <i>09</i> d-88 2.450d-88	4.818d+9# X 4.913d+6# X		:		•	:	•:
2.509d-98 2.559d-98	4.968d+88 X 4.987d+88 X		:		•	:	:
2.6004-88	4.995d+08 X		•			:	:
2.704d-88	4.999d+## X		÷				:
2.558-1-88	5.8884+88 X		:		•	•	
2.9001-08	5.888d+88 X		:		•	•	
2.9500-08 3.8880-88	5.0000+00 X		:		:	:	-
3.100.1-00	5.888d+88 X		:		:	:	-
3.150J-#8 3.200J-00	5.88#d+8# X 5.88#d+8# X		:		:	:	
3.250J-88 3.300J-80	5.000d+00 X 5.000d+00 X		:		:	:	:
3.358±-90 3.400d-98	5.0204+08 X 5.0204+08 X		:		:	:	:
3.450.1-88	5.888d+88 X		:		•	:	:
3.558-1-88 3.688.1-88	5.8F84+88 X				:		
3.6504-00	5.888d+88 X		:		•	:	
3.7581-80	5.0F#d+09 X		:		:	•	:
3.8501-08	5.8884+88 X		:		:	:	•
3.3201-00	5.8284+88 X		:		:	:	*
4、尿道疗:3一道电	5.000d+0# X		:-		:	:	

Figure B-5. SPICE Plot of Internal Node Precharged Decoder with Transmission Gate Widths Twice the Basic Widths.

1*******12/Ø1/84 ******** SPICE 2G.1 (150CT8Ø) *******Ø6:48:28***** Ø CMOS/SOS DECODER FOUR OUTPUT PRECHARGE Ø**** INPUT LISTING TEMPERATURE = 27.000 DEG C

.

ĺ

.WIDTH OUT=8Ø .OPTIONS ITL1=5#Ø ITL5=Ø MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +! FVF! = 1 .MODEL PMOS PMOS (VTO=-IV TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 MD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=15.ØU M2 1 3 5 Ø NMOS L=5.ØU W=15.ØU 1 3 6 Ø NMOS L=5.ØU W=15.ØU MЗ M4 1 3 7 Ø NMOS L=5.00 W=15.00 M5 2 8 1 1 PMOS L=5.00 W=30.00 M6 5 8 1 1 PMOS L=5.00 W=30.00 M7 6 8 1 1 PMOS L=5.00 W=30.00 M8 7 8 1 1 PMOS L=5.00 W=30.00 M8 7 8 1 1 PMOS L=5.00 W=30.00 1 15 12 1 PMOS L=5.80 W=38.80 8 6 17 8 NMOS L=5.80 W=15.80 1 15 6 1 PMOS L=5.80 W=18.80 1 6 17 1 PMOS L=5.80 W=38.80 1 12 5 1 PMOS L=5.80 W=18.80 M22 M23 M24 M25 M25 1 12 5 1 PMOS L=5.00 W=10.00 M26 18 10 \mathscr{J} \mathscr{J} NMOS L=5.00 W=15.00 M27 18 12 5 \mathscr{J} NMOS L=5.00 W=15.00 M28 \mathscr{J} 5 19 \mathscr{J} NMOS L=5.00 W=15.00 M29 1 10 5 1 PMOS L=5.00 W=10.00 M30 1 5 19 1 PMOS L=5.00 W=30.00 M31 1 10 2 1 PMOS L=5.00 W=10.00 M32 2 15 \mathscr{J} \mathscr{J} MOS L=5.00 W=10.00 M31 M32 M31 1 10 2 1 PMOS L=5.00 W=10.00 M32 20 15 0 0 NMOS L=5.00 W=15.00 M33 20 10 2 0 NMOS L=5.00 W=15.00 M34 0 2 21 0 NMOS L=5.00 W=15.00 M35 1 2 21 1 PMOS L=5.00 W=10.00 M36 1 15 2 1 PMOS L=5.00 W=30.00 C37 1 0 0.1145PF C38 0 0 0 561PF C38 Ø Ø Ø.561PF 0 0 0.561PF 15 0 0.157PF 2 0 0.312PF 21 0 0.180PF 10 0 0.151PF 5 0 0.302PF C39 C 4 Ø C41 C42 C43 19 Ø Ø.180PF 19 Ø Ø.180PF 12 Ø Ø.182PF 6 Ø Ø.291PF 17 Ø Ø.180PF C43 C44 C45 C46 C47

C48 9 Ø Ø.171PF C49 7 Ø Ø.229PF C51 3 Ø Ø.50PF VIN1 15 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 1Ø Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP2 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØHS .PLOT TRAN V(14) V(21) V(17) (ØV,5V) .END

1

D

Ε.,

•

و موجود موجود م

B-15

-1

112/0	1/04 *******	SPICE ZG.1	(150CT8#) ***	*****#6:48:28	
# CH0:	S/SOS DECODER I	FOUR OUTPUT	PRECHARGE #3		
8**** TR.	ANSIENT ANALYS	[S 	TEMPERATU	RE = 27. 888	DEG C
•					
#LEGEND:					
*1 V(14) *1 V(21)					
*: V(1/) X TIME	VIIA				
X(*+=)	#. d+i	1.25 A	d+## 2.5#8	id+88 3.75i	id+88 5.888d+88
8. d+88	5.8884+88 X				
5.8580-10 1.8580-89	5.888d+88 X	•			
2.5684-89	5.8684+68 X				*
3.5584-89	5.8884+88 X				:
4.588d-89	5.8884+88 X 5.8884+68 X				
5.808d-89 5.508d-89	4.96#d+## X 4.836d+## X				• •
6.808d-89 6.588d-89	4.41#d+## X 3.599d+## X			•	
7.880d-89 7.580d-89	2.4#5d+## X 1.383d+## X		•		
8.8881-89 8.5884-89	7.888d-81 X 3.157d-61 **	• •			
9.8284-89	1.413d-#1 =+* 5.#*4d-#2 =*	•			
1.899d-80	2.569d-#2 X	•			
1.100J-08	4.616d-#3 X 1.793d-#3 X		•		
1,2001-00	5.400d-04 X			•	
1.3#0d-#8 1.350d-#8	7.688d-85 X 2.894d-85 X				• •
1.450-1-88	1.864d-85 X 3.868d-86 X				•
1.500d-00	1.394d-86 X				
1.680J-80	1.876d-07 X 9.425d-08 X				•
1.7000-08	5.372d-08 X 3.782d-08 X				•
1.500J-08	6.5##d-#4 X 4.5#5d-#3 X				•
1.900-0-00	3.856d-02 X	-		•	
2.0000-08	2.679d-01 = 6.384d-01 =	• • •	•		
2.188d-98 2.158d-88	1.121d+0# = 1.826d+## =	• •	•		
2.2091-88	2.681d+88 =+			•	
2.3080-88	4.2 294+89 X				
2.4000-00	4.828d+88 X			-	•
2.5000-00	4.968d+8# X				
2.6000-08	4.995d+88 X				
2.7500-00	4.999d+00 X			•	
2.800.1-00	5.808d+08 X		•	•	•
2.3001-00	5.8 PS d+8 B X		•		
3.808.1-00	5.000000 X			•	
3.109-3-00	5.8080+08 X			•	
3.2004-00	5.028d+08 X		•	•	•
3.2503-00	5.000d+00 X				
3.409.1-08	5.888d+88 X				•
3.4580-80 3.5080-88	5.0000+00 X		•	•	•
3.5501-80	5.0000+00 X		:	:	•
3.6594-80	5.888d+88 X		:	•	•
3,7584-00	5.000d+00 X		:	•	
3.0003-00	5.800d+00 X		•	•	
3.9503-00 4.89983-08	5. SFSd+08 X				
		-			

Figure B-6. SPICE Plot of Internal Node Precharge Decoder with Transmission Gate Widths Three Times the Basic Widths.

₩,₩

ļ

į



l

•

Figure B-7. Node Plot of Decoder Precharged at External Nodes.

Table B-3

2X4 DECODER ALL OUTPUTS PRECHARGED NODE LIST

GND	ø
Vdd	1
PHOS	1
12	2
128	3
2	A,
3Ø	5
51	6
77	7
59	a
13	้เศ
3	īĩ
75	12
33	13
87	14
47	15
62	10
27	18
38	19
8	211
18	21

----------------.WIDTH OUT=8Ø .OPTIONS ITL1=5#Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=5.ØU M2 1 3 5 Ø NMOS L=5.ØU W=5.ØU M3 3 6 Ø NMOS L=5.ØU W=5.ØU 1 1 3 7 Ø NMOS L=5.ØU W=5.ØU M4 M5 2 8 1 1 PMOS L=5.00 W=10.00 M6 5 8 1 1 PMOS L=5.00 W=10.00

 M6 5 8 1 1
 PMOS L=5.80 W=18.80

 M7 6 8 1 1
 PMOS L=5.80 W=18.80

 M8 7 8 1 1
 PMOS L=5.80 W=18.80

 M9 8 18 9 8
 NMOS L=5.80 W=18.80

 M18 1 9 12 1
 PMOS L=5.80 W=18.80

 M18 1 9 12 1
 PMOS L=5.80 W=18.80

 M11 1 18 9 1
 PMOS L=5.80 W=15.80

 M12 14 13 8 8 NMOS L=5.80 W=15.80

 M13 14 9 12 8 NMOS L=5.80 W=15.80

 M13 12 7 8 NMOS L=5.80 W=15.80

 M15 1 13 12 1 PMOS L=5.80 W=15.80

 M15 1 13 12 1 PMOS L=5.80 W=18.80

 M17 1 9 15 1 PMOS L=5.80 W=18.80

 M17 1 9 15 1 PMOS L=5.80 W=18.80

 M18 8 16 13 8 NMOS L=5.80 W=15.80

 M19 17 16 8 8 NMOS L=5.80 W=15.80

 M20 17 9 15 8 NMOS L=5.80 W=15.80

 M21 1 16 13 1 PMOS L=5.80 W=38.80

 1 Ø Ø.1139PF Ø Ø Ø.561PF C37 C38 C39 C38 Ø Ø Ø.551FF C39 16 Ø Ø.157PF C40 20 Ø Ø.126PF C41 2 Ø Ø.234PF C42 10 Ø Ø.151PF C42 18 Ø Ø.127PF C43 18 Ø Ø.127PF C44 5 Ø Ø.234PF C45 13 Ø Ø.182PF C46 15 Ø Ø.128PF

********11/3Ø/84 ******* SPICE 2G.1 (150CT8Ø) *******23:26:23***** CMOS/SOS 2X4 DECODER WITH OUTPUTS PRECHARGED ***** INPUT LISTING TEMPERATURE = 27.000 DEG C

...

C.

· • • •

C47 6 Ø Ø.234PF C48 9 Ø Ø.171PF C49 12 Ø Ø.128PF C50 7 Ø Ø.234PF C51 3 Ø Ø.5ØPF VIN1 16 Ø PULSE (TV IV SNS ØNS ØNS 1ØNS) VIN2 1Ø Ø PULSE (LV IV SNS ØNS ØNS 1ØNS) VPHI1 3 Ø PULSE (WV LV 4NS ØNS ØNS 2NS) VPHI1BAR 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(7) V(2) V(6) (ØV,5V) .END

ľ

د و می موجود از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراج از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از مراجع از م .

.

....



Į

d.

. .

Figure B-8. SPICE Plot of Decoder Outputs Using Basic Width Transmission Gates.

.

.WIDTH OUT=8Ø .OPTIONS ITL1=500 ITL5=0 MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø M2 1 3 5 Ø NMOS L=5.ØU W=1Ø.ØU M1 1 3 2 Ø NMOS L=5.00 W=10.00 M2 1 3 5 Ø NMOS L=5.00 W=10.00 M3 1 3 6 Ø NMOS L=5.00 W=10.00 M4 1 3 7 Ø NMOS L=5.00 W=20.00 M5 2 8 1 1 PMOS L=5.00 W=20.00 M6 5 8 1 1 PMOS L=5.00 W=20.00 M7 6 8 1 1 PMOS L=5.00 W=20.00 M9 Ø 10 9 Ø NMOS L=5.00 W=20.00 M10 1 9 12 1 PMOS L=5.00 W=15.00 M11 1 10 9 1 PMOS L=5.00 W=15.00 M12 14 13 Ø Ø NMOS L=5.00 W=15.00 M13 14 9 12 Ø NMOS L=5.00 W=15.00 M14 Ø 12 7 Ø NMOS L=5.00 W=15.00 M15 1 13 12 1 PMOS L=5.00 W=15.00 M16 1 12 7 1 PMOS L=5.00 W=15.00 M17 1 9 15 1 PMOS L=5.00 W=10.00 M18 Ø 16 13 Ø NMOS L=5.00 W=15.00 M18 Ø 16 13 Ø NMOS L=5.00 W=10.00 M20 17 9 15 1 PMOS L=5.00 W=15.00 M22 Ø 15 6 Ø NMOS L=5.00 W=15.00 M22 Ø 15 6 Ø NMOS L=5.00 W=15.00 M22 Ø 15 6 Ø NMOS L=5.00 W=15.00 M24 1 15 6 1 PMOS L=5.00 W=15.00 M25 1 13 18 1 PMOS L=5.00 W=15.00 M26 19 10 Ø MOS L=5.00 W=15.00 M27 19 13 18 0 NMOS L=5.00 W=15.00 M26 19 10 Ø MOS L=5.00 W=15.00 M27 19 13 18 0 NMOS L=5.00 W=15.00 M26 19 10 Ø Ø NMOS L=5.00 W=15.00 M27 19 13 18 0 NMOS L=5.00 W=15.00 M26 19 10 Ø Ø NMOS L=5.00 W=15.00 M27 19 13 18 0 NMOS L=5.00 W=15.00 M26 19 10 Ø Ø NMOS L=5.00 W=15.00 M27 19 13 18 0 NMOS L=5.00 W=15.00 M28 Ø 18 5 0 NMOS L=5.00 W=15.00 M30 1 18 5 1 PMOS L=5.00 W=15.00 M32 1 10 20 0 NMOS L=5.00 W=15.00 M34 Ø 20 2 Ø NMOS L=5.00 W=15.00 M35 1 20 2 1 PMOS L=5.00 W=15.00 M35 1 20 2 1 PMOS L=5.00 W=15.00 M34 Ø 20 2 0 NMOS L=5.00 W=15.00 M35 1 20 2 1 PMOS L=5.00 W=15.00 M36 1 16 2.00 1 PMOS L=5.00 W=15.00 M36 1 16 2.00 1 PMOS L=5.00 W=10.00 M36 1 16 2.00 1 PMOS L=5.00 W=10.00 M36 1 16 2.00 1 PMOS L=5.00 W=10.00 M36 1 16 2.00 1 PMOS L=5. 5Ø NMOS L=5.00 W=10.00 NMOS L=5.ØU W=10.ØU M3 1 36Ø C41 C42 C41 2 0 0.234PF C42 10 0 0.151PF C43 18 0 0.127PF C44 5 0 0.234PF C45 13 0 0.182PF C45 13 0 0.182PF C46 15 0 0.234PF

٠..

C48 9 Ø Ø.171PF C49 12 Ø Ø.128PF C5Ø 7 Ø Ø.234PF C5Ø 7 Ø Ø.234PF VIN1 16 Ø PULSE (...V 17 5NS ØNS ØNS 1ØNS) VIN2 1Ø Ø PULSE (...V 57 5NS ØNS ØNS 1ØNS) VPHI1 3 Ø PULSE (...V 4NS ØNS ØNS 2NS) VPHI1BAR 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(7) V(2) V(6) (ØV,5V) .END

.

••••

الم من من من

. . . .



Figure B-9. SPICE Plot of Precharged Decoder Outputs Transmission Gate Widths Doubled.

B-24

۱۰ م مست

1*******11/30/84 ******* SPICE 2G.1 (150CT80) ********23:26:55***** CMOS/SOS 2X4 DECODER WITH OUTPUTS PRECHARGED (3×××× INPUT LISTING TEMPERATURE = 27.000 DEG C ~~~~~~ ******

Ø

.WIDTH OUT≠8Ø .OPTIONS ITL1=5## ITL5=# MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NM M2 1 3 5 Ø NM NMOS L=5.ØU W=15.ØU NMOS L=5.ØU W=15.ØU NMOS L=5.ØU W=15.ØU NMOS L=5.ØU W=15.ØU M3 1 360 M4 378 1 2811 PMOS L=5.00 W=30.00 Μ5 5811 PMOS L=5.0U W=30.0U Μ6 PMOS L=5.00 W=30.00 M7 6811 M8 7 8 1 1 PMOS L=5.80 W=38.80 M9 8 18 9 8 NMOS L=5.80 W=15.80 M18 1 9 12 1 PMOS L=5.80 W=18.80 M11 1 18 9 1 PMOS L=5.80 W=38.80 M12 14 13 Ø Ø NMOS L=5.ØU W=15.ØU M13 14 9 12 Ø NMOS L=5.ØU W=15.ØU M14 Ø 12 7 Ø NMOS L=5.ØU W=15.ØU M15 1 13 12 1 PMOS L=5.ØU W=1Ø.ØU M16 1 12 7 1 PMOS L=5.ØU W=3Ø.ØU M16 1 12 7 1 PMOS L=5.00 W=30.00 M17 1 3 15 1 PMOS L=5.00 W=10.00 M18 0 16 13 0 NMOS L=5.00 W=15.00 M19 17 16 0 0 NMOS L=5.00 W=15.00 M20 17 9 15 0 NMOS L=5.00 W=15.00 M21 1 16 13 1 PMOS L=5.00 W=30.00 M22 0 15 6 0 NMOS L=5.00 W=15.00 M22 0 15 6 1 PMOS L=5.00 W=15.00 M23 1 16 15 1 PMOS L=5.80 W=15.80 M24 1 15 6 1 PMOS L=5.80 W=18.80 M25 1 13 18 1 PMOS L=5.80 W=18.80 M25 1 13 18 1 PMOS L=5.80 W=18.80 M26 19 18 8 8 NMOS L=5.80 W=15.80 M27 19 13 18 8 NMOS L=5.80 W=15.80 M28 8 18 5 8 NMOS L=5.80 W=15.80 M38 1 18 5 1 PMOS L=5.80 W=18.80 M31 1 18 29 1 PMOS L=5.80 W=18.80 M32 21 16 8 8 NMOS L=5.80 W=18.80 M34 8 28 2 8 NMOS L=5.80 W=15.80 M35 1 28 2 8 NMOS L=5.80 W=15.80 M35 1 28 2 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 W=15.80 M35 1 28 3 1 PMOS L=5.80 W=15.80 W=16.80 M35 1 28 3 1 PMOS L=5.80 W=16.80 M35 1 28 3 1 PMOS L=5.80 W=16.80 M35 1 28 3 1 PMOS L=5.80 W=18.80 M35 1 PMOS L=5.80 W=18.80 M35 1 PMOS L=5.80 W=18.80 M35 1 PMOS L=5.80 W=18.80 M35 1 PMOS L=5.80 W=18.80 M35 1 PMOS L=5.80 W=18.80 M35 1 PMOS L=5.80 W=18 PMOS L=5.80 W=18.80 M36 1 PMOS L=5.80 W=18 PMOS L=5.80 W=18.80 M23 1 16 15 1 PMOS L=5.00 W=10.00 16 Ø Ø.157PF 20 Ø Ø.126PF C39 C4Ø 2 Ø Ø.234PF C41 10 0 0.151PF C42 18 Ø Ø.127PF C43 C44 5 Ø Ø.234PF C45 13 Ø Ø.182PF C46 15 Ø Ø.120PF C47 G Ø Ø.234PF

B-25

.....

C/3 9 0 Ø.17127 C49 12 Ø Ø.1232F C5Ø 7 Ø Ø.234PF VIN1 16 Ø PULSE (MV TV 5NS ØNS ØNS 1ØNS) VIN2 1Ø Ø PULSE (1V JV 5NS ØNS ØNS 1ØNS) VPHI1 3 Ø PULSE (1V JV 4NS ØNS ØNS 2NS) VPHI1BAR 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(7) V(2) V(6) (ØV.5V) .END .

ς.



•

۰.

1

Figure B-10. SPICE Plot of Precharged Decoder Outputs Transmission Gate Widths Tripled.





Table B-4

BASIC B REGISTER NODE LIST

Ø1012345670911101010901000155709

1******10/05/34 ******* SPICE 2G.1 (150CT80) *******04:27:06***** Ø CMOS/SOS B REGISTER BASIC TRANSIENT ANALYSIS Ø**** INPUT LISTING TEMPERATURE = 27.000 DEG C Ø****

.WIDTH OUT=8Ø .0PTIONS ITL1=5#Ø ITL5=Ø .Model NMOS NMOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +1 FVF1 =1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) .MODEL PMOS PMOS (VTO=-1V TOX=75NN +LEVEL=1 VDD 1 Ø DC 5V M1 4 3 2 Ø NMOS L=5.ØU W=5.ØU M2 6 3 5 Ø NMOS L=5.ØU W=5.ØU M3 8 3 7 Ø NMOS L=5.ØU W=5.ØU M4 1Ø 3 9 Ø NMOS L=5.ØU W=5.ØU M5 2 11 4 1 PMOS L=5.ØU W=1Ø.ØU M6 5 11 6 1 PMOS L=5.ØU W=1Ø.ØU M7 7 11 8 1 PMOS L=5.ØU W=1Ø.ØU M7 7 11 8 1 PMOS L=5.ØU W=1Ø.ØU M8 9 11 10 1 PMOS L=5.ØU W=1Ø.ØU M1Ø 1 3 12 Ø NMOS L=5.ØU W=1Ø.ØU M1Ø 1 3 12 1 PMOS L=5.ØU W=3Ø.ØU M11 16 2 1 1 PMOS L=5.ØU W=3Ø.ØU M12 17 5 1 1 PMOS L=5.ØU W=3Ø.ØU M13 18 7 1 1 PMOS L=5.ØU W=3Ø.ØU M14 19 9 1 1 PMOS L=5.ØU W=3Ø.ØU M15 1 2Ø 11 1 PMOS L=5.ØU W=3Ø.ØU M13 18 7 1 1 PMOS L=5.00 W=30.00 M14 19 9 1 1 PMOS L=5.00 W=30.00 M15 1 20 11 1 PMOS L=5.00 W=10.00 M16 0 2 16 0 NMOS L=5.00 W=15.00 M17 0 5 17 0 NMOS L=5.00 W=15.00 M19 0 9 19 0 NMOS L=5.00 W=15.00 M19 0 9 19 0 NMOS L=5.00 W=15.00 M20 21 12 0 0 NMOS L=5.00 W=15.00 M21 21 20 11 0 NMOS L=5.00 W=15.00 M22 0 11 3 0 NMOS L=5.00 W=15.00 M23 1 11 3 1 PMOS L=5.00 W=15.00 M24 1 12 11 0 PMOS L=5.00 W=15.00 M25 16 23 22 0 NMOS L=5.00 W=5.00 M26 17 23 24 0 NMOS L=5.00 W=5.00 M27 13 23 25 0 NMOS L=5.00 W=5.00 M28 19 23 26 0 NMOS L=5.00 W=5.00 M30 22 27 16 1 FMOS L=5.00 W=10.00 M31 24 27 17 1 FMOS L=5.00 W=10.00 M32 25 27 13 1 FMOS L=5.00 W=10.00 M33 26 27 19 1 FMOS L=5.00 W=10.00 M34 29 28 0 0 NMOS L=5.00 W=15.00 M35 29 12 27 0 NMOS L=5.00 W=15.00 M34 29 28 0 0 NMOS L=5.00 W=15.00 M35 29 12 27 0 NMOS L=5.00 W=15.00 M37 1 27 23 0 NMOS L=5.00 W=15.00 M33 1 23 27 1 PMOS L=5.00 W=15.00 M33 1 23 27 1 PMOS L=5.00 W=15.00 M34 1 27 23 0 NMOS L=5.00 W=15.00 M35 1 27 23 0 NMOS L=5.00 W=15.00 M37 1 27 23 1 PMOS L=5.00 W=15.00 M33 1 23 27 1 PMOS L=5.00 W=15.00 M33 1 23 27 1 PMOS L=5.00 W=15.00 M33 1 23 27 1 PMOS L=5.00 W=15.00 M33 1 23 27 1 PMOS L=5.00 W=15.00 M34 29 28 0 0 NMOS L=5.00 W=15.00 M35 1 27 23 0 NMOS L=5.00 W=15.00 M37 1 27 23 0 NMOS L=5.00 W=15.00 M37 1 27 23 1 PMOS L=5.00 W=15.00 M33 1 23 27 1 PMOS L=5.00 W=15.00 M34 29 26 0 0.84PF C40 25 0 0.84PF C39 26 Ø J.84PF C4Ø 25 Ø J.84PF C41 24 Ø Ø.84PF 22 Ø Ø.34PF 1 Ø Ø.700PF C42 C43 C44 Ø Ø Ø.505PF C45 27 Ø Ø.181PF C46 23 Ø Ø.128PF

•

C47 12 Ø Ø.23ØPF C48 19 Ø Ø.135PF C49 18 9 Ø.139PF C49 18 9 Ø.139PF C50 17 Ø Ø.145PF C51 16 Ø Ø.151PF C52 11 Ø Ø.200PF C53 3 Ø Ø.155PF C54 9 Ø Ø.111PF C55 7 Ø Ø.109PF C56 5 Ø Ø.114PF C57 2 Ø Ø.12ØPF C58 1Ø Ø Ø.54PF C60 6 Ø Ø.54PF C60 6 Ø Ø.54PF VREGSEL 13 Ø PULSE (ØV ØV ØNS ØNS ØNS 3ØNS) VIN1 2Ø Ø PULSE (ØV 5V 5NS ØNS ØNS 3ØNS) VIN2 28 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT1 1Ø Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT3 6 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT4 4 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VDAT4 4 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(26) V(25) (ØV,5V) .END

۱

İ

)

12

-

•-'

- -1

. _ .



•

 ~ 2

•

Figure B-12. SPICE Plot of Basic B Register Output.



Ż



Table B-5

.

:

.

PRECHARGED 3 REGISTER NODE LIST

GND	I
Vcd	1
HHUS	ø
PMOS	1
6	2
176	3
19	4
5	5
4	6
3	7
163	8
96	9
8Ø	1Ø
138	11
95	12
137	13
94	14
136	15
93	16
135	17
76	13
37	19
12Ø	2Ø
2.0	21
48	22
47	23
-16	24
45	25
85	26
90	27
32	23
24	29
ذ 2	لات

l

 1*******11/28/84
 ********14:23:37*****

 Ø
 CMOS/SOS
 PRECHARGED B
 REGISTER
 TRANSIENT
 ANALYSIS

 Ø****
 INPUT
 LISTING
 TEMPERATURE =
 27.000
 DEG C

••••

.WIDTH OUT=8Ø .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NM M2 1 3 5 Ø NM NMOS L=5.ØU W=5.ØU NMOS L=5.ØU W=5.ØU NMOS L=5.00 W=5.00 M3 1 36Ø NMOS L=5.ØU W=5.ØU PMOS L=5.ØU W=1Ø.ØU MA 137Ø 2 M5 2 8 1 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M6 5 8 1 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M7 6 8 1 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M8 7 8 1 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M9 11 1 \emptyset 9 \emptyset NMOS L=5. \emptyset U W=5. \emptyset U M1 \emptyset 13 1 \emptyset 12 \emptyset NMOS L=5. \emptyset U W=5. \emptyset U M11 15 1 \emptyset 14 \emptyset NMOS L=5. \emptyset U W=5. \emptyset U M12 17 1 \emptyset 16 \emptyset NMOS L=5. \emptyset U W=5. \emptyset U M13 9 18 11 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M14 12 18 13 1 FMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M14 12 18 13 1 FMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M15 14 18 15 1 FMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M16 16 18 17 1 FMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M5 8 1 1 16 13 17 1 FMOS L=5.ØU W=10.ØU Ø 2Ø 19 Ø NMOS L=5.ØU W=15.ØU 1 2Ø 19 1 PMOS L=5.ØU W=3Ø.ØU M16 M17 M18 1 20 19 1 PMUS L=5.00 W=30.00 22 9 1 1 PMUS L=5.00 W=30.00 23 12 1 1 PMUS L=5.00 W=30.00 24 14 1 1 PMUS L=5.00 W=30.00 25 16 1 1 PMUS L=5.00 W=30.00 M19 M2Ø M21 25 16 1 1 PMOS L=5.00 W=30.00 1 26 13 1 PMOS L=5.00 W=10.00 1 26 13 1 PMOS L=5.00 W=10.00 M22 M23 1 19 13 1 22 28 2 Ø M32 PMOS L=5.00 W=10.00 Ø NMOS L=5.ØU W=5.ØU M33 23 28 5 Ø 24 28 6 Ø M34 NMOS L=5.00 W=5.00 24 28 6 25 28 7 1 19 29 M35 NMOS L=5.00 W=5.00 M36 ø NMOS L=5.ØU W=5.ØU 1 PMOS L=5.ØU W=1Ø.ØU M37 M38 2 29 22 1 M39 5 29 23 1 M40 6 29 24 1 M41 7 29 25 1 PMOS L=5.00 W=10.00 PMOS L=5.00 W=10.00 PMOS L=5.00 W=19.00 PMOS L=5.00 W=10.00 21 30 3 0 NMOS L=5.00 W=15.00 31 19 29 0 NMOS L=5.00 W=15.0 M42 M43 31 19 29 Ø NMOS L=5.ØU W=15.ØU M44 Ø 29 23 Ø NMOS L=5.ØU W=15.ØU M45 1 29 23 1 PMOS L=5.ØU W=3Ø.ØU M46 1 3Ø 29 1 PMOS L=5.ØU W=1Ø.ØU C47 7 Ø Ø.94PF

C48 6 Ø Ø. 34PF C:9 5 Ø Ø. 34PF C:J 2 Ø Ø. 34PF C:J 1 J Ø.11Ø3PF C52 Ø Ø Ø. 505PF C53 29 Ø Ø.181PF C54 28 Ø Ø.128PF C55 19 Ø Ø.230PF C55 24 Ø Ø.135PF C57 24 Ø Ø.135PF C59 22 Ø Ø.151PF C60 18 Ø Ø.208PF C61 1Ø Ø Ø.155PF C62 16 Ø Ø.111PF C63 14 Ø Ø.155PF C66 17 Ø Ø.54PF C66 17 Ø Ø.54PF C66 17 Ø Ø.54PF C67 15 Ø Ø.54PF C70 3 Ø Ø.50PF VREGSEL 2Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 3ØNS) VIN1 26 Ø PULSE (ØV 5V 5NS ØNS ØNS 3ØNS) VIN1 25 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT1 17 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT2 15 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT3 13 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP1BAR 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .END

4

••••

.

.

....



• •

Figure B-14. SPICE Plot of Precharged B Register Output Using Basic transmission Gates.

*****11/28/84 ******* SPICE 2G.1 (150CT80) *******12:24:32***** CMOS/SOS PRECHARGED B REGISTER TRANSIENT ANALYSIS - WIDTH TIMES TWO ø TEMPERATURE = INPUT LISTING 27.000 DEG C ------.WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 .MODEL NMOS NHOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=1Ø.ØU M2 1 3 5 Ø NMOS L=5.ØU W=1Ø.ØU M3 1 3 6 Ø NMOS L=5.ØU W=1Ø.ØU M4 1 3 7 Ø NMOS L=5.ØU W=1Ø.ØU 2 8 1 1 PMOS L=5.00 W=10.00 5 8 1 1 PMOS L=5.00 W=20.00 5 8 1 1 PMOS L=5.00 W=20.00 Μ5 M6 6 8 1 1 PMOS L=5.00 W=20.00 7 8 1 1 PMOS L=5.00 W=20.00 7 8 1 1 PMOS L=5.00 W=20.00 Μ7 M8

M33 22 28 2 Ø NMOS L=5.ØU W=5.ØU M34 23 28 5 Ø NMOS L=5.ØU W=5.ØU

M35 M36 M37 M33 M39

M40 M41 M42 M43 M44 M45 M45 M46 C47

 23
 28
 5
 Ø
 NMOS
 L=5.ØU
 W=5.ØU

 5
 24
 28
 5
 Ø
 NMOS
 L=5.ØU
 W=5.ØU

 5
 24
 28
 5
 Ø
 NMOS
 L=5.ØU
 W=5.ØU

 5
 25
 23
 7
 Ø
 NMOS
 L=5.ØU
 W=1Ø.ØU

 6
 29
 22
 1
 PMOS
 L=5.ØU
 W=1Ø.ØU

 7
 29
 23
 1
 PMOS
 L=5.ØU
 W=1Ø.ØU

 7
 29
 23
 1
 PMOS
 L=5.ØU
 W=1Ø.ØU

 7
 29
 24
 1
 PMOS
 L=5.ØU
 W=10.ØU

 7
 29
 25
 1
 PMOS
 L=5.ØU
 W=15.ØU

 21
 3Ø
 Ø
 NMOS
 L=5.ØU
 W=15.ØU

 21
 10
 29
 Ø
 NMOS
 L=5.ØU
 W=15.ØU

 21
 29
 23
 1
 PMOS
 L=5.ØU
 W=3J.ØU

 21
 29
 23
 1
 PMOS
 L=5.ØU
 W=3J.ØU
 </t

. :

·••

•

CA8 6 Ø Ø.94PF C:S 5 Ø Ø.94PF C:Ø 2 Ø Ø.94PF C:Ø 2 Ø Ø.94PF C:Ø 2 Ø Ø.11Ø3PF C:S 29 Ø Ø.1181PF C:S 29 Ø Ø.181PF C:S 28 Ø Ø.123PF C:S 29 Ø Ø.135PF C:S 23 Ø Ø.145PF C:S 23 Ø Ø.120PF C:S 24 Ø Ø.111PF C:S 24 Ø Ø.114PF C:S 24 Ø Ø.120PF C:S 25 Ø Ø.120PF C:S 25 Ø Ø Ø.120PF C:S 25 Ø Ø Ø.120PF C:S 25 Ø Ø Ø.120PF C:S 25 Ø Ø Ø.120PF C:S 25 Ø Ø Ø.120PF C:S 25 Ø Ø Ø.54PF C:S 25 Ø Ø Ø.54PF C:S 3 Ø Ø.54PF C:S 3 Ø Ø.54PF C:S 3 Ø Ø.54PF C:S 3 Ø Ø.54PF C:S 3 Ø Ø.50PF VREGSEL 20 Ø PULSE (ØV 5V 5NS ØNS ØNS 3ØNS) VIN1 26 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT1 17 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT2 15 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT3 13 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (5V ØV 4NS ØNS ØNS 1ØNS) VDAT4 11 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1BAR 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .END .

. مرجد

-

-4

•••

а,

. . .

• •

2



Figure B-15. SPICE Plot of Precharged B Register Output Using Transmission Gate Widths Twice the Basic Widths.

*****11/28/84 ******* SPICE 2G.1 (150CT80) *******12:24:43***** CMOS/SOS PRECHARGED B REGISTER TRANSIENT ANALYSIS - WIDTH TIMES INPUT LISTING TEMPERATURE = 27.000 DEG C .WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=15.ØU M2 1 3 5 Ø NMOS L=5.ØU W=15.ØU 1 3 6 Ø NMOS L=5.ØU W=15.ØU M3 1 3 7 Ø NMOS L=5.ØU W=15.ØU 2 8 1 1 PMOS L=5.ØU W=3Ø.ØU MA M5 Μ6 5 8 1 1 PMOS L=5.ØU W=3Ø.ØU 68 1 1 PMOS L=5.00 W=30.00 Μ7 M7 6 8 1 1 PMOS L=5.00 W=30.00 M8 7 8 1 1 PMOS L=5.00 W=30.00 M9 11 10 9 0 NMOS L=5.00 W=5.00 M10 13 10 12 0 NMOS L=5.00 W=5.00 M11 15 10 14 0 NMOS L=5.00 W=5.00 M12 17 10 16 0 NMOS L=5.00 W=5.00 M13 9 18 11 1 PMOS L=5.00 W=10.00 M14 12 13 13 1 FMOS L=5.00 W=10.00 M15 14 13 15 1 FMOS L=5.00 W=10.00 22 9 1 1 PMOS L=5.ØU W=3Ø.ØU 23 12 1 1 PMOS L=5.ØU W=3Ø.ØU 24 14 1 1 PMOS L=5.ØU W=3Ø.ØU 25 16 1 1 PMOS L=5.ØU W=3Ø.ØU M19 M2Ø M21 M22 M23 1 26 13 1 PMOS L=5.00 W=10.00 M24 Ø 9 22 Ø NHOS L=5.00 W=15.00 M24 Ø 9 22 Ø NHOS L=5.00 W=15.00 M25 Ø 12 23 Ø NHOS L=5.00 W=15.00 M26 Ø 14 24 Ø NHOS L=5.00 W=15.00 M27 Ø 16 25 Ø NMOS L=5.ØU W=15.ØU 27 19 Ø Ø NMOS L=5.ØU W=15.ØU 27 26 18 Ø NMOS L=5.ØU W=15.ØU M23 M29 M3Ø Ø 18 19 Ø NMOS L=5.ØU W=15.ØU M31 1 18 1Ø 1 PMOS L=5.ØU W=3Ø.ØU 1 19 13 1 PMOS L=5.ØU W=1Ø.Ø 22 28 2 Ø NMOS L=5.ØU W=5.ØU PMOS L=5.ØU W=1Ø.ØU M32 M33 23 28 5 Ø 24 28 6 Ø M34 NMOS L=5.ØU W=5.ØU M35 28 6 ø NMOS L=5.00 W=5.00 25 28 7 ø NMOS L=5.ØU W=5.ØU M36 PMOS L=5.ØU W=1Ø.ØU PMOS L=5.ØU W=1Ø.ØU 1 19 29 1 M37 22 23 2 5 29 M33 1 5 29 23 1 6 29 24 1 7 29 25 1 L=5.ØU W=1Ø.ØU L=5.ØU W=1Ø.ØU PMOS M39 PMOS M4Ø 6 PMOS L=5.00 W=10.00 NMOS L=5.00 W=15.00 M41 31 30 0 Ø NMOS M42 31 19 29 Ø NMOS L=5.ØU W=15.ØU Ø 29 23 Ø NMOS L=5.ØU W=15.ØU 1 29 23 1 PMOS L=5.ØU W=3Ø.ØU M43 M44 Ø M45 30 29 1 0 0.94PF 17 PMOS L=5.00 W=10.00 M46

B-40

C43 6 ϑ ϑ , 34PF C 9 5 ϑ ϑ , 34PF C 9 5 ϑ ϑ , 34PF C 1 1 ϑ ϑ , 11 ϑ 3PF C 2 ϑ ϑ , 5 ϑ 5PF C 3 29 ϑ ϑ , 181PF C 53 29 ϑ ϑ , 181PF C 54 23 ϑ ϑ , 123PF C 55 19 ϑ ϑ , 23 ϑ PF C 56 25 ϑ ϑ , 135PF C 57 24 ϑ ϑ , 139PF C 58 23 ϑ ϑ , 145PF C 59 22 ϑ ϑ , 151PF C 60 18 ϑ ϑ , 2 ϑ 8PF C 61 1 ϑ ϑ , 155PF C 62 16 ϑ ϑ , 111PF C 63 14 ϑ ϑ , 199PF C 64 12 ϑ ϑ , 114PF C 65 9 ϑ ϑ , 12 ϑ PF C 66 17 ϑ ϑ , 54PF C 67 15 ϑ ϑ , 54PF C 69 11 ϑ ϑ , 54PF C 7 ϑ 3 ϑ ϑ , 5 ϑ PF V REGSEL 2 ϑ ϑ PULSE (ϑ V ϑ V ϑ NS ϑ NS ϑ NS 3 ϑ NS) VIN1 26 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 3 ϑ NS) VDAT1 17 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 1 ϑ NS) VDAT2 15 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 1 ϑ NS) VDAT3 13 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 1 ϑ NS) VDAT3 13 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 1 ϑ NS) VP1 3 ϑ PULSE (ϑ V 5V 4NS ϑ NS ϑ NS 1 ϑ NS) VP1 3 ϑ PULSE (ϑ V 5V 4NS ϑ NS ϑ NS 1 ϑ NS) VP1 3 ϑ PULSE (ϑ V 5V 4NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 4NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 4NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 4NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 4NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5NS ϑ NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5V 5NS ϑ NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5V 5V 5NS ϑ NS ϑ NS ϑ NS 2NS) VP1 3 ϑ PULSE (ϑ V 5V 5V 5V 5V 5V 5V 5V 5V

.



Figure B-16. SPICE Plot of Precharged B Register Output Using Transmission gate Widths Three Times the Basic Widths.





<u>Table</u> <u>B-6</u>

BASIC RAM CELL NODE LIST

GND Velu NEDS PEDS	Ø 1 Ø 1 2
4	2
16	3
15	4
13	5
14	6
G	7
1I	0
3	9



1********11/28/84 ******* SPICE 2G.1 (150CT8Ø) *******13:21:16***** CMOS/SOS BASIC RAM CELL TRANSIENT ANALYSIS - Ø1Ø INPUT TEMPERATURE = 27.000 DEG C INPUT LISTING ****** .WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PHOS (VTO=-1V TOX=75NM UO=200 NSUD=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 4 3 Ø Ø NMOS L=5.ØU W=15.ØU M2 3 4 Ø Ø NMOS L=5.ØU W=15.ØU W=15.ØU M2 3 4 Ø Ø NMOS L=5.ØU W=15.ØU M3 4 6 5 Ø NMOS L=5.ØU W=5.ØU M4 7 6 3 Ø NMOS L=5.ØU W=5.ØU M5 1 3 4 1 PMOS L=5.ØU W=3Ø.ØU M6 1 4 3 1 PMOS L=5.ØU W=3Ø.ØU M7 9 7 1 1 PMOS L=5.ØU W=3Ø.ØU M8 Ø 7 9 Ø NMOS L=5.ØU W=15.ØU C9 9 Ø Ø.71PF C1Ø Ø Ø Ø.291PF C11 7 Ø Ø.92PF C12 1 Ø Ø.292PF 1 Ø Ø.292PF C12 C13 6 Ø Ø.1Ø8PF 4 Ø Ø.182PF C14 C14 4 0 0.182PF C15 3 0 0.185PF VIN1 7 0 PULSE (0V 5V 5NS 0NS 0NS 10NS) VIN2 5 0 PULSE (5V 0V 5NS 0NS 0NS 5NS) VPASS 6 0 PULSE (0V 0V 0NS 0NS 0NS 10NS) .TRAN 0.5NS 40NS .PLOT TRAN V(9) (0V.5V) .END 1******11/28/84 ******* SPICE 2G.I (150CT8Ø) *******13:22:20***** CMOS/SOS BASIC RAM CELL TRANSIENT ANALYSIS - 101 INPUT ø Ø**** INPUT LISTING TEMPERATURE = 27.000 DEG C ****** .WIDTH OUT=8Ø .0PTIONS ITL1=500 ITL5=0 .Model NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB≈2.5E16 LD≈0.7UM) +LEVEL=1.MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 4 3 Ø Ø NMOS L=5.ØU W=15.ØU M2 3 4 Ø Ø NMOS L=5.ØU W=15.ØU M3 4 6 5 Ø NMOS L=5.ØU W=5.ØU 7 6 3 U NMOS L=5.00 W=5.00 MA 1 3 4 1 PHOS L=5.00 W=30.00 1 4 3 1 PHOS L=5.00 W=30.00 9 7 1 1 PHOS L=5.00 W=30.00 M5 MG Μ7 M7 9 / 1 1 PMOS L=5.00 W=30.00 M3 Ø 7 9 Ø NMOS L=5.00 W=15.00 C9 9 Ø Ø.71PF C10 Ø Ø Ø.291PF C11 7 Ø Ø.92PF C12 f α α.992PF 1 Ø Ø.292PF C12 C13 6 0 0.108PF C14 4 Ø Ø.182PF C14 4 0 0.182PF C15 3 0 0.135PF VIN1 7 0 PULSE (SV 0V 5NS 0NS 0NS 10NS) VIN2 5 0 PULSE (SV 0V 5NS 0NS 0NS 5NS) VPASS 6 0 PULSE (0V 0V 0NS 0NS 0NS 10NS) .TRAN 0.5NS 40NS .PLOT TRAN V(9) (0V.5V) . END

. •

- -

÷.,



Figure B-18. SPICE Plot of RAM Cell with Input 010.



Figure B-19. SPICE Plot of RAM Cell with Input 101.

B-46

•



.

Figure B-20. Precharged RAM Cell Node Plot.

Table B-7

PRECHARGED RAM CELL NODE LIST
and the second and the second second

> .WIDTH OUT=8Ø .OPTIONS ITLI=5£Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +LEVEL=1 . MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 4 3 Ø Ø NMOS L=5.ØU W=15.ØU M2 3 4 Ø Ø NMOS L=5.ØU W=15.ØU M3 4 6 5 Ø NMOS L=5.ØU W=5.ØU M4 7 6 3 Ø NMOS L=5.ØU W=5.ØU M5 1 3 4 1 PMOS L=5.ØU W=3Ø.ØU 1 4 3 1 PMOS L=5.ØU W=3Ø.ØU MG M7 1 10 9 0 NMOS L=5.00 W=30.00 M8 9 7 1 1 PMOS L=5.00 W=30.00 M9 9 11 1 1 PMOS L=5.00 W=30.00 M10 0 7 9 0 NMOS L=5.00 W=15.00 C11 9 Ø Ø.31PF C12 Ø Ø Ø.243PF C13 7 Ø Ø.92PF C14 1 Ø Ø.389PF C15 6 Ø Ø.1Ø8PF C15 6 Ø Ø.108PF C16 4 Ø Ø.182PF C17 3 Ø Ø.185PF VIN1 7 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 5 Ø PULSE (5V ØV 5NS ØNS ØNS 5NS) VPASS 6 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VP1 1Ø Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 11 \Im PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(9) (ØV.5V) END .END

> > B-48

1. S. S. S. S. S

. .



> Figure B-21. SPICE Plot of Precharged RAM Output Using Basic Transmission Gate Widths and Input 010.

•

.WIDTH OUT=80 .OFTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDO 1 0 DC 5V M1 4 3 0 0 NMOS L=5.00 W=15.00 M2 3 4 0 0 NMOS L=5.00 W=15.00 M3 4 6 5 0 NMOS L=5.00 W=5.00 M4 7 6 3 0 NMOS L=5.00 W=5.00 M5 1 3 4 1 PMOS L=5.00 W=30.00 M6 1 4 3 1 PMOS L=5.00 W=30.00 M7 1 10 9 0 NMOS L=5.00 W=30.00 M8 9 7 1 1 PMOS L=5.00 W=10.00 M10 0 7 9 0 NMOS L=5.00 W=15.00 C11 9 0 0.31PF C12 0 0 0.243PF C13 7 0 0.92PF C14 1 0 0.389PF C15 6 0 0.108PF C16 4 0 0.389PF C16 4 0.0389FF VIN1 7 0 PULSE (5V 0V 5NS 0NS 0NS 10NS) VIN2 5 0 PULSE (5V 0V 5NS 0NS 0NS 10NS) VIN2 5 0 PULSE (5V 0V 5NS 0NS 0NS 10NS) VP1BAR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VP1BAR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) .FRAN 0.5NS 40NS .PLOT TRAN V(9) (0V.5V) .END

.



Figure B-22. SPICE Plot of Precharged RAM Cell Output Using Basic Transmission Gate and Input 101.

B-51

.

•

4

٠.;

•. .

. .

.WIDTH OUT=80 .PTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 0 DC 5V M1 4 3 0 0 NMOS L=5.0U W=15.0U M2 3 4 0 0 NMOS L=5.0U W=15.0U M3 4 6 5 0 NMOS L=5.0U W=5.0U M4 7 6 3 0 NMOS L=5.0U W=5.0U M5 1 3 4 1 PMOS L=5.0U W=30.0U M6 1 4 3 1 PMOS L=5.0U W=30.0U M7 1 10 9 0 NMOS L=5.0U W=20.0U M8 9 7 1 1 PMOS L=5.0U W=20.0U M10 0 7 9 0 NMOS L=5.0U W=15.0U C11 9 0 0.31PF C12 0 0 0.243PF C13 1 0.389PF C14 1 0 0.389PF C15 6 0 0.108PF VIN1 7 0 PULSE (0V 5V 5NS 0NS 0NS 10NS) VIN2 5 0 PULSE (0V 5V 5NS 0NS 0NS 10NS) VIN2 5 0 PULSE (0V 5V 4NS 0NS 0NS 10NS) VIN2 5 0 PULSE (0V 5V 4NS 0NS 0NS 10NS) VIN2 5 0 PULSE (0V 5V 4NS 0NS 0NS 2NS) VPASS 6 0 PULSE (0V 5V 4NS 0NS 0NS 2NS) VPASR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VPASR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VPASR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VPASR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VPASR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VPASR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VPASR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VPASR 11 0 PULSE (5V 0V 4NS 0NS 0NS 2NS) VPASR 10 0 (0V.5V) .END



Ľ.

Figure B-23. SPICE Plot of Precharged RAM Cell Output with Input 010 and Transmission Gate Widths Twice the Basic Widths.

B-53

.WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 0 DC 5V M1 4 3 0 0 NMOS L=5.00 W=15.00 M2 3 4 0 0 NMOS L=5.00 W=15.00 M3 4 6 5 0 NMOS L=5.00 W=5.00 M4 7 6 3 0 NMOS L=5.00 W=5.00 M5 1 3 4 1 PMOS L=5.00 W=30.00 M6 1 4 3 1 PMOS L=5.00 W=30.00 M7 1 10 9 0 NMOS L=5.00 W=30.00 M8 9 7 1 1 PMOS L=5.00 W=20.00 M10 0 7 9 0 NMOS L=5.00 W=20.00 M10 0 7 9 0 NMOS L=5.00 W=20.00 M10 0 0 0.31FF C12 0 0 0.243FF C13 7 0 0.32FF C14 1 0 0.389FF C15 6 0 0.103FF C16 4 0 0.103FF VIN1 7 0 PULSE (5V 0V 5NS 0NS 0NS 10NS) VIN2 5 0 PULSE (5V 0V 5NS 0NS 0NS 10NS) VIN2 5 0 PULSE (5V 0V 5NS 0NS 0NS 10NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 0NS 2NS) VP12 A 11 0 PUSE (5V 0V 4NS 0NS 0NS 0NS 2NS) .END



•

ディント 日本にないたち

Figure B-24. SPICE Plot of Precharged RAM Cell with Input 101, and Transmission Gate Widths Twice the Basic Widths.

B-55

.WIDTH OUT=8Ø .OPTIONS ITL1=5¢Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=2ØØ NSUB=3E15 LD=Ø.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 4 3 Ø Ø NMOS L=5.ØU W=15.ØU M2 3 4 Ø Ø NMOS L=5.ØU W=15.ØU M3 4 6 5 Ø NMOS L=5.ØU W=5.ØU M4 7 6 3 Ø NMOS L=5.ØU W=5.ØU M5 1 3 4 1 PMOS L=5.ØU W=3Ø.ØU M5 1 3 4 1 PMOS L=5.ØU W=3Ø.ØU M6 1 4 3 1 PMOS L=5.ØU W=3Ø.ØU M7 1 1 Ø 9 Ø NMOS L=5.ØU W=15.ØU M8 9 7 1 1 PMOS L=5.ØU W=3Ø.ØU M1Ø Ø 7 9 Ø NMOS L=5.ØU W=15.ØU M1Ø Ø 7 9 Ø NMOS L=5.ØU W=15.ØU M1Ø Ø 7 9 Ø NMOS L=5.ØU W=15.ØU VI1 9 Ø Ø.81PF C12 Ø Ø Ø.243PF C13 7 Ø Ø.92PF C14 1 Ø Ø.389FF C15 6 Ø Ø.1Ø8PF C16 4 Ø Ø.182PF C17 3 Ø Ø.185PF VIN1 7 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 5 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VIN2 5 Ø PULSE (ØV 5V 4NS ØNS ØNS 1ØNS) VP1 BAR 11 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 11 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .RAN Ø.5NS 4ØNS .PLOT TRAN V(9) (ØV.5V) .END



D

1

•

Figure B-25. SPICE Plot of Precharged RAM Cell Output, with Input 010 and Transmission Gate Widths Tripled.

B-57

.

1

· . .

.WIDTH OUT=8Ø .OPTIONS ITL1=5£Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=2ØØ NSUB=3E15 LD=Ø.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 4 3 Ø Ø NMOS L=5.ØU W=15.ØU M2 3 4 Ø Ø NMOS L=5.ØU W=15.ØU M3 4 6 5 Ø NMOS L=5.ØU W=5.ØU M4 7 6 3 Ø NMOS L=5.ØU W=5.ØU M5 1 3 4 1 PMOS L=5.ØU W=3Ø.ØU M6 1 4 3 1 PMOS L=5.ØU W=3Ø.ØU M7 1 1Ø 9 Ø NMOS L=5.ØU W=3Ø.ØU M8 9 7 1 1 PMOS L=5.ØU W=3Ø.ØU M1Ø Ø 7 9 Ø NMOS L=5.ØU W=15. U C11 9 Ø Ø.31FF C12 Ø Ø Ø.243PF C13 7 Ø Ø.92PF C14 1 Ø Ø.389FF C15 6 Ø 0.1Ø3PF C16 4 Ø Ø.183PF C17 3 Ø Ø.185PF VIN1 7 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 5 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VIN2 5 Ø PULSE (ØV ØN ØNS ØNS ØNS 1ØNS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS) VP12 R 11 Ø PULSE (5V ØV 5NS ØNS ØNS ØNS 2NS)



ł

Figure B-26. SPICE Plot of Precharged RAM Cell With Input 101 and Transmission Gate Widths Tripled.

B-59

Appendix C

This appendix contains SPICE simulation data on the three sections of the ALU bitslice circuit.

Each basic section was simulated first, then two different precharge configurations were simulated for each section. The first section was simulated with precharge of its two output lines and with precharge of four internal nodes. The second section was simulated with precharge of its single output node and with precharge of two internal nodes. The third section was simulated with precharge of its single output and with precharge of four internal nodes. For each precharge configuration, simulations were performed using basic transmission gates, transmission gates with twice the basic transmission gate widths and with transmission gate widths three times the basic widths.

For each circuit configuration, both 010 and 101 inputs were applied to the circuit inputs to permit evaluation of the low-to-high and high-to-low signal waveform transitions.

Ę

Figure C-1. Basic Section 1 CLL Plot.



Figure C-2. Basic Bitslice Section 1 Node Plot.

<u>Table C-1</u>

GND Vdd NMOS 94 113 112 8 1005 99 39 6 74 1005 86 83 82 52 46 5 57	Ø 1 Ø 1 2 3 4 5 6 7 8 9 Ø 1 1 2 3 4 5 6 7 8 9 Ø 1 2 3 4 5 6 7 8 9 Ø 1 2 3 4 5 6 7 8 9 Ø 1 1 2 3 4 5 6 7 8 9 Ø 1 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 5 6 7 8 9 1 2 3 4 5 5 6 7 8 9 1 2 3 4 5 5 7 8 9 1 2 3 4 5 5 7 8 9 1 2 3 4 5 5 7 8 9 8 7 8 9 1 2 3 4 5 1 1 2 1 2 1 1 1 2 1 1 1 1 1 1 1 1 1 1	
5 46 45	19 2Ø 21	
57 1Ø 41	22 23 24	
3 2ø 26	25 26 27	
21 14 23	28 29	
	20	

L

BASIC BITSLICE SECTION 1 NODE REFERENCE LIST

*****12/01/84 ******* SPICE 2G.1 (150CT80) *******00:19:16***** CMOS/SOS BASIC SECTION 1 TRANSIENT ANALYSIS - OUTPUT 101 α 7**** TEMPERATURE = INPUT LISTING

27.000 DEG C

.WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 1 PMOS L=5.ØU W=1Ø.ØU M2 5 3 2 Ø NHOS L=5.ØU W=3Ø.ØU M3 7 6 5 Ø NMOS L=5.ØU W=3Ø.ØU M4 1 6 2 1 PMOS L=5.ØU W=1Ø.ØU M5 1 2 8 1 PMOS L=5.ØU W=1Ø.ØU M6 Ø 9 7 Ø NMOS L=5.ØU W=3Ø.ØU M17 1 16 11 1 PMOS L=5.00 W=10.00 M13 18 19 6 1 PMOS L=5.00 W=10.00 M13 18 19 6 1 PMOS L=5.00 W=10.00 M19 0 6 16 0 NHOS L=5.00 W=15.00 M20 1 6 16 1 PHOS L=5.00 W=30.00 M19 Ø G 16 Ø NHUS L=5.00 W=15.00 M20 1 G 16 1 PHOS L=5.00 W=13.00 M21 1 16 20 1 PHOS L=5.00 W=10.00 M22 22 21 Ø Ø NHOS L=5.00 W=15.00 M23 22 16 20 Ø NHOS L=5.00 W=15.00 M24 1 20 23 1 PHOS L=5.00 W=10.00 M25 1 21 20 1 PHOS L=5.00 W=10.00 M26 24 20 23 Ø NHOS L=5.00 W=30.00 M27 Ø 25 9 Ø NHOS L=5.00 W=30.00 M28 27 26 24 Ø NHOS L=5.00 W=30.00 M30 1 26 23 1 PHOS L=5.00 W=10.00 M31 1 23 26 1 PHOS L=5.00 W=10.00 M32 29 17 25 Ø NHOS L=5.00 W=10.00 M33 Ø 25 27 Ø NHOS L=5.00 W=10.00 M34 30 6 Ø Ø NHOS L=5.00 W=15.00 M35 30 22 26 Ø NHOS L=5.00 W=15.00 M36 1 6 26 1 PHOS L=5.00 W=15.00 M37 29 19 25 1 FHOS L=5.00 W=15.00 M36 1 6 26 1 PHOS L=5.00 W=15.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 25 23 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M36 1 6 26 1 PHOS L=5.00 W=10.00 M37 29 19 25 1 FHOS L=5.00 W=10.00 M38 1 25 23 1 PHOS L=5.00 W=10.00 M39 25 20 207PF C41 17 Ø 0.1174PF C40 6 0 0.1174PF Ø Ø Ø.1174PF C42 C43 6 Ø Ø.297PF 23 Ø Ø.134PF C44 26 0 Ø.111PF 27 0 Ø.3GPF 9 0 Ø.202PF C45

C46 C47

C48 20 0 0.121PF C49 16 0 0.151PF C50 11 0 0.151PF C51 15 0 0.86PF C52 2 0 0.172PF C53 8 0 0.134PF C54 7 0 0.86PF VA0 29 0 PULSE (0V 0V 0NS 0NS 0NS 10NS) VFHI 17 0 PULSE (0V 0V 0NS 0NS 0NS 10NS) VPHI1 17 0 PULSE (0V 0V 0NS 0NS 0NS 10NS) VPHI1EAR 19 0 PULSE (0V 0V 0NS 0NS 0NS 10NS) VS0 28 0 PULSE (0V 0V 0NS 0NS 0NS 10NS) VS1 21 0 PULSE (0V 0V 0NS 0NS 0NS 10NS) VS2 3 0 PULSE (5V 5V 0NS 0NS 0NS 10NS) VS2 3 0 PULSE (5V 5V 0NS 0NS 0NS 10NS) VS3 14 0 PULSE (5V 5V 0NS 0NS 0NS 10NS) TRAN 0.5NS 40NS .PLOT TRAN V(8) V(23) (0V,5V) .END





.

.WIDTH OUT=8Ø

.OPTIONS ITL1=5## ITL5=# .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=4## NSUB=2.5E16 LD=#.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 1 PMC M2 5 3 2 Ø NMC PMOS L=5.ØU W=1Ø.ØU NMOS L=5.ØU W=3Ø.ØU L=5.ØU W=3Ø.ØU MЗ 76 5Ø NMOS M3 7 6 5 8 MMOS L=5.80 W=38.80 M4 1 6 2 1 PMOS L=5.80 W=18.80 M5 1 2 8 1 PMOS L=5.80 W=18.80 M6 Ø 9 7 Ø NMOS L=5.80 W=38.80 M7 12 11 Ø Ø NMOS L=5.80 W=15.80 M8 12 2 8 Ø NMOS L=5.80 W=15.80 M9 1 11 8 1 PMOS L=5.80 W=18.80 M18 1 9 2 1 PMOS L=5.80 W=18.80 M11 1 9 11 1 PMOS L=5.60 W=18.80 1 9 2 1 PMOS L=5.00 W=10.00 1 9 11 1 PMOS L=5.00 W=10.00 13 9 11 0 NMOS L=5.00 W=30.00 15 14 13 0 NMOS L=5.00 W=30.00 1 14 11 1 PMOS L=5.00 W=10.00 0 16 15 0 NMOS L=5.00 W=30.00 18 17 6 0 NMOS L=5.00 W=5.00 M11 M12 M13 M14 M15 M16 M16 18 17 6 Ø NMOS L=5.ØU W=5.ØU M17 1 16 11 1 PMOS L=5.ØU W=1Ø.ØU M13 18 19 6 1 PMOS L=5.ØU W=1Ø.ØU M19 Ø 6 16 Ø NMOS L=5.ØU W=15.ØU M2Ø 1 6 16 1 PMOS L=5.ØU W=3Ø.ØU M21 1 16 2Ø 1 PMOS L=5.ØU W=1Ø.ØU M22 22 21 Ø Ø NMOS L=5.ØU W=15.ØU M23 22 16 2Ø Ø NMOS L=5.ØU W=15.ØU M24 1 2Ø 23 1 PMOS L=5.ØU W=1Ø.ØU M25 1 21 29 1 PMOS L=5.ØU W=1Ø.ØU M26 24 2Ø 23 Ø NMOS L=5.ØU W=1Ø.ØU M26 24 28 23 Ø NMOS L=5.ØU W=10.ØU M26 24 28 23 Ø NMOS L=5.ØU W=30.ØU M27 Ø 25 9 Ø NMOS L=5.ØU W=15.ØU M28 27 26 24 Ø NMOS L=5.ØU M26 24 20 23 0 MMOS L=5.00 W=30.00 M27 0 25 9 0 NMOS L=5.00 W=30.00 M28 27 26 24 0 MMOS L=5.00 W=30.00 M29 1 25 9 1 PMOS L=5.00 W=30.00 M30 1 26 23 1 PMOS L=5.00 W=10.00 M31 1 23 26 1 PMOS L=5.00 W=10.00 M32 29 17 25 0 MMOS L=5.00 W=10.00 M33 0 25 27 0 MMOS L=5.00 W=15.00 M34 30 6 0 0 MMOS L=5.00 W=15.00 M35 30 23 26 0 MMOS L=5.00 W=15.00 M36 1 6 26 1 PMOS L=5.00 W=10.00 M37 29 19 25 1 FMOS L=5.00 W=10.00 M38 1 25 23 1 PMOS L=5.00 W=10.00 M38 1 25 23 1 PMOS L=5.00 W=10.00 M38 1 25 23 1 PMOS L=5.00 W=10.00 C39 1 0 0.1143PF C40 25 0 0.207PF C41 17 0 0.51PF C42 0 0 0.1174PF C43 6 0 0.297PF C44 23 0 0.134PF C45 26 0 0.111PF C46 27 0 0.86PF

C47 9 Ø Ø.202PF C43 20 Ø Ø.121PF C54 7 Ø Ø.36PF C53 8 Ø Ø.134PF C54 7 Ø Ø.36PF VAØ 29 Ø PULSE (5V ØV ØNS ØNS ØNS 1ØNS) VPHI 1 7 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI 12AR 19 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 28 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS1 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS3 24 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS3 14 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) TRAN Ø.5NS 4ØNS .PLOT TRAN V(8) V(23) (ØV,5V) .END

? ?

-2

مرجع مرجع مرجع مرجع



.



C-9

•



Figure C-5. CLL Plot Section 1 with Two Node Precharge.



Figure C-6. Node Plot Section 1 With Two Nodes Precharged



· . •

• •.

BITSLICE SECTIO	N 1	PRECHARGED	AT	TWO	NODES	-	NODE	REFERENCE	LIST
GND Ø Vdd 1 NMOS Ø PMOS 1 18 2 132 3 1 4 191 5 125 6 96 7 115 8 107 11 39 12 6 13 76 14 107 14 107 14 104 15 88 16 85 17 84 18 52 19 4 2.00 71 21 5 2.22 46 23 45 2.4 57 25 41 26 3 27 200 28 26 29 21 3.00 14 31 23 32									

- -

<u> An ann an Anna an Anna an Anna an Anna an Anna an Anna an Anna an Anna an Anna an Anna an Anna an Anna an Anna</u>

 1*******12/Ø1/84
 ********Ø5:Ø8:19*****

 Ø
 CMOS/SOS BITSLICE SECTION 1 PRERCHARGED AT TWO NODES ~ OUTPUT 1Ø1

 Ø****
 INPUT LISTING

 TEMPERATURE =
 27.000 DEG C

.WIDTH OUT=8Ø .OPTIONS ITL1=5£Ø ITL5=Ø MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 MODEL PMOS PMOS (VTO=-1V TOX=75NM UO≈200 NSUB=3E15 LD=0.7UM) +LEVEL≠1 VDD 1 Ø DC 5V M1 1 3 2 Ø NM NMOS L=5.00 W=5.00 M2 35Ø NMOS L=5.ØU W=5.ØU PMOS L=5.00 W=10.00 M3 2 6 1 1 5 6 1 1 PMOS L=5.ØU W=1Ø.ØU 1 8 7 1 PMOS L=5.ØU W=1Ø.ØU 9 8 7 Ø NMOS L=5.ØU W=3Ø.ØU M4 5 6 1 1 1871 М5 M6 11 10 9 Ø NMOS L=5.ØU W=30.ØU 1 10 7 1 PMOS L=5.ØU W=10.ØU 1 7 5 1 PMOS L=5.ØU W=10.ØU M7 **M8** M9 M17 1C 17 16 Ø NMOS L=5.ØU W=3Ø.ØU M18 1 17 14 1 PMOS L=5.ØU W=1Ø.ØU M19 Ø 19 13 Ø NMOS L=5.ØU W=3Ø.ØU M2Ø 21 2Ø 1Ø Ø NMOS L=5.ØU W=5.ØU M21 1 19 14 1 PMOS L=5.ØU W=1Ø.ØU M22 21 22 1Ø 1 FMOS L=5.ØU W=1Ø.ØU M23 Ø 1Ø 19 Ø NMOS L=5.ØU W=15.ØU M24 1 1Ø 19 1 PMOS L=5.ØU W=3Ø.ØU M25 1 19 23 1 PMOS L=5.ØU W=1Ø.ØU M26 25 24 Ø Ø NMOS L=5.ØU W=15.ØU Ø Ø.1478PF C43 1 27 0 0.207PF 20 0 0.51PF 0 0 0.1174PF 10 0 0.297PF C44 C45 C46 C47

C48 2 Ø Ø.234PF C 6 9 28 Ø Ø.111PF C50 29 Ø Ø.86PF C51 12 Ø Ø.202PF C52 23 Ø Ø.121PF C53 19 Ø Ø.151PF C54 14 Ø Ø.171PF C55 18 Ø Ø.86PF C56 7 Ø Ø.172PF C57 5 Ø Ø.234PF C58 11 Ø Ø.86PF VAØ 31 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VBØ 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 2Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 2Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 3Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS3 17 Ø PULSE (EV 5V ØNS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (EV 5V ØNS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (EV ØV ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 ØNS ØNS 4ØNS PLOT TRAN V(5) V(2) (ØV,5V) .END

.



A SALALS

Figure C-7. SPICE Plot Section 1 with Input 010 and Two Node Precharge Basic Gate Widths

INPUT LISTING TEMPERATURE = 27.888 DEG CWIDTH OUT=8Ø .0PTIONS ITL1=5£Ø ITL5=Ø .Model NMOS NMOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=280 NSUB=3E15 LD=0.7UM) +LEVEL≠1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=5.ØU M2 1 3 5 Ø NMOS L=5.ØU W=5.ØU PMOS L=5.ØU W=1Ø.ØU MЗ 2611 PMOS L=5.ØU W=1Ø.ØU 5611 Μ4 M23 Ø 1Ø 19 Ø NMOS L=5.ØU W=15.ØU M24 1 1Ø 19 1 PMOS L=5.ØU W=3Ø.ØU

1******12/Ø1/84 ******* SPICE 2G.1 (150CT8Ø) *******Ø5:2Ø:12*

CMOS/SOS BITSLICE SECTION 1 PRERCHARGED AT TWO NODES - OUTPUT Ø1Ø

C45 20 0 Ø.51PF C46 0 0 Ø.1174PF C47 10 J Ø.297PF

C-15

• ÷

C48 2 Ø Ø.234PF C49 28 Ø Ø.111PF C50 29 Ø Ø.86PF C51 12 Ø Ø.202PF C52 23 Ø Ø.121PF C53 19 Ø Ø.151PF C54 14 Ø Ø.171PF C55 18 Ø Ø.86PF C56 7 Ø Ø.172PF C57 5 Ø Ø.234PF C58 11 Ø Ø.86PF VAØ 31 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VBØ 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 20 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 20 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS1 24 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS1 24 Ø PULSE (EV 5V 4NS ØNS ØNS 1ØNS) VS3 17 Ø PULSE (EV 5V ØNS ØNS ØNS 1ØNS) VS3 17 Ø PULSE (EV 5V ØNS ØNS ØNS 1ØNS) VS1 30 PULSE (EV 5V ØNS ØNS ØNS 1ØNS) VS1 30 PULSE (EV 5V ØNS ØNS ØNS 1ØNS) VS1 30 PULSE (EV 5V ØNS ØNS ØNS 1ØNS) VS1 30 PULSE (EV 5V ØNS ØNS ØNS 2NS) VP1BAR 6 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(5) V(2) (ØV,5V) .END

.

.

.

• .

2 • . •

-

1

5

.



Figure C-8. SPICE Plot Section 1 Input 101 and Two Node Precharge with Basic Gate Widths

.WIDTH OUT=8Ø .OPTIONS ITL1=5& ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL≠ .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=1Ø.ØU 1 3 5 Ø NMOS L=5.ØU W=1Ø.ØU 2 6 1 1 PMOS L=5.ØU W=2Ø.ØU M2 M3 5 6 1 1 PMOS L=5.ØU W=2Ø.ØU 1 8 7 1 PMOS L=5.ØU W=1Ø.ØU 9 8 7 Ø NMOS L=5.ØU W=3Ø.ØU M4 5 6 1 M5 Μ6

 9
 8
 7
 Ø
 NMOS
 L=5.ØU
 W=30.ØU

 1
 1.0
 9
 Ø
 NMOS
 L=5.ØU
 W=30.ØU

 1
 1.0
 7
 1
 PMOS
 L=5.ØU
 W=10.ØU

 1
 7
 5
 1
 PMOS
 L=5.ØU
 W=10.ØU

 0
 0
 1.2
 1.1
 Ø
 NMOS
 L=5.ØU
 W=30.ØU

 1
 15
 1.4
 Ø
 Ø
 NMOS
 L=5.ØU
 W=15.ØU

 2
 15
 7
 5
 Ø
 NMOS
 L=5.ØU
 W=15.ØU

 3
 1
 1.4
 5
 1
 PMOS
 L=5.ØU
 W=10.ØU

 3
 1
 1.4
 5
 1
 PMOS
 L=5.ØU
 W=10.ØU

 M7 Μ8 M9 M1Ø M11 M12 M13 M13 1 14 5 1 PMOS L=5.00 W=10.00 M14 1 12 7 1 PMOS L=5.00 W=10.00 M15 1 12 14 1 PMOS L=5.00 W=10.00 M16 16 12 14 0 NMOS L=5.00 W=30.00 M17 10 17 16 0 NMOS L=5.00 W=30.00 M18 1 17 14 1 PMOS L=5.00 W=10.00 M19 0 19 18 0 NMOS L=5.00 W=30.00 M20 12 00 10 0 NMOS L=5.00 W=30.00 21 20 10 0 NMOS L=5.00 W=5.00 1 19 14 1 PMOS L=5.00 W=10.00 M2Ø M21 21 22 10 1 FMOS L=5.00 W=10.00 M22 Ø 10 19 Ø NMOS L=5.ØU W=15.ØU M23 M24 I 1Ø 19 1 PMOS L=5.ØU W=3Ø.ØU M25 1 19 23 1 PMOS L=5.ØU W=1Ø.ØU M25 1 19 23 1 PMOS L=5.00 W=10.00 M26 25 24 Ø Ø NMOS L=5.00 W=15.00 M27 25 19 23 Ø NMOS L=5.00 W=15.00 M28 1 23 2 1 PMOS L=5.00 W=10.00 M30 26 23 2 Ø NMOS L=5.00 W=10.00 M31 Ø 27 12 Ø NMOS L=5.00 W=15.00 M32 Q=20 20 20 0 MMOS L=5.00 W=15.00 M31 \emptyset 27 12 \emptyset NMOS L=5. \emptyset U W=15. ϑ U M32 29 23 26 \emptyset NMOS L=5. ϑ U W=3 ϑ . ϑ U M33 1 27 12 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M34 1 23 2 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M35 1 3 ϑ 23 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M36 31 2 ϑ 27 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M36 31 2 ϑ 27 ϑ NMOS L=5. ϑ U W=2 ϑ . ϑ U M37 ϑ 27 29 ϑ NMOS L=5. ϑ U W=2 ϑ . ϑ U M38 32 1 ϑ ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M39 32 3 ϑ 23 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M40 1 1 ϑ 23 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M41 31 22 27 1 FMOS L=5. ϑ U W=1 ϑ . ϑ U M42 1 27 2 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U C43 1 ϑ ϑ .1470PF C44 27 ϑ ϑ . φ 7PF 27 Ø Ø._Ø7PF 20 Ø Ø.51PF Ø Ø Ø.1174PF 10 Ø Ø.297PF C44 C45 C46 C47

L*******12/Ø1/84 ******** SPICE 2G.1 (150CT8Ø) *******Ø5:21:25***** Ø CMOS/SOS BITSLICE SECTION 1 PRERCHARGED AT TWO NODES - OUTPUT Ø1Ø D**** INPUT LISTING TEMPERATURE = 27.ØØØ DEG C

7

C48 2 Ø Ø.234PF C49 28 Ø Ø.111PF C50 29 Ø Ø.36PF C51 12 Ø Ø.202PF C52 23 Ø Ø.121PF C53 19 Ø Ø.151PF C54 14 Ø Ø.171PF C55 18 Ø Ø.86PF C56 7 Ø Ø.172PF C57 5 Ø Ø.234PF C58 11 Ø Ø.86PF VAØ 31 Ø PULSE (ØV 5V ØNS ØNS ØNS 1ØNS) VPHI 20 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI12AR 22 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS4 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS4 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS4 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS4 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS4 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS2 8 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS1 24 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 1ØNS) VS1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 6 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(5) V(2) (ØV,5V) .END

١,



....

Figure C-9. SPICE Plot Section 1 Input 010 and Two Node Precharge with Basic Widths x2

.WIDTH OUT=8Ø .OPTIONS ITL1=500 ITL5=0 MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD≠0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=1Ø.ØU M2 1 3 5 Ø NMOS L=5.ØU W=1Ø.ØU 2 6 1 1 PMOS L=5.0U W=20.0U M3 5 6 1 1 PMOS L=5.00 W=20.00 1 8 7 1 PMOS L=5.00 W=10.00 9 8 7 0 NMOS L=5.00 W=30.00 M4 M5 MG 21 20 10 0 NMOS L=5.80 W=5.80 1 19 14 1 PMOS L=5.80 W=10.80 21 22 10 1 FMOS L=5.80 W=10.80 M2Ø M21 M21 1 19 14 1 PMOS L=5.00 W=10.00 M22 21 22 10 1 FMOS L=5.00 W=10.00 M23 0 10 19 0 NMOS L=5.00 W=15.00 M24 1 10 19 1 PMOS L=5.00 W=10.00 M25 1 19 23 1 PMOS L=5.00 W=10.00 M26 25 24 0 0 NMOS L=5.00 W=10.00 M27 25 19 23 0 NMOS L=5.00 W=10.00 M29 1 24 23 1 PMOS L=5.00 W=10.00 M30 26 23 2 0 NMOS L=5.00 W=10.00 M31 0 27 12 0 NMOS L=5.00 W=30.00 M31 0 27 12 0 NMOS L=5.00 W=30.00 M32 1 23 2 1 PMOS L=5.00 W=30.00 M32 29 28 26 0 NMOS L=5.00 W=30.00 M34 1 28 2 1 PMOS L=5.00 W=30.00 M35 1 30 23 1 PMOS L=5.00 W=30.00 M36 31 20 27 0 NMOS L=5.00 W=10.00 M37 0 27 29 0 NMOS L=5.00 W=10.00 M36 31 20 27 0 NMOS L=5.00 W=10.00 M37 0 27 29 0 NMOS L=5.00 W=10.00 M39 32 30 23 0 NMOS L=5.00 W=15.00 M40 1 10 23 1 PMOS L=5.00 W=15.00 M40 1 10 23 1 PMOS L=5.00 W=15.00 M40 1 10 23 1 PMOS L=5.00 W=10.00 M40 1 10 23 1 PMOS L=5.00 W=10.00 M41 31 22 27 1 PMOS L=5.00 W=10.00 M40 1 10 23 1 PMOS L=5.00 W=10.00 M41 31 22 27 1 PMOS L=5.00 W=10.00 M42 1 27 2 1 PMOS L=5.00 W=10.00 M42 1 27 2 1 PMOS L=5.00 W=10.00 M42 1 0 0.1478PF C44 27 0 0.207PF C45 20 0 0.51PF C46 0 0 .1174PF M22 M42 C43 C44 C45 C45 2Ø Ø Ø.51PF C46 Ø Ø Ø.1174PF C47 1Ø Ø Ø.297PF

C48 2 Ø Ø.234PF C49 28 Ø Ø.111PF C5Ø 29 Ø Ø.86PF C51 12 Ø Ø.202PF C52 23 Ø Ø.121PF C53 19 Ø Ø.151PF C54 14 Ø Ø.171PF C55 18 Ø Ø.86PF C56 7 Ø Ø.172PF C57 5 Ø Ø.234PF C58 11 Ø Ø.86PF VAØ 31 Ø PULSE (5V ØV ØNS ØNS ØNS 1ØNS) VBØ 21 Ø PULSE (5V ØV ØNS ØNS ØNS 1ØNS) VPHI120 Ø PULSE (5V ØV ØNS ØNS ØNS 1ØNS) VPHI1BAR 22 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 3Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 3Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS1 24 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 8 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS3 17 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VS1 3 Ø PULSE (5V V2) (ØV,5V) .END



Figure C-10. SPICE Plot Section 1 Input 101 Two Node Precharge with Gate Widths x2

C-23

.WIDTH OUT=8Ø .OPTIONS ITLI=5¢Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX≈75NM UO≈4ØØ NSUB≈2.5E16 LD=Ø.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO≈200 NSUB=3E15 LD=0.7UM) LEVEL=1 VDD 1 Ø DC M1 1 3 2 Ø M2 1 3 5 Ø 5V M1 1 M2 1 NMOS L=5.ØU W=15.ØU NMOS L=5.ØU W=15.ØU PMOS L=5.0U W=30.0U M3 26 1 1 PMOS L=5.00 W=30.00 5 6 1 1 M4 5 6 1 1 PMOS L=5.ØU W=3Ø.ØU M5 1 8 7 1 PMOS L=5.ØU W=1Ø.ØU M6 9 8 7 Ø NMOS L=5.ØU W=3Ø.ØU M7 11 1Ø 9 Ø NMOS L=5.ØU W=3Ø.ØU M8 1 1Ø 7 1 PMOS L=5.ØU W=1Ø.ØU M1Ø Ø 12 11 Ø NMOS L=5.ØU W=1Ø.ØU M10 Ø 12 11 Ø NMOS L=5.ØU W=3Ø.ØU M11 15 14 Ø Ø NMOS L=5.ØU W=15.ØU M12 15 7 5 Ø NMOS L=5.ØU W=15.ØU M13 1 14 5 1 PMOS L=5.ØU W=1Ø.ØU M14 1 12 7 1 PMOS L=5.ØU W=1Ø.ØU MA M14 1 12 7 1 PMOS L=5.80 W=18.80 M15 1 12 14 1 PMOS L=5.80 W=18.80 M16 16 12 14 \emptyset NMOS L=5.80 W=38.80 M17 10 17 16 \emptyset NMOS L=5.80 W=38.80 M18 1 17 14 1 PMOS L=5.80 W=18.80 M19 \emptyset 19 18 \emptyset NMOS L=5.80 W=38.80 M18 1 17 14 1 PMOS L=5.00 W=10.00 M19 0 19 18 0 NMOS L=5.00 W=30.00 M20 21 20 10 0 NMOS L=5.00 W=30.00 M21 1 19 14 1 PMOS L=5.00 W=10.00 M22 21 22 10 1 FMOS L=5.00 W=10.00 M23 0 10 19 0 NMOS L=5.00 W=10.00 M24 1 10 19 1 PMOS L=5.00 W=10.00 M25 1 19 23 1 PMOS L=5.00 W=15.00 M26 25 24 0 0 NMOS L=5.00 W=15.00 M27 25 19 23 0 NMOS L=5.00 W=15.00 M28 1 23 2 1 PMOS L=5.00 W=10.00 M30 26 23 2 0 NMOS L=5.00 W=10.00 M31 0 27 12 0 NMOS L=5.00 W=10.00 M32 29 28 26 0 NMOS L=5.00 W=30.00 M33 1 27 12 0 NMOS L=5.00 W=30.00 M34 1 28 2 1 PMOS L=5.00 W=30.00 M35 1 30 23 1 PMOS L=5.00 W=30.00 M36 31 20 27 0 NMOS L=5.00 W=30.00 M36 31 20 27 0 NMOS L=5.00 W=30.00 M36 31 20 27 0 NMOS L=5.00 W=10.00 M36 31 20 27 0 NMOS L=5.00 W=10.00 M36 31 20 27 0 NMOS L=5.00 W=15.00 M36 31 20 27 0 NMOS L=5.00 W=15.00 M36 32 10 0 NMOS L=5.00 W=15.00 M37 0 27 29 0 NMOS L=5.00 W=15.00 M40 1 10 23 1 PMOS L=5.00 W=15.00 M42 1 27 2 1 PMOS L=5.00 W=10.00 M43 1 0 0.1470PF M44 1 27 2 0 0 M0 M45 L=5.00 W=10.00 M45 1 10 20 100 L=5 C43 1 Ø Ø.1478PF C44 27 Ø Ø.2Ø7PF C45 2Ø Ø Ø.51PF ~ 1174PF C46 Ø Ø Ø.1174PF C47 1Ø Ø Ø.297PF

******12/Ø1/84 ******** SPICE 2G.1 (150CT8Ø) *******Ø5:22:28***** CMOS/SOS BITSLICE SECTION 1 PRERCHARGED AT TWO NODES - OUTPUT 1Ø1 **** INPUT LISTING TEMPERATURE = 27.ØØØ DEG C
C48 2 Ø Ø.234PF C49 28 Ø Ø.111PF C53 29 Ø Ø.86PF C51 12 Ø Ø.202PF C52 23 Ø Ø.121PF C53 19 Ø Ø.151PF C54 14 Ø Ø.171PF C55 18 Ø Ø.86PF C56 7 Ø Ø.172PF C57 5 Ø Ø.234PF C58 11 Ø Ø.86PF VAØ 31 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 2Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 2Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 3Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSG 3Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS2 8 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS3 17 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V V 2) (ØV,5V) .END



Figure C-11. SPICE Plot Section 1 Input 010 Two Node Precharge with Gate Widths x3

.WIDTH OUT=8Ø .OPTIONS ITL1=5£Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +LEVEL = 1.MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=15.ØU M2 1 3 5 Ø NMOS L=5.ØU W=15.ØU VDC L=5.ØU W=15.ØU PMOS L=5.ØU W=3Ø.ØU PMOS L=5.ØU W=3Ø.ØU M3 2 6 1 1 5611 MA M4 5 6 1 1 PMOS L=5.00 W=30.00 M5 1 8 7 1 PMOS L=5.00 W=10.00 M6 9 8 7 0 NMOS L=5.00 W=30.00 M7 11 10 9 0 NMOS L=5.00 W=30.00 M8 1 10 7 1 PMOS L=5.00 W=10.00 M9 1 7 5 1 PMOS L=5.00 W=10.00 M10 0 12 11 0 NMOS L=5.00 W=30.00 M11 15 14 0 0 NMOS L=5.00 W=15.00 M12 15 7 5 0 NMOS L=5.00 W=15.00 M13 1 4 5 1 PMOS L=5.00 W=15.00 W=15.00 M13 1 4 5 1 PMOS L=5.00 W=15.00 W=15 M12 15 7 5 Ø NHOS L=5.ØU W=15.ØU M13 1 14 5 1 PHOS L=5.ØU W=1Ø.ØU M14 1 12 7 1 PHOS L=5.ØU W=1Ø.ØU M15 1 12 14 1 PHOS L=5.ØU W=1Ø.ØU M16 16 12 14 Ø NHOS L=5.ØU W=3Ø.ØU M17 1C 17 16 Ø NHOS L=5.ØU W=3Ø.ØU M18 1 17 14 1 PHOS L=5.ØU W=3Ø.ØU M19 Ø 19 13 Ø NHOS L=5.ØU W=3Ø.ØU M20 21 2Ø 1Ø Ø NHOS L=5.ØU W=1Ø.ØU M22 21 22 1Ø 1 FHOS L=5.ØU W=1Ø.ØU M23 Ø 1Ø 19 Ø NHOS L=5.ØU W=1Ø.ØU M24 1 1Ø 19 1 PHOS L=5.ØU W=10.ØU M25 1 19 23 1 PHOS L=5.ØU W=1Ø.ØU M24 1 10 19 1 PMOS L=5.00 W=30.00 M25 1 19 23 1 PMOS L=5.00 W=10.00 M26 25 24 \emptyset \emptyset NMOS L=5.00 W=15.00 M27 25 19 23 \emptyset NMOS L=5.00 W=15.00 M28 1 23 2 1 PMOS L=5.00 W=10.00 M30 26 23 2 \emptyset NMOS L=5.00 W=30.00 M31 \emptyset 27 12 \emptyset NMOS L=5.00 W=30.00 M32 29 28 26 \emptyset NMOS L=5.00 W=30.00 M32 1 27 12 1 PMOS L=5.00 W=30.00 M32 1 27 12 1 PMOS L=5.00 W=30.00 M34 1 23 2 1 PMOS L=5.00 W=10.00 M35 1 30 27 \emptyset NMOS L=5.00 W=10.00 M36 31 20 27 \emptyset NMOS L=5.00 W=10.00 M37 \emptyset 27 29 \emptyset NMOS L=5.00 W=10.00 M36 31 20 27 \emptyset NMOS L=5.00 W=10.00 M37 \emptyset 27 29 \emptyset NMOS L=5.00 W=15.00 M39 32 30 23 \emptyset NMOS L=5.00 W=15.00 M40 1 10 23 1 PMOS L=5.00 W=10.00 M41 31 22 27 1 FMOS L=5.00 W=10.00 M42 1 27 2 1 PMOS L=5.00 W=10.00 M42 1 27 2 1 PMOS L=5.00 W=10.00 M42 1 27 2 1 PMOS L=5.00 W=10.00 C43 1 \emptyset 0.1478PF C44 27 \emptyset 0.207PF C45 2 \emptyset \emptyset 0.51PF C46 \emptyset \emptyset 0.1174PF M25 1 19 23 1 PMOS L=5.ØU W=1Ø.ØU C46 C47 Ø Ø Ø.1174PF 10 J Ø.297PF

C48 2 Ø Ø.234PF
C49 28 Ø Ø.111PF
C5Ø 29 Ø Ø.86PF
C51 12 Ø Ø.202PF
C52 23 Ø Ø.121PF
C53 19 Ø Ø.151PF
C54 14 Ø Ø.171PF
C55 18 Ø Ø.86PF
C56 7 Ø Ø.172PF
C57 5 Ø Ø.234PF
C58 11 0 0.86PF
VAN 31 N PHISE (5V NV ANS ANS ANS LANS)
VB0 21 0 PULSE (OV OV ONS ONS ONS 100S)
VPHT1 20 0 PULSE (5V 5V ONS ONS ONS 10NS)
VPHTIBAR 22 A PHISE (AV AV ANS ANS ANS LANS)
VSA 30 0 PULSE (AV AV ANS ANS ANS LANS)
VS1 24 G PULSE (AV AV ANS ANS ANS LANS)
VS2 8 0 PULSE (EV 5V DNS DNS DNS LDNS)
VS2 17 A PHUSE (5V 5V ANS ANS ANS 1ANS)
VPI 2 0 PHISE (OV EV ANS ONS ONS 2NS)
VRIDAD & A PHISE (BV AV ANS ANS ANS DOD
TDAN A ENC AANS
PLOT TEAN V(E) V(2) (AV EV)
END



Figure C-12. SPICE Plot Section 1 Input 101 Two Node Precharge with Gate Widths x3



•

Figure C-13. CLL Plot Section 1 Four Node Precharge



Figure C-14. Section 1 Four Node Precharge Node Plot

C-30





MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A

Table	C-3

.

BITSLICE	FIRST	SECTION	WITH	FOUR	NODES	PRECHARGED	- NODE	REFERENCE	LIST
BITSLICE BITSLICE SND VMOS 20 140 140 157 127 114 157 100 100 100 100 100 100 100 10	F F RST Ø 1 Ø 1 2 3 4 5 6 7 8 9 1 1 1 2 3 4 5 6 7 8 9 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2	SECTION	WITH 1	FOUR	NODES	PRECHARGED	- NODE	REFERENCE	LIST
21	30								
14 23	31 32								

.WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NM M2 1 3 5 Ø NM NMOS L=5.00 W=5.00 NMOS L=5.ØU W=5.ØU NMOS L=5.ØU W=5.ØU 36Ø M3 1 NMOS L=5.ØU W=5.ØU PMOS L=5.ØU W=1Ø.ØU M4 1 3 7 ø Μ5 28 1 1 PMOS L=5.0U W=10.0U PMOS L=5.0U W=10.0U M6 581 1 M7 68 1 1 16 6 Ø Ø NMOS L=5.ØU W=15.ØU 16 7 13 Ø NMOS L=5.ØU W=15.ØU 1 6 13 1 PMOS L=5.ØU W=10.ØU 1 14 7 1 PMOS L=5.ØU W=10.ØU 1 14 6 1 PMOS L=5.ØU W=10.ØU M15 M16 M17 M181111PHOSL=5.00W=10.00M1911461PMOSL=5.00W=30.00M20171460NMOSL=5.00W=30.00M211918170NMOSL=5.00W=30.00M2211861PMOSL=5.00W=30.00M23000NMOSL=5.00W=30.00M242221110NMOSL=5.00W=30.00M242221110NMOSL=5.00W=10.00M242221110NMOSL=5.00W=10.00M2512061PMOSL=5.00W=10.00M262223111FMOSL=5.00W=10.00M27011200NMOSL=5.00W=10.00M281122050NMOSL=5.00W=15.00M3025240NMOSL=5.00W=15.00WM0M3125240NMOSL=5.00W=10.00WM3215261PMOSL=5.00W=10.00M3125240NMOSL=5.00W=10.00WM3215260NMOSL=5.00W=10.00M3312451PMOSL=5.00W=10.00</t M18

C48 28 Ø Ø.207PF C49 21 Ø Ø.51PF C50 Ø Ø Ø.1174PF C51 11 Ø Ø.297PF C52 26 Ø Ø.134PF C53 2 Ø Ø.274PF C54 29 Ø Ø.86PF C55 14 Ø Ø.202PF C56 5 Ø Ø.283PF C57 20 Ø Ø.151PF C58 6 Ø Ø.303PF C59 19 Ø Ø.86PF C60 7 Ø Ø.271PF C61 13 Ø Ø.134PF C62 12 Ø Ø.86PF C63 3 Ø Ø.50PF VAØ 31 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VBØ 22 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 24 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS3 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS3 18 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VPI BAR 8 Ø PULSE (5V 6V ANS ØNS ØNS 2NS) VPIBAR 8 Ø PULSE (5V 6V 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(13) V(26) (ØV,5V) .END .

<u>e de contra</u>



Figure C-15. SPICE Plot Section 1 Input 010 Four Node Precharge with Basic Gate Widths

والمراجع فيراجع فيراجع فيراجع فيتعاد

C-34

TEMPERATURE =

27.000 DEG C

.WIDTH OUT=8Ø .OPTIONS ITL1=5#Ø ITL5=Ø MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC M1 1 3 2 Ø M2 1 3 5 Ø 5V NMOS L=5.ØU W=5.ØU NMOS L=5.ØU W=5.ØU M2 1 3 5 Ø NMOS L=5.ØU W=5.ØU M3 1 3 6 Ø NMOS L=5.ØU W=5.ØU M4 1 3 7 Ø NMOS L=5.ØU W=5.ØU M5 2 8 1 1 PMOS L=5.ØU W=1Ø.ØU M6 5 8 1 1 PMOS L=5.ØU W=1Ø.ØU M7 6 8 1 1 PMOS L=5.ØU W=1Ø.ØU M8 7 8 1 1 PMOS L=5.ØU W=1Ø.ØU M10 10 9 7 Ø NMOS L=5.ØU W=1Ø.ØU M11 12 11 10 Ø NMOS L=5.ØU W=1Ø.ØU M12 1 11 7 1 PMOS L=5.ØU W=1Ø.ØU M13 1 7 13 1 PMOS L=5.ØU W=1Ø.ØU M14 Ø 14 12 Ø NMOS L=5.ØU W=1Ø.ØU M15 16 6 Ø Ø NMOS L=5.ØU W=1Ø.ØU M16 16 7 13 Ø NMOS L=5.ØU W=1Ø.ØU M17 1 6 13 1 PMOS L=5.ØU W=10.ØU M18 1 4 7 1 PMOS L=5.ØU W=10.ØU M18 1 4 7 1 PMOS L=5.ØU W=10.ØU M18 1 4 7 1 PMOS L=5.ØU W=10.ØU M19 1 4 6 Ø NMOS L=5.ØU W=10.ØU M20 17 14 6 Ø NMOS L=5.ØU W=3Ø.ØU M21 19 18 17 Ø NMOS L=5.ØU W=3Ø.ØU M22 1 13 6 1 PMOS L=5.ØU W=10.ØU M23 Ø 2Ø 19 Ø NMOS L=5.ØU W=10.ØU M24 22 21 11 Ø NMOS L=5.ØU W=10.ØU M25 1 2Ø 6 1 PMOS L=5.ØU W=10.ØU M26 22 23 11 1 FMOS L=5.ØU W=10.ØU M26 22 23 11 1 FMOS L=5.ØU W=10.ØU M27 Ø 11 2Ø Ø NMOS L=5.ØU W=10.ØU M28 1 11 20 Ø NMOS L=5.ØU W=10.ØU M29 1 11 20 Ø NMOS L=5.ØU W=10.ØU M29 1 11 20 Ø NMOS L=5.ØU W=10.ØU M20 17 14 6 Ø NMOS L=5.ØU W=10.ØU M26 22 23 11 1 FMOS L=5.ØU W=10.ØU M26 22 23 11 1 FMOS L=5.ØU W=10.ØU M26 22 23 11 1 FMOS L=5.ØU W=10.ØU M27 Ø 11 20 Ø NMOS L=5.ØU W=10.ØU M28 1 11 20 0 PMOS L=5.ØU W=10.ØU M29 0 11 20 0 PMOS L=5.ØU W=10.ØU M29 0 11 20 0 PMOS L=5.ØU W=10.ØU M29 0 11 20 0 PMOS L=5.ØU W=10.ØU M29 0 11 20 0 PMOS L=5.ØU W=10.0U M29 0 11 20 0 PMOS L=5.ØU W=10.0U M29 0 11 20 0 PMOS L=5.ØU W=10.0U M29 0 11 20 0 PMOS L=5.ØU W=10.0U M29 0 11 20 0 PMOS L=5.ØU W=10.0U M29 0 11 20 0 PMOS L=5.ØU W=10.0U M29 0 11 20 0 PMOS L=5.ØU W=10.0U M29 0 11 20 0 PMOS L=5.ØU W=10.0U M20 0 PMOS NMOS L=5.ØU W=5.ØU M3 3637 ž ø 1 M27 Ø 11 2Ø Ø NMOS L=5.ØU W=15.ØU M28 1 11 2Ø 1 PMOS L=5.ØU W=3Ø.ØU M29 1 20 5 1 PMOS L=5.00 W=10.00 M30 25 24 0 0 NMOS L=5.00 W=15.00 M31 25 20 5 0 NMOS L=5.00 W=15.00 5 26 1 PMOS L=5.00 W=10.00 24 5 1 PMOS L=5.00 W=10.00 M32 1 M33 27 5 26 Ø NMOS L=5.ØU W=3Ø.ØU Ø 23 14 Ø NMOS L=5.ØU W≠15.ØU M34 M35 29 2 27 Ø NHOS L=5.ØU W=30.ØU 1 23 14 1 PMOS L=5.ØU W=30.ØU 1 2 26 1 PMOS L=5.ØU W=10.ØU 1 30 2 1 PMOS L=5.ØU W=10.ØU 29 M36 M37 M38 M39 M4Ø 31 21 28 Ø NMOS L=5.ØU W=5.ØU M41 Ø 23 29 Ø NMOS L=5.ØU W=3Ø.ØU M42 32 11 Ø Ø NMOS L=5.ØU W=15.ØU 2 Ø NMOS L=5.ØU W=15.ØU 2 1 PMOS L=5.ØU W=10.ØU M43 32 ЗØ 1 11 2 M44 31 23 28 1 FMOS L=5.ØU W=10.ØU 1 23 26 1 PMOS L=5.ØU W=10.ØU M45 M46 1 J Ø.16Ø2PF C47

INPUT LISTING

C48 2C Ø Ø.207PF C49 21 Ø Ø.51PF C59 Ø Ø Ø.1174PF C51 11 Ø Ø.297PF C52 26 Ø Ø.134PF C53 2 Ø Ø.274PF C55 2 Ø Ø.283PF C55 14 Ø Ø.283PF C57 2Ø Ø Ø.151PF C58 6 Ø Ø.303PF C59 19 Ø Ø.86PF C60 7 Ø Ø.271PF C61 13 Ø Ø.134PF C62 12 Ø Ø.86PF C62 3 Ø Ø.50PF VAØ 31 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPH11 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPH11 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 3Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 3Ø Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 18 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS 2NS) VF1 5 ØNS ØNS ØNS ØNS ØNS 2NS) VF1 5 ØNS ØNS ØNS ØNS ØNS ØNS ØNS ØNS 2NS) VF1 5 ØNS ØNS ØNS ØNS ØNS ØNS ØNS ØNS 2NS) VF1 5 ØNS ØNS ØNS ØNS ØNS ØNS ØNS ØNS ØNS 0 0 0 0 0 0 0 0 0 0 0 0

.

. . .



.

•

Figure C-16. SPICE Plot Section 1 Input 101 Four Node Precharge With Basic Widths

 1*******12/Ø1/84
 ********ØØ:34:49*****

 Ø
 CMOS/SOS BITSLICE SECTION 1 WITH FOUR NODES PRECHARGED

 Ø*****
 INPUT LISTING

 TEMPERATURE =
 27.000 DEG C

.WIDTH OUT=8Ø .OPTIONS ITL1=5#Ø ITL5=Ø MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NM M2 1 3 5 Ø NM NMOS L=5.00 W=10.00 NMOS L=5.00 W=10.00 NMOS L=5.00 W=10.00 NMOS L=5.00 W=10.00 1360 M3 NMOS L=5.ØU PMOS L=5.ØU W=10.0U MΔ 137Ø M5 28 W=2Ø.ØU 1 1 PMOS L=5.80 W=28.80 PMOS L=5.80 W=28.80 PMOS L=5.80 W=28.80 PMOS L=5.80 W=28.80 PMOS L=5.80 W=18.80 MG 58 1 1 6 8 M7 1 1 M8 78 1 1 7 Μ9 19 1 M1Ø M11 M12 M13 M14 Ø 14 12 Ø NMOS L=5.ØU W=3Ø.ØU M15 16 6 Ø Ø NMOS L=5.ØU W=15.ØU M16 16 7 13 Ø NMOS L=5.ØU W=15.ØU 7 13 Ø NMOS L=5.ØU W=15.ØU 13 1 PMOS L=5.ØU W=1Ø.ØU 1 G M17 1 14 7 1 PMOS L=5.ØU W=1Ø.ØU 1 14 6 1 PMOS L=5.ØU W=1Ø.ØU M18 M19 M19 I 14 6 I FMOS L=5.00 W=10.20 M20 17 14 6 \emptyset NMOS L=5.00 W=30.00 M21 19 13 17 \emptyset NMOS L=5.00 W=30.00 M22 1 18 6 1 PMOS L=5.00 W=10.00 M23 \emptyset 20 19 \emptyset NMOS L=5.00 W=30.00 M24 22 21 11 \emptyset NMOS L=5.00 W=5.00 W=5.00 W=5.00 M25 L=5.00 W=5.00 5 1 PMOS L=5.ØU W=10.0U 11 1 FMOS L=5.0U W=10.0U M25 1 20 6 M26 22 23 Ø 11 20 Ø NMOS L=5.00 W=15.00 1 11 20 1 PMOS L=5.00 W=30.00 M27 M28 1 20 5 1 PMOS L=5.00 W=10.00 M29 25 24 Ø Ø NMOS L=5.ØU W=10.ØU 25 2Ø 5 Ø NMOS L=5.ØU W=15.ØU 1 5 26 1 PMOS L=5.ØU W=10.ØU 1 24 5 1 PMOS L=5.ØU W=10.ØU мзø M31 M32 M33 M34 27 5 26 Ø NMOS L=5.ØU W=3Ø.ØU M35 Ø 23 14 Ø NMOS L=5.ØU W=15.ØU 27 Ø NMOS L=5.ØU W=3Ø.JU 14 1 PMOS L=5.ØU W=3Ø.JU M36 29 2 1 20 14 1 PMOS L=5.00 W=30.00 1 2 26 1 PMOS L=5.00 W=10.00 1 30 2 1 PMOS L=5.00 W=10.00 1 30 2 1 PMOS L=5.00 W=10.00 M37 M38 M39

C-38

C48 28 Ø Ø.207PF C39 21 Ø Ø.51PF C50 Ø Ø Ø.1174?F C51 11 Ø Ø.297PF C52 26 Ø Ø.134PF C53 2 Ø Ø.274PF C54 29 Ø Ø.36PF C55 14 Ø Ø.202PF C56 5 Ø Ø.283PF C59 19 Ø Ø.86PF C59 19 Ø Ø.86PF C60 7 Ø Ø.271PF C61 13 Ø Ø.134PF C63 3 Ø Ø.50PF VAØ 31 Ø PULSE (ØV SV ØNS ØNS ØNS 1ØNS) VBØ 22 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1BAR 23 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS1 24 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (SV SV ØNS ØNS ØNS 1ØNS) VS3 18 Ø PULSE (SV SV ØNS ØNS ØNS 1ØNS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) VPIBAR 3 Ø PULSE (SV ØV 4NS ØNS ØNS ØNS 2NS) SO SONS 2NS 2NS) SONS 2NS 2NS 2NS 2NS) SONS 2NS 2NS 2NS 2NS) SONS 2NS 2NS 2NS 2NS 2NS) SONS 2NS 2NS 2NS 2NS 2NS 2NS) SONS 2NS 2NS 2NS 2NS 2NS 2NS 2NS 2NS 2NS) SONS 2NS 2NS 2NS

> > .

.

D

Ĭ

•••

ا از معرف از معالم می باشد. محمد از معالم معرف می موجود از معالم معرف محمد معرف معرف معرف معرف معرف معرف معرف مع



-

• `•

·'.

Figure C-17. SPICE Plot Section 1 Input 010 Four Node Precharge with Gate Widths x2

C-40

 1*******12/Ø1/84
 *********Ø5:46:23*****

 Ø
 CMOS/SOS BITSLICE SECTION 1 WITH FOUR NODES PRECHARGED

 Ø*****
 INPUT LICTING

 TEMPERATURE =
 27.ØØØ DEG C

۰.

.WIDTH OUT=8Ø .OPTIONS ITL1=580 ITL5=0 MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL= .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU ₩=1Ø.ØU M2 1 3 5 Ø NMOS L=5.ØU W=1Ø.ØU M3 1 3 6 Ø NMOS L=5.ØU W=1Ø.ØU 3 7 Ø NMOS L=5.ØU W=1Ø.ØU M4 2811 PMOS L=5.00 W=20.00 M5 58 1 1 PMOS L=5.00 W=20.00 Мб M7 6 8 1 1 PMOS L=5.00 W=20.00 78 1 1 PMOS L=5.00 W=20.00 **M8** 1 9 7 1 PMOS L=5.ØU W=1Ø.ØU 10 9 7 Ø NHOS L=5.ØU W=3Ø.ØU 19 M9 MIØ M11 12 11 10 0 NMOS L=5.00 W=30.00 M11 12 11 10 0 NMOS L=5.0U W=30.00 M12 1 11 7 1 PMOS L=5.0U W=10.0U M13 1 7 13 1 PMOS L=5.0U W=10.0U M14 0 14 12 0 NMOS L=5.0U W=30.0U M15 16 6 0 0 NMOS L=5.0U W=15.0U M16 16 7 13 0 NMOS L=5.0U W=15.0U M16 16 7 13 0 NMOS L=5.0U W=10.0U M17 1 G 13 1 PMOS L=5.0U W=10.0U M19 1 14 6 1 PMOS L=5.0U W=10.0U M20 1 7 14 6 0 PMOS L=5.0U W=30.0U N.1311471PMGSL=5.00W=10.00M1911461PMOSL=5.00W=10.00M20171460NMOSL=5.00W=30.00M211913170NMOSL=5.00W=30.00M211913170NMOSL=5.00W=30.00M2211861PMOSL=5.00W=30.00M23020190NMOSL=5.00W=30.00M242221110NMOSL=5.00W=10.00M2512061PMOSL=5.00W=10.00M262223111FMOSL=5.00W=10.00M2701170NMOSL=5.00W=10.00M23111201PMOSL=5.00W=10.00M3025240NMOSL=5.00W=15.00M3125261PMOSL=5.00W=10.00M3212451PMOSL=5.00W=10.00M3312451PMOSL=5.00W=30.00M331261PMOSL=5.00W=30.00M331261PMOSL=5.00W=30.00M331261PMOSL=5.00W=10.00M331261PMOS</td M43 32 30 2 0 MMOS L=5.00 W=15.00 M44 1 11 2 1 PMOS L=5.00 W=10.00 M45 31 23 28 1 FMOS L=5.00 W=10.00 M46 1 23 25 1 PMOS L=5.00 W=10.00 C47 1 ΰ Ø.1602PF C47

C43 23 Ø Ø.2072F C49 21 Ø Ø.51PF C50 Ø Ø Ø.1174PF .END

••••

7

----.•



• •

Figure C-18. SPICE Plot Section 1 Input 101 Four Node Precharge with Widths x2

C-43

• *

INPUT LISTING TEMPERATURE = .WIDTH OUT=8Ø OPTIONS ITL1=5&Ø ITL5=Ø MODEL NMOS NHOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL= .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=15.ØU M2 1 35Ø NMOS L=5.ØU W=15.ØU MЗ 1 3 6 Ø NMOS L=5.ØU W=15.ØU 37Ø NMOS L=5.ØU W=15.ØU Μ4 1 M5 2811 PMOS L=5.ØU W=3Ø.ØU 8 1 1 PMOS 8 1 1 PMOS L=5.ØU W=3Ø.ØU L=5.ØU W=3Ø.ØU 5811 М6 MG 5 8 1 1 PMOS L=5.00 W=30.00 M7 6 8 1 1 PMOS L=5.00 W=30.00 M8 7 8 1 1 PMOS L=5.00 W=30.00 M9 1 9 7 1 PMOS L=5.00 W=30.00 M10 10 9 7 0 NMOS L=5.00 W=30.00 M11 12 11 1 0 0 NMOS L=5.00 W=30.00 M12 1 11 7 1 PMOS L=5.00 W=10.00 M13 1 7 13 1 PMOS L=5.00 W=10.00 M14 0 14 12 0 NMOS L=5.00 W=10.00 M15 16 6 0 0 NMOS L=5.00 W=15.00 M16 16 7 13 0 NMOS L=5.00 W=15.00 M17 1 6 13 1 PMOS L=5.00 W=10.00 M18 1 14 7 1 PMOS L=5.00 W=10.00 M19 1 14 6 1 PMOS L=5.00 W=10.00 M19 1 14 6 1 PMOS L=5.00 W=10.00 M20 17 14 6 0 NMOS L=5.00 W=10.00 M21 19 13 17 0 NMOS L=5.00 W=30.00 M22 1 18 6 1 PMOS L=5.00 W=30.00 M23 0 20 19 0 NMOS L=5.00 W=30.00 M24 22 21 11 0 NMOS L=5.00 W=30.00 M25 1 20 6 1 PMOS L=5.00 W=10.00 M26 22 23 11 1 FMOS L=5.00 W=10.00 M27 0 11 20 0 NMOS L=5.00 W=10.00 M28 1 11 20 1 PMOS L=5.00 W=10.00 M29 1 20 5 1 PMOS L=5.00 W=10.00 M29 1 20 5 1 PMOS L=5.00 W=10.00 M29 1 20 5 1 PMOS L=5.00 W=10.00 M29 1 20 5 1 PMOS L=5.00 W=10.00 M29 1 20 5 1 PMOS L=5.00 W=10.00 M29 1 20 5 1 PMOS L=5.00 W=10.00 M20 12 10 5 1 PMOS L=5.00 W=10.00 M20 12 10 5 1 PMOS L=5.00 W=10.00 M20 12 10 6 1 PMOS L=5.00 W=10.00 M20 12 10 6 1 PMOS L=5.00 W=10.00 M20 12 10 6 1 PMOS L=5.00 W=10.00 M20 12 10 6 1 PMOS L=5.00 W=10.00 M20 12 0 7 11 20 0 NMOS L=5.00 W=10.00 M20 12 0 7 11 20 0 NMOS L=5.00 W=10.00 M20 12 0 7 10 PMOS L=5.00 W= M7 õ 1 20 1 PMOS L=5.00 W=30.00 1 20 5 1 PMOS L=5.00 W=10.00 25 24 0 0 NMOS L=5.00 W=15.00 25 20 5 0 NMOS L=5.00 W=15.00 1 5 26 1 PMOS L=5.00 W=10 00 1 5 26 1 PMOS L=5.00 W=10 00 M29 мзø M31 1 PHOS L=5.00 W=10.00 1 PHOS L=5.00 W=10.00 1 PHOS L=5.00 W=10.00 M32 M33 1 24 5 27 5 26 Ø NMOS L=5.00 W=30.00 Ø 23 14 Ø NMOS L=5.00 W=15.00 M34 29 2 27 Ø NMOS L=5.ØU W=15.ØU 1 28 14 1 PMOS L=5.ØU W=3Ø.ØU 1 2 26 1 PMOS L=5.ØU W=3Ø.ØU 1 2 26 1 PMOS L=5.ØU W=3Ø.ØU M35 29 M36 M37 26 1 PMOS L≈5.00 W=10.00 0 2 1 PMOS L≈5.00 W=10.00 21 28 0 NMOS L=5.00 W=10.00 21 28 0 NMOS L=5.00 W≈5.00 M38 M39 3Ø 2 1 Ø Ø.16Ø2PF C47

*****12/Ø1/84 ******* SPICE 2G.1 (150CT8Ø) *******Ø5:49:3Ø* CMOS/SOS BITSLICE SECTION 1 WITH FOUR NODES PRECHARGED

27.000

ø

Į.

•

CAS 23 Ø Ø.207PF CA9 21 Ø Ø.51PF CA9 21 Ø Ø.51PF CA9 21 Ø Ø.51PF CA9 21 Ø Ø.51PF CA9 21 0 Ø.1174PF CA9 20 Ø 0.127PF CA9 20 Ø.284PF C52 26 Ø Ø.283PF C55 14 Ø Ø.282PF C56 5 Ø Ø.283PF C59 19 Ø Ø.86PF C69 7 Ø Ø.271PF C61 13 Ø Ø.134PF C62 12 Ø Ø.86PF C63 3 Ø Ø.50PF VAØ 31 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VBØ 22 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI 24 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS1 24 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS3 18 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VPIBAR 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(13) V(26) (ØV,5V) .END

C-45

. De server en entre en entre en entre en entre en entre en entre en entre en entre en entre en entre en entre en

•.

. . .

-

- -

.

1.

•

1... CHOS/SOS BITSLICE SECTION 1 WITH FOUR NODES PRECHARGED #\$ TRANSIENT ANALYSIS TEMPERATURE + 27.888 DEG C *: V(13) *: V(26) TIME V(13))----d+66 1.25#d+## 2.5##d+## 3.75#d+## 5.888d+88
 d+##

 5.##d-#3

 1.##d-#3

 2.##d-#3

 2.##d-#3

 3.##d-#3

 3.##d-#3

 3.##d-#3

 3.##d-#3

 3.##d-#3

 3.##d-#3

 3.##d-#3

 5.#dd-#3

 5.#dd-#3

 5.#dd-#3

 5.#dd-#3

 5.#dd-#3

 7.5#d-#3

 9.##d-#3

 7.5#d-#3

 9.##d-#3

 9.##d-#3

 9.##d-#3

 9.##d-#3

 9.##d-#3

 9.##d-#3

 9.##d-#3

 1.##d-#3

 1. SES4+88

.

Figure C-19. SPICE Plot Section 1 Input 010 Four Node Precharge with Widths x3 1******12/01/84 ******** SPICE 2G.1 (150CT80) *******05:54:47* CMOS/SOS BITSLICE SECTION 1 WITH FOUR NODES PRECHARGED а 12**** INPUT LISTING TEMPERATURE = 27.000 .WIDTH OUT=8Ø .OPTIONS ITL1=5## ITL5=# .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=4000 NSUB=2.5E16 LD=0.7UM) +LEVEL= .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NM M2 1 3 5 Ø NM NMOS L=5.00 V=15.00 NMOS L=5.ØU W=15.ØU NMOS L=5.ØU W=15.ØU M3 1 36Ø 3 7 Ø NMOS L=5.00 W=15.00 ΜÂ 1

 M5 1
 S 1
 PMOS L=5.00 W=30.00

 M6 5
 S 1
 PMOS L=5.00 W=30.00

 M7 6
 S 1
 PMOS L=5.00 W=30.00

 M8 7
 S 1
 PMOS L=5.00 W=30.00

 M9 1
 9
 7
 PMOS L=5.00 W=30.00

 M10
 10
 9
 7
 PMOS L=5.00 W=30.00

 M11
 12
 11
 10
 0
 MMOS L=5.00 W=30.00

 M12
 11
 7
 1
 PMOS L=5.00 W=30.00

 M13
 1
 7
 13
 PMOS L=5.00 W=10.00

 M14
 0
 14
 12
 0
 NMOS L=5.00 W=10.00

 M15
 16
 6
 0
 NMOS L=5.00 W=10.00
 MI0

 M15
 14
 7
 1
 PMOS L=5.00 W=10.00
 MI0

 M12
 13
 1
 PMOS L=5.00 W=10.00
 MI0
 MI0

 M20
 17
 14
 6
 L=5.ØU W=3Ø.ØU L=5.ØU W=3Ø.ØU L=5.ØU W=3Ø.ØU M5 2811 PMOS 5811 PMOS M6 1 28 14 1 PMOS L=5.00 W=30.00 1 2 26 1 PMOS L=5.00 W=10.00 1 30 2 1 PMOS L=5.00 W=10.00 1 30 2 1 PMOS L=5.00 W=10.00 M37 M38 M39 31 21 Ø 23 2 23 Ø NMOS L=5.ØU W=5.ØU M4Ø 29 Ø NMOS L=5.00 W=30.00 M 4 1 22 11 3 0 NHOS L=5.00 W=15.00 22 11 3 0 NHOS L=5.00 W=15.00 1 11 2 1 PHOS L=5.00 W=10.00 31 23 28 1 FMOS L=5.00 W=10.00 M42 MAG M44 28 1 FMOS L=5.ØU W=10.ØU M45 1 23 26 1 PMOS L=5.00 W=10.00 1 0 0.1602PF M46 C47

1.15

ý

C48 28 Ø Ø.207?F C49 21 Ø Ø.51PF C5Ø Ø Ø Ø.1174PF C51 11 Ø Ø.297PF C52 26 Ø Ø.134PF C52 26 Ø Ø.134PF C53 2 Ø Ø.274PF C55 14 Ø Ø.202PF C56 5 Ø Ø.283PF C57 2Ø Ø Ø.151PF C59 19 Ø Ø.303PF C59 19 Ø Ø.303PF C60 7 Ø Ø.271PF C61 13 Ø Ø.134PF C62 12 Ø Ø.86PF C63 3 Ø Ø.50PF VAØ 31 Ø PULSE (5V ØV ØNS ØNS ØNS 1ØNS) VBØ 22 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI1 21 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VPHI1 21 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VPHI12AR 23 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSØ 30 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS2 9 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VS3 18 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØV 4NS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V ØNS ØNS ØNS ØNS ØNS 2NS) VP1 3 Ø PULSE (5V 5V 6V 5V 5V 5V 5V) END

.

۰.



Figure C-20. SPICE Plot Section 1 Input 101 Four Node Precharge with Widths x3

C-49

1.1.1

.

Figure C-21. Basic Section 2 CLL Plot.



Figure C-22. Basic Section 2 Node Plot.

<u>Table</u> <u>C-4</u>

BITSLICE BASIC SECTION 2 NODE REFERENCE LIST

GND Vdd NMOS	9 1 9
PMOS	ĩ
28	2
7	3
1	4
4	5
3.0	6
4/	
2/	8
3	1 07
13	11
9	12
17	13
• •	

.

F

.WIDTH OUT=80 .OPTIONS ITLI=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 0 DC 5V M1 1 3 2 1 PMOS L=5.00 W=10.00 M2 7 6 0 0 NMOS L=5.00 W=15.00 M4 1 6 2 1 PMOS L=5.00 W=15.00 M4 1 6 2 1 PMOS L=5.00 W=15.00 M5 1 2 8 1 PMOS L=5.00 W=15.00 M6 0 9 6 0 NMOS L=5.00 W=15.00 M7 11 10 0 0 NMOS L=5.00 W=15.00 M8 11 2 8 0 NMOS L=5.00 W=15.00 M1 1 3 10 L 5.00 W=15.00 M1 1 3 10 L 5.00 W=15.00 M12 13 9 0 0 NMOS L=5.00 W=15.00 M13 1 3 12 10 0 NMOS L=5.00 W=15.00 M14 1 9 10 1 PMOS L=5.00 W=15.00 M15 0 3 12 0 NMOS L=5.00 W=15.00 M15 0 3 12 0 NMOS L=5.00 W=15.00 M15 0 3 12 0 NMOS L=5.00 W=15.00 M16 1 3 12 1 PMOS L=5.00 W=15.00 M17 1 0 0 0 1 PMOS L=5.00 W=15.00 M18 1 3 0 0 1 PMOS L=5.00 W=15.00 M13 1 3 12 10 0 NMOS L=5.00 W=15.00 M14 1 9 10 1 FMOS L=5.00 W=15.00 M15 0 3 12 0 NMOS L=5.00 W=15.00 M15 0 3 12 0 NMOS L=5.00 W=15.00 M16 1 3 12 1 PMOS L=5.00 W=15.00 M17 1 0 0 0 1.83PF C21 12 0 0 0.183PF C21 12 0 0.183PF C21 12 0 0.183PF C23 8 0 0.183PF C24 2 0 0.183PF C25 6 0 0.131PF V1M1 3 0 PULSE (00 5V 5NS 0NS 0NS 10NS) V1M2 9 0 PULSE (00 5V 5NS 0NS 0NS 10NS) .TRAH 0.5NS 40NS .PLOT TRAH V(3) (00V,5V) .END

.

•

D

 1*******12/Ø1/84
 ********Ø2:56:27*****

 Ø
 CMOS/SOS BASIC BITSLICE SECTION 2 TRANSIENT ANALYSIS

 Ø****
 INPUT LISTING

 TEMPERATURE =
 27.ØØØ DEG C

.

••• TI	ANSIENT ANA	ALYSIS	7	EMPERATURE -	27. 888 DEG C	
********	*********					•
TIME	V(8)					
	₽.	d+ 55	1.25#d+##	2.5 88 d+88	3.75#d+##	5. <i>888</i> d+8
#. d+##	5.888d+88					
5. <i>000</i> d-1 <i>0</i> 1. <i>000</i> d-109	5. <i>888</i> d+ <i>88</i> 5. <i>888</i> d+88	:	:	:	:	
1.588d-#9	5.888d+88 5.888d+88	•	•	•	•	:
2.5884-89	5.888d+88	•	:	:	:	
3.5681-89	5.###d+##	:	:	•	•	-
4.889d-89 4.589d-89	5. <i>989</i> d+88 5. <i>989</i> d+88	•	:	:	:	:
5.888d-89	5.828d+88	• •	•	•	•	:
6.888d-89	5.888d+88	:			:	•
7.808d-89	4.996d+88	:	:	:	:	:
7.5 <i>08</i> 0-#9 8. <i>908</i> 0-#9	4.937d+## 4.768d+##	:	:	•	•	· • *:
8.508d-89	4.438d+68	•	•	•		•
9.5884-89	3.428d+##	:	•	: .	•	:
1.#5#d-#8	2.81#d+## 2.145d+##	:	:	•	:	:
1.189d-88	1.547d+88	•	• •	•	•	•
1.200d-#8	6.5#7d-#1	÷ •	:	:	:	:
1.3500-08	2.136d-01	֥ "	:	:	:	:
1.350d-08 1.400d-08	1.238d-#1 6.7##d-#2	:	:		:	•
1.4584-88	3.736d-82	•	•		•	:
1.5584-88	1.988d-82	*	:	:	:	:
1.65#d-#8	6.149d-#3 3.495d-#3	:	:	:	:	:
1.7884-88	2.1210-03	:	•	•	•	
1.8##d-#8	1.1##d-#1		:	:	:	:
1.85#d~#8 1.9##d~#8	3.2210-01	÷ •	•	:		:
1.958d~88	1.1684+98	•	•	•	•	•
2.858d-88	2.3614+88	:	:	• :		:
2.150d-08	2.9520*88 3.465d+88	•	•	: -	• :	:
2.2884-88 2.2504-88	3.874d+98 4.161d+88	:	:	•		•
2.3084-88	4.4364+88			:	•	• .
2.4084-88	4.725d+8#	:	:	:	:	•
2.4580-88	4.017d+8#	•	•	•	•	•.
2.5584-98	4.9144+88	:	:	:	:	
2.65#d~#8 2.65#d~#8	4.942d+## 4.96#d+##	:	:	:	:	*:
2.7084-08	4.974d+##	•	•	•		•
2.000-08	4.968d+88	:	•	:	:	
2.05#d-#8 2.97#d-#8	4.992d+ 88 4.994d+88	:	:	:	•	
2.3504-08	4.996d+88	•	:	•	•	:
3.050.1-00	4.998d+08	:	:	:	:	•
3.15#d-#8	4.999d+## 4.999d+##	:	:	:	:	*
3.200J-08 3.2504-08	4.999d+0# 5.###d+0#	•	•	•	•	
3.300-1-00	5.008d+00			:	:	•
3.400d-00	2.0664+08 2.0664+08	:	:	:	•	
3.4584-88 3.5884-88	5.88884+88 5.8884+88	:	:	•	:	:
3.5501-08	5.009d+00	•		:		•
3.650d-#8	5.000d+00 5.000d+00	•	:	÷	:	
3.7884-88	5.888d+88	•	•	•	•	:
3.800-1-08	5.800d+88	:	:	:	•	:
3.0000-00 3.0000-00	5.888d+88 5.888d+88	:	:	:	•	:
3.9504-08	5.888d+88					•

Figure C-23. SPICE Plot Basic Section 2 Input 010.

.

ď

Ľ

.WIDTH OUT=80 .OPTIONS ITL1=520 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 0 DC 5V M1 1 3 2 1 PMOS L=5.00 W=10.00 M2 7 6 0 0 NMOS L=5.00 W=15.00 M4 1 6 2 1 PMOS L=5.00 W=15.00 M4 1 6 2 1 PMOS L=5.00 W=15.00 M5 1 2 8 1 PMOS L=5.00 W=15.00 M6 0 9 6 0 NMOS L=5.00 W=15.00 M7 11 10 0 0 NMOS L=5.00 W=15.00 M7 11 10 0 0 NMOS L=5.00 W=15.00 M10 1 9 6 1 PMOS L=5.00 W=15.00 M11 1 10 8 1 PMOS L=5.00 W=10.00 M12 13 9 0 0 NMOS L=5.00 W=15.00 M13 13 12 10 0 NMOS L=5.00 W=15.00 M14 1 9 10 1 PMOS L=5.00 W=15.00 M15 0 3 12 0 NMOS L=5.00 W=15.00 M16 1 3 12 1 PMOS L=5.00 W=15.00 M16 1 3 12 1 PMOS L=5.00 W=30.00 C17 1 0 0.051PF C20 3 0 0.183PF C21 12 0 0.116PF C23 10 0.116PF C23 10 0.116PF C23 10 0.116PF C23 10 0.116PF C24 2 0.113PF VIN1 3 0 PULSE (5V 5V 5NS 0NS 0NS 10NS) V1N2 9 0 PULSE (5V 5V 5NS 0NS 0NS 10NS) V1N2 9 0 PULSE (5V 5V 5NS 0NS 0NS 10NS) .TRAM 0.5NS 40NS .PLOT TRAM V(0) (0V.5V) END



••••

والمعادية والمعاد والمعاد والمعاد والمعاد والمعاد والمعاد والمعاد والمعاد والمعاد والمعاد والمعاد والمعاد والم

Ţ

Figure C-24. SPICE Plot Basic Section 2 Input 101.

C-55

۰.



Figure C-25. Section 2 One Node Precharge CLL Plot



Figure C-26. Section 2 One Node Precharge Node Plot.

<u>Table</u> <u>C-5</u>

BITSLICE SECTION 2 NODE REFERENCE LIST - ONE NODE PRECHARGE

C-57

GND Vdd NMOS PMOS 27 62 1 58	Ø 1 9 1 2 3 4 5
28 7	6 7
4	8
47	- 1ø
3	11
32	13
9	14
17	15

•

1******12/Ø1/84 ******* SPICE 2G.1 (150CT8Ø) *******Ø4:46:26***** Ø CMOS/SOS BITSLICE SECTION 2 PRECHARGED AT ONE NODE Ø**** INPUT LISTING TEMPERATURE = 27.000 DEG C

.WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VOD I Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=5.ØU M2 2 5 1 1 PMOS L=5.ØU W=1Ø.ØU M3 1 7 6 1 PMOS L=5.ØU W=1Ø.ØU M4 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M5 1Ø 7 6 Ø NMOS L=5.ØU W=15.ØU M6 1 9 6 1 PMOS L=5.ØU W=1Ø.ØU M8 Ø 11 9 Ø NMOS L=5.ØU W=15.ØU M1 1 3 5 2 Ø NMOS L=5.ØU W=15.ØU M10 13 5 2 Ø NMOS L=5.ØU W=30.ØU +LEVEL=1 M10 13 6 2 ϑ NMOS L=5. ϑ U W=15. ϑ U M11 1 11 9 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M12 1 12 2 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M13 1 14 12 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M14 15 11 ϑ ϑ NMOS L=5. ϑ U W=15. ϑ U M15 15 14 12 ϑ NMOS L=5. ϑ U W=15. ϑ U M16 1 11 12 1 PMOS L=5. ϑ U W=15. ϑ U M17 ϑ 7 14 ϑ NMOS L=5. ϑ U W=15. ϑ U M18 1 7 14 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U C19 1 ϑ ϑ .1 ϑ 74PF C2 ϑ 11 ϑ ϑ .279PF C21 ϑ ϑ ϑ 51PF C21 Ø Ø Ø.951PF C22 7 Ø Ø.183PF C22 C23 14 Ø Ø.1ØØPF 12 Ø Ø.116PF 2 Ø Ø.234PF C24 C25 C26 6 Ø Ø.125PF C27 9 Ø Ø.1Ø1PF C28 5 0 Ø.103PF C29 3 0 Ø.103PF VIN1 11 Ø PULCE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 Ø PULCE (ØV 5V 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULCE (ØV 5V 4NS ØNS ØNS 2NS) VP1EAR 5 Ø PULCE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.505 4005 .PLOT TRAN V(2) (ØV.5V) . END


.

· • •

. ...

- - --

Figure C-27. SPICE Plot Section 2 Input 010 One Node Precharge with Basic Gate Widths

1

. 7

.MODEL PMOS PMOS (VIO=-IV IOX=75NM 00=200 NS +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=18.ØU M2 2 5 1 1 PMOS L=5.ØU W=18.ØU M3 1 7 6 1 PMOS L=5.ØU W=15.ØU M5 1Ø 7 6 Ø NMOS L=5.ØU W=15.ØU M5 1Ø 7 6 Ø NMOS L=5.ØU W=15.ØU M6 1 9 6 1 PMOS L=5.ØU W=15.ØU M7 1 6 2 1 PMOS L=5.ØU W=15.ØU M8 Ø 11 9 Ø NMOS L=5.ØU W=15.ØU M1Ø 13 6 2 Ø NMOS L=5.ØU W=15.ØU M1Ø 13 6 2 Ø NMOS L=5.ØU W=15.ØU M10 13 6 2 Ø NMOS L=5.ØU W=15.ØU M11 1 11 9 1 PMOS L=5.ØU W=15.ØU M12 1 2 2 1 PMOS L=5.ØU W=15.ØU M13 1 14 12 1 PMOS L=5.ØU W=16.ØU M14 15 11 Ø Ø NMOS L=5.ØU W=16.ØU M15 15 14 12 Ø NMOS L=5.ØU W=15.ØU M16 1 11 12 1 PMOS L=5.ØU W=15.ØU M17 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M18 1 7 14 1 PMOS L=5.ØU W=15.ØU M18 1 7 14 1 PMOS L=5.ØU W=3Ø.ØU C19 1 Ø Ø.1874PF C20 11 Ø Ø.951PF C22 7 Ø Ø.183PF C23 14 Ø Ø.100PF C24 12 Ø Ø.116FF C25 2 Ø Ø.234PF C26 6 Ø Ø.125PF C27 9 Ø Ø.103PF VIN1 11 Ø PULSE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (5V 6V 4NS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (5V 6V 4NS ØNS ØNS 2NS) VIN2 7 Ø PULSE (5V 6V 4NS ØNS ØNS 2NS) VIN2 7 Ø PULSE (5V 6V 4NS ØNS ØNS 2NS) VIN2 7 Ø PULSE (5V 6V 4NS ØNS ØNS 2NS) VIN2 7 Ø PULSE (5V 6V 4NS ØNS ØNS 2NS) VIN2 7 Ø PULSE (6V 5V 5V) .END



. . . .

Figure C-28. SPICE Plot Section 2 Input 101 One Node Precharge with Basic Widths -1

 #****
 INPUT LISTING
 TEMPERATURE = 27.888 DEG C

 #*****
 .WIDTH OUT-88

 .OPTIONS ITLI=588 ITL5=8

 .MODEL NMOS NMOS (VT0=1V TOX=75NM U0=488 NSUB=2.5E16 LD=8.7UM)

 *LEVEL=1

 .MODEL NMOS NMOS (VT0=-1V TOX=75NM U0=288 NSUB=2.5E16 LD=8.7UM)

 *LEVEL=1

 .MODEL PMOS PMOS (VT0=-1V TOX=75NM U0=288 NSUB=3E15 LD=8.7UM)

 *LEVEL=1

 .VD0 1 # 0 C SV

 M1 1 3 2 Ø NMOS L=5.80 V=18.80

 M3 1 7 6 1 PMOS L=5.80 V=18.80

 M4 16 9 Ø 8 NMOS L=5.80 V=15.80

 M5 16 7 6 Ø NMOS L=5.80 V=15.80

 M6 19 7 6 Ø NMOS L=5.80 V=15.80

 M7 1 6 2 1 PMOS L=5.80 V=15.80

 M8 8 11 9 Ø NMOS L=5.80 V=15.80

 M11 1 1 9 1 PMOS L=5.80 V=15.80

 M13 1 4 12 1 PMOS L=5.80 V=18.80

 M13 1 4 12 1 PMOS L=5.80 V=18.80

 M13 1 4 12 1 PMOS L=5.80 V=18.80

 M14 15 11 Ø Ø NMOS L=5.80 V=18.80

 M15 15 14 12 Ø NMOS L=5.80 V=18.80

 M14 15 11 Ø Ø NMOS L=5.80 V=18.80

 M15 15 14 12 Ø PMOS L=5.80 V=18.80

 M16 1 11 1 2 PMOS L=5.80 V=18.80

 M17 Ø 7 14 Ø NMOS L=5.80 V=18.80

 M18 1 7 14 1 PMOS L=5.80 V=18.80

 M18 1 7 14 9 PMOS L=5.80 V=18.80

 M18 1 7 14 9 PMOS L=5.80 V=18.80

1*******12/Ø1/84 ******** SPICE 2G.1 (150CT8Ø) *******Ø4:55:19***** Ø CMOS/SOS BITSLICE SECTION 2 PRECHARGED AT ONE NODE

•_

٦.

C-62



Figure C-29. SPICE Plot Section 2 Input 010 One Node Precharge with Widths x2

******* .WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=1Ø.ØU M2 2 5 1 1 PMOS L=5.ØU W=2Ø.ØU M3 1 7 6 1 PMOS L=5.ØU W=1Ø.ØU M4 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M5 1Ø 7 6 Ø NMOS L=5.ØU W=15.ØU M5 10 7 6 Ø NMOS L=5.ØU W=15.ØU M5 1 \emptyset 7 6 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M6 1 9 6 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M7 1 6 2 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M8 \emptyset 11 9 \emptyset NMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M9 13 12 \emptyset \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M1 \emptyset 13 6 2 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M11 1 11 9 1 PMOS L=5. \emptyset U W=15. \emptyset U M12 1 12 2 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M13 1 14 12 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M14 15 11 \emptyset \emptyset NMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M15 15 14 12 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M16 1 11 12 1 PMOS L=5. \emptyset U W=15. \emptyset U M17 \emptyset 7 14 \emptyset NMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M18 1 7 14 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M18 1 7 14 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U C19 1 Ø Ø.1Ø74PF C2Ø 11 Ø Ø.279PF C21 Ø Ø Ø.951PF C21 2 C22 7 Ø Ø.183Fr C23 14 Ø Ø.100PF C24 12 Ø Ø.116PF C25 2 Ø Ø.234PF 7 9.125PF C27 9 J Ø.101PF C28 5 Ø Ø.1Ø3PF C29 3 Ø Ø.103PF C29 3 Ø Ø.103PF VIN1 11 Ø PULSE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (2V 5V 4NS ØNS ØNS 1ØNS) VP12AR 5 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAH Ø.5NS 4ØNS .PLOT TRAN V(2) (ØV,5V) .END

INPUT LISTING TEMPERATURE = 27.000 DEG C

CMOS/SOS BITSLICE SECTION 2 PRECHARGED AT ONE NODE



.

P

ŀ

Figure C-30. SPICE Plot Section 2 Input 101 One Node Precharge with Widths x2.

INPUT LISTING TEMPERATURE = 27.000 DEG C .WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 MODEL NMOS NMOS (VTO=1V TOX≖75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=15.ØU M2 2 5 1 1 PMOS L=5.ØU W=3Ø.ØU M3 1 7 6 1 PMOS L=5.ØU W=1Ø.ØU 10 9 8 8 MMOS L=5.80 W=18.80 10 9 8 8 MMOS L=5.80 W=15.80 10 7 6 8 MMOS L=5.80 W=15.80 1 9 6 1 PMOS L=5.80 W=18.80 1 6 2 1 PMOS L=5.80 W=18.80 M4 Μ5 M6 M7 C19 1 Ø Ø.1Ø74PF czø 11 Ø Ø.279PF C21 Ø Ø Ø.951PF C22 7 Ø Ø.183PF C23 14 Ø Ø.1ØØPF C24 12 Ø Ø.116PF 2 Ø Ø.234PF 6 Ø Ø.125PF C25 C26 C27 9 J Ø.1Ø1PF C28 5 Ø Ø.103PF C29 3 Ø Ø.103PF C29 3 0 0.103PF VIN1 11 0 PULSE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 0 PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VP1 3 0 PULSE (ØV 5V 4NS ØNS ØNS 1ØNS) VP1BAR 5 0 PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAH V(2) (ØV,5V) .END

CMOS/SOS BITSLICE SECTION 2 PRECHARGED AT ONE NODE

*****12/Ø1/84 *******

SPICE 2G.1 (150CT8Ø) *******Ø4:55:4Ø*****



Figure C-31. SPICE Plot Section 2 Input 010 One Node Precharge with Widths x3.

C~67

 1*******12/Ø1/84
 ******** SPICE 2G.1 (150CT8Ø)
 *******Ø4:56:ØØ*****

 Ø
 CMOS/SOS BITSLICE SECTION 2 PRECHARGED AT ONE NODE

 Ø****
 INPUT LISTING
 TEMPERATURE = 27.ØØØ DEG C

1	•••••12/#1	1/84 *****	••• sr	ICE 26.1	(1500	:T8Ø) *	******	84:56: <i>88</i> *		
	CHOS	SISOS BITSU	ICE SE	ECTION 2	PRECHA	RGED AT	ONE NO	DE XB		
***	** TR/	NSIENT ANA	LYSIS			TEMPERA	TURE .	27.585	DEG C	
	********	********		*******	*****		******	*******	*****	•
v										
^	TIME	V(2)								
x		.	d+ 88	1.25	#d+##	2.5	##d+##	3.75/	ld+88	5. <i>585</i> d+88
	. d+##	1.355d-#7		••••						
5	.898d-18 .8888d-89	1.355d-#7 1.355d-#7	:		:		:			:
1 Z	.588d-89 .888d-89	1.355d-07	-		:		:			:
23	.500d-09 .000d-29	1.355d-#7 1.355d-#7	:		:		:			:
3	.588d-89	1.3554-87	•		:		:			•
į	.5001-89	5.5720-01	•	•	•				•	
Š	.5081-89	2.568d+08	:		:			•		:
ĕ	.5881-89	3.548d+0#	:		:		:		•	:
;	.580d-89	3.4660+80 3.294d+80	:		:		:	:		:
8	. 999d-89 . 588d-89	3.328d+0# 3.498d+0#	:		:		:	· ·		:
9	.0084-89 .5884-89	3.697d+## 3.889d+##	:		:		:	•	•	:
i	.8084-88 .8584-88	4.875d+88 4.205d+88	÷		:		:			•
i	.1084-88	4.3724+08	:		:		-		•	
i	.2004-88	4.5784+00	:		:		•		•	•
i	.3004-08	4.723d+08	:		:		:		•	
1	.350d-88 .400d-08	4.776d+09 4.822d+09	:		:		:		:	•
1	.4500-08 .500J-00	4.854d+## 4.863d+##	:		:		:		•	•:
	.550d-00 .600-1-08	4.927d+08 4.925d+08	:		:		:			:
i	.6504-80	4.9480+88	•				•		•	
į	.750-1-98	4.9534+88	:		:		:			
i	.0500-08	4.6940+08	:		:		:			• •
i	. 258-1-80	4.8274+08	:		:		:		•	• :
2	.#8#d-#8 .#501-88	3.616d+08 3.168d+08	:		:			• -	•	:
2	.1001-00 .1501-00	2.743d+00 2.311d+00	:		:					:
2	.2001-08	1.860d+08 1.525d+88	:		: •	•	:		:	:
2	.390J-08	1.1964+08	•		•		•		•	•
ž	.408-1-88	7.8274-01	•	. •	:		:			:
2	.5084-08	3.8694-#1	÷ •	-	:		:			:
2	.550d-08 .600d-08	2.849d-01 2.928d-01			:		÷		•	:
2	.6503-08 .7#84-08	1.5£7d-#1 1.#79d-#1			:		÷		•	•
2	.750d-08	7.786d-82 5.548d-82	:		•		•		•	
ž	.0501-08	3.9660-02			:		÷			:
2	. 258.1-88	2.071d-02	•		:		:		•	:
3	.0501-00	1.427d-02 1.063d-02	-		:		:			:
3	.1003-00 .1503-00	7.533d-#3 5.341d-#3	:		:		:			:
3	.2993-08 .250d-08	3.892d-03 2.725d-03	:		:		÷		•	•
3	380-1-08	1.9754-03	:		•		•		•	:
3	4004-00	9.821d-84	:		:		:			:
3	.5000-08	5.141d-04	:		:		:		•	:
3	.550J-08 .670J-08	3.6430-84 2.6550-04	÷		:		:		•	:
3	.650d-00 .700d-00	1.859d-04 1.347d-04	:		:		:		•	:
3	.7583-88 .0884-88	9.671d-05	:		:					
3	.0504-08	4.9594-85	:		•		:		•	•
3	.9563-68	2.4950-05	:		:		:		•	:
4	.8889-00	1.766d-05			:		. :		:	:

Figure C-32. SPICE Plot Section 2 Input 101 One Node Precharge with Widths x3.







Figure C-34. Section 2 Two Node Precharge Node Plot.

· .

<u>Table C-6</u>

BITSLICE	PRECHARGED	SECTION	2	NODE	REFERENCE	LIST
GND Vdd NMOS PMOS 13 67 1 28 68 7 4 38 47 27 39 32 9 17	Ø 1 Ø 1 2 3 4 5 6 7 8 9 1 Ø 1 1 1 2 1 3 1 4 1 5					

C-71

. . .

CMOS/SOS PRECHARGED BITSLICE SECTION 2 TRANSIENT ANALYSIS INPUT LISTING TEMPERATURE = 27.888 DEG C .WIDTH OUT=8Ø .OPTIONS ITL1=5₽Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=4ØØ NSUB≠2.5E16 LD=Ø.7UM) +LEVEL ≠1 MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL≠1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=5.ØU M2 1 3 5 Ø NMOS L=5.ØU W=5.ØU 1 PMOS L=5.ØU W=10.ØU 1 PMOS L=5.ØU W=10.ØU M3 2 6 1 1 5 6 1 1 7 5 Μ4 1 7 5 1 PMOS L=5.00 W=10.00 10 9 0 0 NMOS L=5.00 W=15.00 10 7 5 0 NMOS L=5.00 W=15.00 1 9 5 1 PMOS L=5.00 W=10.00 1 9 5 1 PMOS L=5.00 W=10.00 M5 M6 Μ7 M8 1 9 5 1 PMOS L=5.00 W=10.00 M9 1 5 11 1 PMOS L=5.00 W=10.00 M10 0 12 9 0 NMOS L=5.00 W=15.00 M11 13 2 0 0 NMOS L=5.00 W=15.00 M12 13 5 11 0 NMOS L=5.00 W=15.00 M13 1 2 9 1 PMOS L=5.00 W=30.00 M14 1 2 11 1 PMOS L=5.00 W=10.00 M14 1 2 11 1 PMOS L=5.00 W=10.00 M15 1 14 2 1 PMOS L=5.00 W=10.00 M16 15 12 0 0 NMOS L=5.00 W=15.00 M17 15 14 2 0 NMOS L=5.00 W=15.00 M18 1 12 2 1 PMOS L=5.00 W=15.00 M19 0 7 14 0 NMOS L=5.00 W=15.00 M19 0 7 14 0 NMOS L=5.00 W=15.00 M20 1 7 14 1 PMOS L=5.00 W=30.00 C21 1 0 0.1135PF C22 12 0 0.279PF M8 C21 C22 C23 C24 C25 C26 C27 1 Ø Ø.1135PF 12 Ø Ø.279PF Ø Ø Ø.951PF 7 Ø Ø.183PF 14 Ø Ø.100PF 2 Ø Ø.288PF 11 Ø Ø.134PF 5 Ø Ø.260PF 9 Ø Ø.101PF 6 J Ø 103PF C28 C29 9 Ø Ø.1Ø1PF C3Ø 6 J Ø.1Ø3PF C31 3 Ø Ø.1Ø3PF VIN1 12 Ø PULSE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (ØV 5V 5NS ØHS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 6 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAH Ø.5NS 4ØNS .PLOT TRAN V(11) (ØV.5V) .END

*****12/Ø1/84 ******* SPICE 2G.1 (150CT8Ø) *******Ø4:21:59**

a

1******12/#1/84 ******* SPICE 2G.1 (150CT8#) *******#4:21:59***** CHOS/SOS PRECHARGED BITSLICE SECTION 2 TRANSIENT ANALYSIS XI TEMPERATURE . 27.888 DEG C TRANSIENT ANALYSIS ******************************* TIME V(11) 1.2584+88 . 2.5884+88 3.75#d+## 5.###d+## 4+88 $\begin{array}{c} \text{S.} \textit{SFW} \text{d} + \textit{SF} \\ \text{S.} \textit{SF} \text{d} + \textit{SF} \\ \text{S.} \textit{SF} \text{d} + \textit{SF} \\ \text{S.} \textit{SF} \text{d} + \textit{SF} \\ \text{S.} \textit{SF} \text{d} + \textit{SF} \\ \text{S.} \textit{SF} \text{d} + \textit{SF} \\ \text{S.} \textit{SF} \text{d} + \textit{SF} \\ \text{S.} \text{SF} \text{d} + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S.} \text{SF} \d + \text{SF} \\ \text{S} \SF} \d + \text{SF} \\ \text{S} \SF} \d + \text{SF} \\ \text{S} \SF} \d + \text{SF} \SF} \d + \text{SF} \SF} \d + \text{SF} \SF} \d + \text{SF} \SF} \d + \text{SF} \SF} \d + \text{SF}$

Figure C-35. SPICE Plot Section 2 Input 010 Two Node Precharge Using Basic Widths

1*******12/Ø1/84 ******* SPICE 2G.1 (150CT8Ø) ********Ø4:22:Ø9***** CMOS/SOS PRECHARGED BITSLICE SECTION 2 TRANSIENT ANALYSIS а (***** INPUT LISTING TEMPERATURE = 27.000 DEG C ************* .WIDTH OUT=3Ø .OPTIONS ITLI=5#0 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=5.ØU M2 1 3 5 Ø NMOS L=5.ØU W=5.ØU M3 2 6 1 1 PMOS L=5.ØU W=1Ø.ØU M3 2 6 1 1 PMOS L=5.0U W=10.0U M4 5 6 1 1 PMOS L=5.0U W=10.0U M5 1 7 5 1 PMOS L=5.0U W=10.0U M6 10 9 0 0 NMOS L=5.0U W=15.0U M7 10 7 5 0 NMOS L=5.0U W=15.0U M8 1 9 5 1 PMOS L=5.0U W=10.0U M10 0 12 9 0 NMOS L=5.0U W=15.0U M11 13 2 0 0 NMOS L=5.0U W=15.0U M12 13 5 11 0 NMOS L=5.0U W=15.0U M13 1 12 9 1 PMOS L=5.0U W=30.0U M14 1 2 11 1 PMOS L=5.0U W=10.0U M15 1 14 2 1 PMOS L=5.0U W=10.0U M16 15 12 0 0 NMOS L=5.0U W=15.0U C21 1 Ø Ø.1135PF 12 Ø Ø.279PF C22 C22 12 Ø Ø.279PF C23 Ø Ø Ø.951PF C24 7 Ø Ø.183PF C25 14 Ø Ø.100PF C26 2 Ø Ø.238PF C27 11 Ø Ø.134PF C28 5 Ø Ø.260PF C29 9 Ø Ø.101PF C30 6 Ø Ø.103PF C3Ø 6 Ø Ø.1Ø3PF C30 6 0 0.103PF C31 3 0 0.103PF VIN1 12 0 PULSE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 0 PULSE (5V 0V 5NS ØNS ØNS 1ØNS) VP1 3 0 PULSE (4V 5V 4NS ØNS ØNS 2NS) VP1BAR 6 0 PULSE (5V 0V 4NS ØNS ØNS 2NS) .TRAN 0.5NS 40NS .PLOT TRAN V(11) (0V,5V) END .END

· · .

. . .

.



1******12/#1/84 ******* SPICE 2G.1 (150CT6#) *******#4:22:#9*****

Node Precharge with Basic Widths.

C-75

•

-

•

.WIDTH OUT=8Ø .OPTIONS ITL1=5£0 ITL5=0 MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=1Ø.ØU M2 1 3 5 Ø NMOS L=5.ØU W=1Ø.ØU M3 2 6 1 1 PMOS L=5.ØU W=2Ø.ØU M4 5 6 1 1 PMOS L=5.ØU W=2Ø.ØU M5 1 7 5 1 PMOS L=5.ØU W=1Ø.ØU M6 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M7 1Ø 7 5 Ø NMOS L=5.ØU W=15.ØU M8 1 9 5 1 PMOS L=5.ØU W=18.ØU M1Ø Ø 12 9 Ø NMOS L=5.ØU W=15.ØU M11 13 2 Ø Ø NMOS L=5.ØU W=15.ØU M12 13 5 11 Ø NMOS L=5.ØU W=15.ØU M13 1 12 9 1 PMOS L=5.ØU W=15.ØU M14 1 2 11 1 PMOS L=5.ØU W=18.ØU M15 1 14 2 1 PMOS L=5.ØU W=18.ØU M16 15 12 Ø Ø NMOS L=5.ØU W=15.ØU M14 1 2 11 2 PMOS L=5.ØU W=18.ØU M15 1 14 2 1 PMOS L=5.ØU W=15.ØU M16 15 12 Ø Ø NMOS L=5.ØU W=15.ØU M17 15 14 2 Ø NMOS L=5.ØU W=15.ØU M18 1 12 2 1 PMOS L=5.ØU W=15.ØU M18 1 12 2 1 PMOS L=5.ØU W=15.ØU M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU VDD 1 Ø DC 5V M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M2Ø 1 7 14 1 PMOS L=5.ØU W=3Ø.ØU C21 1 Ø Ø.1135PF 12 Ø Ø.279PF C22 C23 Ø Ø Ø.951PF C24 7 Ø Ø.183PF C25 14 Ø Ø.180PF C26 2 Ø Ø.288PF C27 11 Ø Ø.134PF C28 5 Ø Ø.26ØPF C29 9 Ø Ø.1Ø1PF C3Ø 6 Ø Ø.1Ø3PF C31 3 Ø Ø.1Ø3PF VIN1 12 Ø PULSE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 6 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(11) (ØV.5V) .END



Figure C-37. SPICE Plot Section 2 Input 010 Two Node Precharge with Widths x2.

C-77

. . .

<u>.</u>

 1*******12/Ø1/84
 ********Ø4:23:51*****

 Ø
 CMOS/SOS PRECHARGED BITSLICE SECTION 2 TRANSIENT ANALYSIS

 Ø*****
 INPUT LISTING

 TEMPERATURE =
 27.000 DEG C

. WIDTH OUT=8Ø . OPTIONS ITL1=5ØØ ITL5=Ø . MODEL NMOS NHOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +LEVEL=1 . MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=2ØØ NSUB=3E15 LD=Ø.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=1Ø.ØU M2 1 3 5 Ø NMOS L=5.ØU W=1Ø.ØU M4 2 6 1 1 PMOS L=5.ØU W=2Ø.ØU M4 5 6 1 1 PMOS L=5.ØU W=2Ø.ØU M5 1 7 5 1 PMOS L=5.ØU W=2Ø.ØU M6 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M7 1Ø 7 5 Ø NMOS L=5.ØU W=15.ØU M9 1 5 11 1 PMOS L=5.ØU W=15.ØU M1Ø Ø 12 9 Ø NMOS L=5.ØU W=15.ØU M1Ø Ø 12 9 Ø NMOS L=5.ØU W=15.ØU M10 Ø 12 9 Ø NMOS L=5.ØU W=15.ØU M11 Ø 12 9 Ø NMOS L=5.ØU W=15.ØU M12 13 5 11 Ø NMOS L=5.ØU W=15.ØU M13 1 12 9 1 PMOS L=5.ØU W=15.ØU M14 1 2 11 1 PMOS L=5.ØU W=15.ØU M15 1 14 2 1 PMOS L=5.ØU W=15.ØU M16 15 12 Ø Ø NMOS L=5.ØU W=15.ØU M17 15 14 2 7 PMOS L=5.ØU W=15.ØU M18 1 12 2 1 PMOS L=5.ØU W=15.ØU M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M17 15 14 2 Ø N NMOS L=5.ØU W=15.ØU M17 15 14 2 Ø N NMOS L=5.ØU W=15.ØU M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M19 Ø 7 14 Ø NMOS L=5.ØU W=15.ØU M20 1 7 14 Ø NMOS L=5.ØU W=15.ØU M20 1 7 14 Ø NMOS L=5.ØU W=15.ØU M20 1 7 14 Ø NMOS L=5.ØU W=10 M20 1 7 14 Ø NMOS L=5.ØU W=10 M20 1 7 14 Ø NUSE (5V ØV SNS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (5V ØV SNS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (5V ØV SNS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (5V ØV SNS ØNS ØNS 2NS) . PLOT TRAN V(11) (ØV,5V) . END

	CHU	SISOS PRECI	WARGED B	ITSLICE SECT	IUN Z IRANS	3541	ANALYSIS AZ	
	TR	ANSIENT ANA	ALYSIS		TEMPERATUR	£ -	27. 888 DEG (:
	*******	*********		***********	*********	****	*********	•
	TIME	V(11)						
		∎.	_d+##	1.2584+88	2.5884	+##	3.75#d+##	5.###d+#
Į.	0+88	9.8354-88	:	•	•		•	•
1.6	001-09	9.0354-08	•	:	:			:
	5 <i>88</i> 4- 89 1884-89	9.035d-08 9.035d-08	•	:	:		:	:
. 5	588d-89	9.#354-#8	*	•	•		•	•
. ŝ	884-89	9.0350-08	•	:	:		:	:
	588J-89	9.#35d-#8		:	:		:	:
	1884-89 1884-89	9.#35d-#8 9.#35d-#8	:	:	:		•	
	60-b00	9.8470-88	:		:		•	
	884-59	9.3264-00	•	•	:		•	:
/.5).8	500d~09 500d~09	4.21#d-#5 4.55#d-#3		:	:		:	:
1.5	88d~89	2.254d-82 5.1684-82		•	•		•	
	88d-89	1.2484-81		:	:		•	:
	1984-88 1584-80	3.7794-#1	: -	:	:		:	:
.1	800-88 580-88	6.2#9d-#1 9.148d-#1	: '	•	:		•	
. 2	804-98	1.3254+08	•		. :			
. 1	3894-88	2.263d+00	:	:	• :	_	:	:
. 4	35 <i>8</i> 3~88 1883~88	2.798d+88 3.263d+88	:	:	:	•	• :	:
	503-88	3.686d+08	•	•	•		•	•
	08-1022	4.3#34+08	:	:	:			• . :
	500d-08	4.5230+08 4.648d+08	:	:	:		:	· • :
• 7	7 <i>09</i> d- <i>0</i> 0 7581-08	4.7493+08	·	:	:		•	•
	88-L98	4.8654+66	:	•	•			
	88-1906	4.8474+08	:	:	:		:	•
	9504-08 7004-08	4.7550+08	:	•	:		:	• •
	158d-88	4.2794+68	•	•	:		•	•
. 1	501-00	3.451d+08	:	:	:		•	:
1	250d-00	2.3640+00	:	:	• :	-	•	:
	3001-00 2501-00	1.8154+68	:	:•	• :		:	:
	80-1-88	8.926d-01	•	•	•		•	
	5884-88	3.4#8d-#1	•	:	:		:	:
2.4	5 <i>8</i> d- <i>8</i> 8 688d-88	2.184d-#1 1.243d-#1	: .	:	:		:	:
ş. 6	658d-88	6.9784-82	:	•	:		:	:
	7501-08	2.2454-02	•	:				
2.0	890d-88 850d-80	1.1784-02 7.237d-03	:	:	:		:	•
	780-J-08	3.8140-83	•	•	:		:	:
1.	600.1-68	1.1834-03	•	:				
3. I	050J-08 100J-08	5.964d-04 3.459d-04		:	:		:	:
3.	150J-00 200J-00	1.872d-84 9.195d-85	:	:	:		:	:
	2504-08	5.594d-05	:		•			
3.	3504-08	1.5620-05	•		:		:	
3.	400-J-08 450-J-00	8.894d-06 4.496d-06		:	:		:	
ş.,	5401-00	2.6244-06	:	•	•		•	•
3.0	599-1-110 6 8 9-1-80	7.5484-87	-	:	:		:	
3.	650J-88	4.9354-07	•	•	•		:	:
j:	750.1-08	2.0184-07		:	:		•	
3.1	5999-00 5999-00	1.5240-07	•	:	:		:	
3.	300 1-00	1.0850-07		•	•		÷	
3 - '	889-1-88	9.5120-08	•	:				

í

÷

Ĩ

Ľ

2

Figure C-38. SPICE Plot Section 2 Input 101 Two Node Precharge with Widths x2.

•

.WIDTH OUT=8Ø .OPTIONS ITL1=5#Ø ITL5=Ø MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +1 EVEL = 1.MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) .MODEL PMOS PMOS (VTO=-1V TOX=75NN +LEVEL=1 VDD 1 \emptyset DC 5V M1 1 3 2 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M2 1 3 5 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M3 2 6 1 1 PMOS L=5. \emptyset U W=3 ϑ . ϑ U M4 5 6 1 1 PMOS L=5. \emptyset U W=3 ϑ . ϑ U M5 1 7 5 1 PMOS L=5. \emptyset U W=1 ϑ . ϑ U M6 1 ϑ 9 ϑ ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M6 1 ϑ 9 ϑ ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M6 1 ϑ 9 ϑ ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M6 1 ϑ 9 ϑ ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M6 1 ϑ 9 ϑ ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M6 1 ϑ 9 ϑ ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M1 1 ϑ 7 5 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M1 1 2 5 11 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M12 13 5 11 ϑ NMOS L=5. ϑ U W=15. ϑ U M13 1 12 9 1 PMOS L=5. ϑ U W=15. ϑ U M14 1 2 11 PMOS L=5. ϑ U W=1 ϑ . ϑ U M15 1 14 2 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M16 15 12 ϑ ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M16 15 12 ϑ ϑ NMOS L=5. ϑ U W=15. ϑ U M17 1 ϑ 14 2 1 PMOS L=5. ϑ U W=15. ϑ U M18 1 12 2 1 PMOS L=5. ϑ U W=15. ϑ U M19 ϑ 7 14 ϑ NMOS L=5. ϑ U W=15. ϑ U M2 ϑ 1 7 14 1 PMOS L=5. ϑ U W=15. ϑ U M2 ϑ 1 7 14 1 PMOS L=5. ϑ U W=15. ϑ U M2 ϑ 1 7 14 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M2 ϑ 1 7 14 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M2 ϑ 1 7 14 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M2 ϑ 1 7 14 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M2 ϑ 1 7 14 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U +LEVEL=1 C21 1 Ø Ø.1135PF C22 12 Ø Ø.279PF C23 Ø Ø Ø.951PF C24 7 Ø Ø.133PF C25 14 Ø Ø.1ØØPF C26 2 Ø Ø.288PF C27 11 Ø Ø.134PF C27 11 Ø Ø.134PF C28 5 Ø Ø.250PF C29 9 Ø Ø.1Ø1PF C3Ø 6 Ø Ø.1Ø3PF C31 3 Ø Ø.1Ø3PF VIN1 12 Ø PULSE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VPIBAR 6 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(11) (ØV,5V) .END

. .

*** TR/	ANSIENT ANA	LYSIS		т	EMPERATU	RE -	27.88	DEG C	:
********		*****	*******			*****			•
TIME	VIII								
	# .		1.25		2.588	d+88	3.7	5 9 d+99	5.888d+
Ø. d+88 S 8884-18	5.888d+88	•		•	:			:	
1.8084-89	5.000d+09			•				•	
1.5094-89	5.82#d+8#	:		:	:			:	
2.588d-89	5.000d+00 5.000d+00	:		:	:			:	
3.5004-89	5.888d+88	•		•	•			•	
4.5881-89	5.0004+00	:							
5.000d-09 5.500d-09	4.9693+88 4.8643+88	:		:	:			:	•
6.0001-09	4.52#d+8#	•		•	•			÷.	•
7.8884-89	3.261d+0#	:		•			•	:	
7.5880-89 8.8880-89	2.5440+00 1.87#d+08	:		:	• .			:	
8.5091- 99 9.8601-89	1.3#6d+## 8.474d-#1	:	•		•			:	
9.5084-89	5.4120-01	•	•	•				•	
1.0000-08 1.0501-08	1.753d-#1	1. • T		:	:			:	
1.1900-08	1.864d-01 5.895d-82	:		:				:	
1.2004-08	3.3224-02	-		•				:	
1.300-1-08	1.061d-02	-		:				:	
1.358-1-00	5.674d-03 2.987d-03	-		•				:	
1.450.3-08	1.728d-83	:						•	
1.5504-00	4.576d-04	•		:				:	
1.6004-00	2.593d-04 1.466d-04	-		:				:	
1.7504-08	8.7450-05	:		•				•	
1.0004-08	4.796d-04	*		:				:	
1.0501-00	1.010d-02 3.373d-02	-		:				:	
1.9504-00	7.316d-02	•••		•				•	
2.0504-08	2.766d-81	. •		:				:	
2.1590-08	7.31#d-01	:	•	:	•			:	
2.2004-08	1.4944+08	:	*	: •				:	
2.309.1-08	1.9644+90				•				
2.400-00	2.969d+08	:		:		•		:	
2.4500-00 2.5000-00	3,425d+00 3,824d+00	:		:			•		
2.5504-08	4.1384+88	•		:				•	
2.6500-08	4.5684+08	:		:				:	•
2.7001-00 2.7501-08	4.688d+08 4.786d+08	:		:				:	•.
2.8091-98	4.8550+00	•		•				:	•
2.300-00	4.933d+00	:		:				:	
2.3504-08	4.953d+88 4.968d+88	:		:				:	•
3.0501-00	4.9794+08			•					
3.150-0-00	4.398d+88	:		:				:	
3,2004-00	4.9944+09	:		:				:	
3.0000-00	4.9974+10	•		•		•		•	
3.1004-00	4.9994+44	:		:	•			:	
3.500d-00	4.999d+00 4.999d+00	:		:				:	
3.1504-00	5.080d+00	:		•				•	
3.65#d-#8	5.8884+88	:		:				:	
3.7093-08 3.7503-08	5.868d+08	:		:				:	
3.0001-00	5.008d+08	·		•				•	
3,700.1-08	5.868d+UR	:		:		•		:	
3,7504-88	5.0000+00	•						•	

ł

ĺ

1

1

1

)

.

•

•

Figure C-39. SPICE Plot Section 2 Input 010 Two Node Precharge with Widths x3.

C-81

.WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 .MODEL NMOS NHOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL =1 .MODEL PMOS PMOS (VTO=-IV TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=15.ØU M2 1 3 5 Ø NMOS L=5.ØU W=15.ØU M2 1 3 5 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M3 2 6 1 1 PMOS L=5. \emptyset U W=3 \emptyset . \emptyset U M4 5 6 1 1 PMOS L=5. \emptyset U W=3 \emptyset . \emptyset U M5 1 7 5 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M6 1 \emptyset 9 \emptyset \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M7 1 \emptyset 7 5 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M8 1 9 5 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M9 1 5 11 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M1 \emptyset \emptyset 12 9 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M12 13 5 11 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M12 13 5 11 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M12 13 5 11 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M13 1 12 9 1 PMOS L=5. \emptyset U W=3 \emptyset . \emptyset U M14 1 2 11 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M15 1 14 2 1 PMOS L=5.ØU W=1Ø.ØU M16 15 12 \emptyset Ø NMOS L=5.00 W=10.00 M16 15 12 \emptyset Ø NMOS L=5.00 W=15.00 M17 15 14 2 Ø NMOS L=5.00 W=15.00 M18 1 12 2 1 PMOS L=5.00 W=10.00 M19 Ø 7 14 Ø NMOS L=5.00 W=15.00 M20 1 7 14 1 PMOS L=5.00 W=30.00 C21 1 Ø Ø.1135PF C22 12 Ø Ø.279PF C23 Ø Ø Ø.951PF C24 7 Ø Ø.183PF C25 14 Ø Ø.1ØØPF C25 2 Ø Ø.288PF C27 11 Ø Ø.134PF C28 5 Ø Ø.26ØPF C29 9 J Ø.1Ø1PF C3Ø 6 Ø Ø.1Ø3PF C31 3 Ø Ø.1Ø3PF VIN1 12 Ø PULSE (5V 5V 5NS ØNS ØNS 1ØNS) VIN2 7 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 6 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(11) (ØV,5V) .END

.

۰.

<u>'</u>

٠.

7

. .

٠.

÷

C-82

· · ·

1******12/81/84 *******	SPICE 26.1 (1)	50CT8#	***#4:25:34****	
CMOS/SOS PRECHARG	ED BITSLICE SET	TEMPERATURE	ENT ANALYSIS AS	~
8+++++++++++++++++++++++++++++++++++++	,	*********	- 27.000 UEG	••
X TIME V(11)				
X 8. d+	## 1.25#d+	2.586d.	-## 3.75#d+##	5.888d+88
#. d+## 9.#35d-#8			•••••••	
1.808-1-89 9.805d-88 *	:	:	:	:
2.0001-09 9.0351-08	:	:	:	:
3.868d-89 9.835d-08 *	:	:	:	:
4.8884-89 9.8054-88 *	:	:	:	:
5.000j-09 9.005d-00 *	:	-	:	
6.8884-89 9.8474-88	:		•	
7.888d-89 9.299d-88 *	:		:	
8.888d-89 3.762d-83 *			:	
9,8084-89 4.6784-82 .*			:	
1.888d-88 2.186d-81 . *	•			-
1.180d-88 5.913d-81 . 1.158d-88 8.761d-81 .				
1.288d-88 1.278d+88 . 1.258d-88 1.727d+88 .	•	•		
1.388d-88 2.218d+88 . 1.358d-88 2.736d+88 .		•	•	
1.488d-88 3.222d+88 . 1.458d-88 3.625d+88 .			•	-
1,508J-00 3.985d+08 . 1,550d-08 4.274d+08 .			•	•
1.6803-00 4.4624+00 . 1.6503-00 4.6344+00 .		:	:	• • :
1.708d-80 4.739d+08 . 1.750J-80 4.811d+88 .	•	:	:	
1.000J-00 4.857d+00 . 1.050J-00 4.879d+00 .	•	:	•	•:
1.950J-08 4.849d+00 . 1.950J-08 4.709d+00 .	:	:	:	
2.0000-08 4.568d+08 . 2.058d-08 4.284d+00 .	:	•	:	• •
2.1001-00 3.9774+00 . 2.1504-00 3.4724+00 .	•			:
2.2504-00 2.3584+00 . 2.2504-00 2.3584+00 .	:	•:	•	:
2.350d-08 1.312d+08 .		• :	•	:
2.450d-00 8.799d-01 . 2.450d-08 5.763d-01 .	. • •	:	•	:
2.558d-88 3.613d-81 . * 2.558d-88 2.819d-81 . *		:	•	:
2.6501-00 1.2/20-01 .*	:	:	•	:
2.7580-88 2.2420-82	•	•	:	:
2.850d-08 6.788d-03		:	:	:
2.0504-80 2.1534-03	:	•	:	:
3.0504-08 1.0650-03 -	:	:	:	:
3.15#d-#8 1.814d-#4 *	:	:	:	:
3.2500-00 1.2270-05 *	:	:	:	:
3.3583-88 2.9783-85 - 3.3583-88 1.625d-05 *	:	•		:
3.4503+08 4.0790-06 *	:		:	•
3.5504-00 1.411d-06 *	:	•	:	:
3.6501-00 4.6640-07 *	:	:	•	•
3.7504-08 2.4634-07 *	:	•	:	•
3.658d-08 1.246d-07 * 3.380d-00 1 #81d-07 *	:	•	:	•
3.9584-80 9.9614-88 *	•	•	•	•
•••				

Ĩ

Ľ

Figure C-40. SPICE Plot Section 2 Input 101 Two Node Precharge with Widths x3.



. .

Figure C-41. Basic Section 3 CLL Plot.



Figure C-42. Basic Section 3 Node Plot.

<u>Table</u> <u>C-7</u>

BITSLICE SECTION 3 NODE LIST

GND	ø
Vdd	1
NMOS	ø
PMOS	1
11	2
51	3
1	4
3.Ø	5
2	Ğ
42	7
54	8
5	ğ
46	โต
29	îĩ
13	12
35	12
14	14
2	14
3	15
19	16

ŀ

 1*******12/Ø1/84
 *******Ø1:15:15*****

 Ø
 CMOS/SOS BITSLICE SECTION 3 BASIC TRANSIENT ANALYSIS

 Ø****
 INPUT LISTING

 TEMPERATURE =
 27.ØØØ DEG C

L

. . . .

.WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS INOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 0 DC 5V M1 1 3 2 1 PMOS L=5.00 W=10.00 M3 3 7 0 0 NMOS L=5.00 W=10.00 M4 8 3 2 0 NMOS L=5.00 W=10.00 M5 1 0 9 0 0 NMOS L=5.00 W=15.00 M5 1 0 9 0 0 NMOS L=5.00 W=15.00 M7 10 2 5 0 NMOS L=5.00 W=15.00 M7 10 2 5 0 NMOS L=5.00 W=15.00 M1 1 1 2 11 PMOS L=5.00 W=15.00 M1 1 1 5 11 1 PMOS L=5.00 W=15.00 M1 1 1 5 11 1 PMOS L=5.00 W=15.00 M1 1 1 5 11 1 PMOS L=5.00 W=15.00 M1 2 1 2 1 0 NMOS L=5.00 W=15.00 M1 3 1 2 12 1 0 NMOS L=5.00 W=15.00 M1 4 0 2 1.0 NMOS L=5.00 W=15.00 M1 5 16 15 0 0 NMOS L=5.00 W=15.00 M1 6 14 12 0 NMOS L=5.00 W=15.00 M1 7 12 14 1 PMOS L=5.00 W=15.00 M1 6 15 9 1 PMOS L=5.00 W=15.00 M1 7 12 14 1 PMOS L=5.00 W=15.00 M1 7 12 14 1 PMOS L=5.00 W=15.00 M1 7 12 14 1 PMOS L=5.00 W=15.00 M1 7 12 14 1 PMOS L=5.00 W=15.00 M1 4 0 5 19 0 NMOS L=5.00 W=15.00 M1 5 16 15 0 1 PMOS L=5.00 W=15.00 M2 7 1 0 1 2 1 PMOS L=5.00 W=15.00 M1 7 1 2 14 1 PMOS L=5.00 W=15.00 M1 9 0 5 9 0 N MOS L=5.00 W=15.00 M1 9 0 5 9 0 N MOS L=5.00 W=15.00 M1 9 0 5 9 0 N MOS L=5.00 W=15.00 M2 0 1 1 0 PMOS L=5.00 W=15.00 M2 0 0 0.1417F C2

	CHO	S/SOS BITSL	ICE SE	CTION 3 B	ASIC	TRANSIEN	T ANAL	4212		
B****	TR/	ANSIENT ANA	LVSIS			TEMPERAT	URE =	27. 888	DEG C	
ge======		**********	******	********			*****	*******	*****	•
~										
x										
<u>"</u> тін	IE	V(11)								
¥		s .	d+##	1.250	d+88	2.54	#d+##	3.75	8d+55	5.8884+88
5.888	1+ <i>81</i> 1-1 <i>8</i>	5.888d+88 5.888d+88	:	:			:		:	
1.000	1-89	5.###d+##		•			•		•	:
1.5884	1-89 1-89	5.888d+88	•	:			:		:	-
2.500	1-89	5.Ø@#d+88	:	:					•	•
3.808	1-#9	5. 898 4+88	·	•			•		:	:
4.868	1-89	5. <i>888</i> d+88	:	:			:		:	•
4.500	1-89	5.808d+88	•	•			•		•	:
5.508	1-09	5.888d+88	:	:			:		:	•
6.888	1-89	5.888d+88	•	•			•		·	
7.868	1-89	5. <i>988</i> d+88	:	:			:		:	•
7.5000	1-89	4.998d+88	•	•			•		•	.*
0.508	1-87	4.741d+UB	:	:			:		:	•
9.808	1-89	4.418d+88	•	•			•		•	• .
9.500 1.000	1-89 1-88	4.0594+00 3.4514+00	:	•			:		: -	:
1.858	9-08	2.85#d+##	:	•			•		•	
1.150	1-00 1-00	2.174d+00 1 541d+00	:	:		•	:		:	•
1.200	4-98	1.8410+88	:	• :			:			:
1.250	1-#8	6.65#d-#1	•	• .			•		•	•
1.250	1-08	2.342d-01	•	:			:		:	:
1.400	0-80	1.2454-81	-t	•			•		•	•
1.500	1-00 1-00	3.76#d-02		:			:		:	:
1.550	1-98	2.0754-02	:	•			•		•	•
1.680	d-08 d-08	6.714d-02		:			:		:	:
1.709	0-08	3.939d-#3					•		•	•
1.750	d-08 d-68	1.1740-02		:			:		:	:
1.058	J-80	1.5#2d-#1	•				•		•	
1.900	-1-98 	4.0030-01		•			:		:	:
2.800	d-00	1.245d+0#	:	é		•			•	
2.950	d≁Ø8 ⊲∞ø8	1.814d+00 2.420d+00	•			•			:	:
2.158	1-90	2.99#d+##						•	•	
2.200	d-88	3.4980+88	•	-			·	•		•
2.300	d-08	4.213d+0#	:	:			:		:	•
2.350	4-00	4.4444+88	•	•			•		•	•••
2 450	4-88	4.737d+08	:				:		:	• `
2.500	d-88	4.815d+##	•				•		•	• •
2.600	J-90	4.918d+0#	:	:			:		:	•
2.650	1-90 1-90	4.942d+88	·	•			·		•	•:
2.750	d-99	4.9744+08	:	:			:		:	÷
2.800	1-08	4.982d+88	•	•			•		•	:
2.000	u∽#8 ∐-910	4.9924+88	:	•			:		:	
2.050	ũ-00	4.995d+08	•				•		•	:
3.000	1-68	4.9960+08	:				:		:	
3.109	J-00	4.998d+##					:			•
3.15#	1-08 1-90	4.9994+0#	•	•			•		:	:
3.250	1-98	5.0204+08	•	:						•
3.300	d-98	5.000d+08	•	•			•		•	:
3.400	1-00	5.868d+08	:				:		:	•
3.458	d-#8	5.000d+00	•	•			•		•	:
3.550	1-86	5.000400	:	:			:		:	-
3.648	1-98	5.888d+88	•	-			•		•	•
3.700	.j~00	5.888d+88					:		:	
3.758	J-80	5.0404+00								•
3.000	-1-80 -1-80	3.070d+08 5.0786d+08					:		:	
3.200	1-00	5.008d+08							•	•
3.350	.j-00	5.0F8d+08 5.0F8d+A8	:	•			:		:	*
			÷				_ <u> </u>		÷	

1*******12/81/84 ******** SPICE 26.1 (150CT88) *********************************

Figure C-43. SPICE Plot Basic Section 3 Input 010.

C-87

1.10

INPUT LISTING TEMPERATURE = 27.ØØØ DEG C ***** *********************** .WIDTH OUT=80 .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75N) +LEVEL=1 VDD 1 & DC 5V M1 1 3 2 1 PMOS L=5.ØU W=1Ø.ØU M2 1 2 5 1 PMOS L=5.ØU W=1Ø.ØU M3 8 7 Ø Ø NMOS L=5.ØU W=15.ØU M4 8 3 2 Ø NMOS L=5.ØU W=15.ØU M5 1 7 2 1 PMOS L=5.ØU W=15.ØU M6 1Ø 9 Ø Ø NMOS L=5.ØU W=15.ØU M7 1Ø 2 5 Ø NMOS L=5.ØU W=15.ØU M3 1 9 5 1 PMOS L=5.ØU W=15.ØU M1Ø 13 12 Ø Ø NMOS L=5.ØU W=16.ØU .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) M12 1 12 11 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M13 1 14 12 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M14 ϑ 2 14 ϑ NMOS L=5. ϑ U W=15. ϑ U M15 16 15 ϑ ϑ NMOS L=5. ϑ U W=15. ϑ U M16 16 14 12 ϑ NMOS L=5. ϑ U W=15. ϑ U M17 1 2 14 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M13 1 15 12 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M19 ϑ 15 9 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M2 ϑ 1 15 9 1 PMOS L=5. ϑ U W=15. ϑ U M2 ϑ 1 15 9 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U C21 1 ϑ ϑ .6 ϑ IPF Č22 Ø J Ø.465PF 15 g g.148PF C23 C24 9 0 0.141PF C25 2 0 0.230PF C26 12 0 0.130PF C27 14 J J.134PF C28 11 J Ø.134PF C28 11 0 0.134PF C29 5 Ø Ø.137PF VIN1 15 Ø PULCE (5V ØV 5NS ØNS ØNS 1ØNS) VCAR 3 Ø PULCE (ØV ØV ØNS ØNS ØNS 1ØNS) VSEL 7 Ø PULCE (5V 5V ØNS ØNS ØNS 1ØNS) .TEAN Ø.5NS 4ØNS .PLOT TRAN V(11) (ØV.5V) END .END

*****12/Ø1/34 ******** SPICE 2G.1 (150CT8Ø) ******Ø1:19:51***** CMOS/SOS BITSLICE SECTION 3 BASIC TRANSIENT ANALYSIS

• **.** -____

1*******12/#1/84 ******* SPICE 2G.1 (150CT8#) *******#1:19:51***** . CHOS/SOS BITSLICE SECTION 3 BASIC TRANSIENT ANALYSIS TRANSIENT ANALYSIS TEMPERATURE = 27.888 DEG C g * g. TIME V(11)

 #.
 d+##

 9.#35d-##
 9.#35d-##

 1.357d+##
 9.#35d-##

 1.357d+##
 9.#35d-##

 1.237d+###
 1.

 3.95#d+##
 1.

 3.95#d+##
 1.

 3.572d+###
 1.

 3.572d+###
 1.

 3.572d+###
 1.

 3.572d+###
 1.

 3.572d+##
 1.

 3.572d+##
 1.

 3.572d+###
 1.

 3.572d+###
 1.

 3.572d+###
 1.

 3.572d+###
 1.

 3.572d+###
 1.

 4.372d+###
 1.

 3.75#d+## 5.###d+## х 1.258d+88 2.5**88**d+88 : . 4 • . .

ىد مىك يىل بىك بىك <u>مەركىك مەركىك مەركىك مەركىك بەركىك بەركىك بەركىك بەركىك بەركىك بەركىك بەركىك بەركىك بەركىك</u>

1.

Figure C-44. SPICE Plot Basic Section 3 Input 101.

- -

C-89

- <u>-</u> -

. ÷ - -

• • • . •••• ____



••••

4

Figure C-45. CLL Plot Section 3 One Node Precharge.



Figure C-46. Section 3 One Node Precharge Node Plot.

<u>Table</u> <u>C-8</u>

•

· · ·

·. .

BITSLICE	SECTION	3	NODE	REFERENMCE	LIST	-	ONE	NODE	PRECHARGE
GND VNMOS PMOS 33 85 1 81 3 66 42 5 5 7 8 1 7 8 6 1 7 7 8 8 29 8 8 22	Ø 1 Ø 1 2 3 4 5 6 7 8 9 1 Ø 1 1 2 3 4 5 6 7 8 9 1 Ø 1 2 3 4 5 6 7 8 9 1 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 8 9 1 2 3 4 5 6 7 7 8 9 1 2 3 4 5 6 7 7 8 9 1 2 3 4 5 6 7 7 8 9 1 2 3 4 5 6 7 7 8 9 1 2 3 4 5 6 7 7 8 9 1 2 3 4 5 6 7 7 8 9 1 2 3 4 5 6 7 7 8 9 1 2 3 4 5 6 7 7 8 9 1 2 3 4 5 6 7 7 8 9 1 1 2 3 4 5 6 7 7 8 9 1 1 2 3 4 5 6 7 7 8 9 1 1 2 3 4 5 6 7 7 8 9 1 1 2 3 4 5 7 7 8 9 1 1 2 3 1 1 2 3 1 2 3 1 1 2 3 1 2 3 1 1 2 3 1 1 2 3 1 1 2 3 1 1 2 1 1 1 2 1 1 1 1								



CMOS/SOS BITSLICE SECTION 3 PRECHARGED AT ONE NODE INPUT LISTING TEMPERATURE = 27.ØØØ DEG .WIDTH OUT=8Ø .OPTIONS ITLI=5¢Ø ITL5=Ø .Model NMos NHOS (VTO=1V TOX=75NM UO=4ØØ NSUB=2.5E16 LD=Ø.7UM) +LEVEL= .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=5.ØU 2 5 1 1 PHOS L=5.ØU W=10.ØU M2 1 7 6 1 PMOS L=5.ØU W=1Ø.ØU 1 9 8 1 PMOS L=5.ØU W=1Ø.ØU 12 11 Ø Ø NMOS L=5.ØU W=15.ØU 12 7 6 Ø NMOS L=5.ØU W=15.ØU 1 11 6 1 PMOS L=5.ØU W=10.ØU M3 Μ4 Μ5 M6 M7 M24 1 9 19 1 PHOS L=5.00 W=30.00 1 J D.1239PF C25 C26 6 J Ø.316PF Č27 9 J J.232PF C28 Ø U Ø. 940PF C29 19 Ø Ø.111PF C3Ø 15 Ø Ø.12ØPF C31 13 Ø Ø.1Ø3PF 2 0 0.234PF C32 C33 17 J J. 1PF C33 17 σ σ .1PF C32 8 σ σ .119PF VIN1 9 σ PULCE (σ V SV SNS σ NS σ NS 1 σ NS) VP1 3 σ PULCE (σ V SV 4NS σ NS σ NS 2NS) VP15AD 5 σ PULCE (σ V σ V 4NS σ NS σ NS 2NS) VCAP 10 σ PULCE (σ V σ V σ NS σ NS 1 σ NS) VCAP 11 σ PULCE (σ V σ V σ NS σ NS 1 σ NS) VCAP 11 σ PULCE (σ V σ V σ NS σ NS 1 σ NS) VCAP 11 σ PULCE (σ V σ V σ NS σ NS 1 σ NS) VCAP 11 σ PULCE (σ V σ V σ NS σ NS 2 σ S) VFM12 10 σ PULCE (σ V σ V σ NS σ NS σ NS 2 σ S) TPAN J.5H3 4 σ NS .PLOT TRAH V(2) (σ V.5V) .EHD . EilD

******12/01/04 ******** SPICE 2G.1 (150CT80) *******19:46:30***

C-92

÷.,

· · · :

· · ·



•

•

•.

. ∎3.≂≈

•

•

ļ

1



.

INPUT LISTING .WIDTH OUT=8Ø OPTIONS ITL1=540 ITL5=0 .MODEL NMOS NHOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PHOS (VTO=-1V TOX=75NM UO≠200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=5.ØU M2 2 5 1 1 PHOS L=5.ØU W=1Ø.ØU M3 1 7 6 1 PMOS L=5.ØU W=1Ø.ØU M4 1 9 8 1 PMOS L=5.ØU W=1Ø.ØU C25 1 J Ø.1230PF .END

******12/01/84 ******** SPICE 2G.1 (150CT80) *******19:46:01****

....

1.

وموجعه والمراجع والمعاد والمراجع المراجع والمراجع والمراجع المراجع المراجع المراجع المراجع المراجع المراجع الم

CMOS/SOS BITSLICE SECTION 3 PRECHARGED AT ONE NODE TEMPERATURE = 27.000 DEG C

• • • • • • •

C-94


•

.

-

Ľ

P

Figure C-48. SPICE Plot Section 3 Input 101 One Node Precharge with Basic Widths

 1********12/J1/G4 ********
 SPICE 2G.1 (150CT8Ø) *******19:47:5Ø*****

 Ø
 CMOS/SOS BITSLICE SECTION 3 PRECHARGED AT ONE NODE

 J*****
 INPUT LISTING

 TEMPERATURE =
 27.ØØØ DEG (

7

.WIDTH OUT=3Ø .OPTIONS ITL1=500 ITL5=0 MODEL NMOS MHOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 +LEVEL=1 VDD 1 σ DC 5V M1 1 3 2 σ NMOS L=5. σ U W=1 σ . σ U M2 2 5 1 1 PMOS L=5. σ U W=2 σ . σ U M3 1 7 6 1 PMOS L=5. σ U W=1 σ . σ U M4 1 9 3 1 PMOS L=5. σ U W=1 σ . σ U M4 1 9 3 1 PMOS L=5. σ U W=1 σ . σ U M5 12 11 σ σ NMOS L=5. σ U W=15. σ U M6 12 7 6 σ NMOS L=5. σ U W=15. σ U M6 12 7 6 σ NMOS L=5. σ U W=15. σ U M3 14 13 σ σ NMOS L=5. σ U W=15. σ U M3 14 13 σ σ NMOS L=5. σ U W=15. σ U M1 σ 1 13 3 1 PMOS L=5. σ U W=15. σ U M1 σ 1 13 3 1 PMOS L=5. σ U W=15. σ U M1 σ 1 13 2 1 PMOS L=5. σ U W=15. σ U M1 σ 1 13 2 1 PMOS L=5. σ U W=15. σ U M1 σ 1 15 2 σ NMOS L=5. σ U W=15. σ U M1 σ 1 15 2 σ NMOS L=5. σ U W=15. σ U M1 σ 1 15 2 σ NMOS L=5. σ U W=15. σ U M1 σ 1 15 2 σ NMOS L=5. σ U W=15. σ U M1 σ 1 15 2 σ NMOS L=5. σ U W=15. σ U M1 σ 1 15 1 PMOS L=5. σ U W=15. σ U M1 σ 1 19 15 1 PMOS L=5. σ U W=15. σ U M1 σ 1 19 15 1 PMOS L=5. σ U W=15. σ U M1 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M1 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 15 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 σ NMOS L=5. σ U W=15. σ U M2 σ 1 19 σ NMOS L=5. σ U W=10. σ U VDD 1 0 DC 5V 1 J J.1230PF C25 .TEAN 0.503 4008 .PLOT TRAN V(2) (0V.5V) .EN0



Figure C-49. SPICE Plot Section 3 Input 010 One Node Precharge with Widths x2.

C-97

F

CMOS/SOS BITSLICE SECTION 3 PRECHARGED AT ONE NODE ø INPUT LISTING TEMPERATURE = 27.000 DEG <u> (7****</u> *********** .WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 .MODEL NMOS NHOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=1Ø.ØU M2 2 5 1 1 PMOS L=5.ØU W=2Ø.ØU M3 1 7 6 1 PMOS L=5.ØU W=1Ø.ØU M1 1 3 2 Ø NMOS L=5.ØU W=1Ø.ØU M2 2 5 1 1 PMOS L=5.ØU W=2Ø.ØU M4 1 9 3 1 PMOS L=5.ØU W=1Ø.ØU M5 12 11 Ø Ø NMOS L=5.ØU W=1Ø.ØU M6 12 7 6 Ø NMOS L=5.ØU W=15.ØU M7 1 11 6 1 PMOS L=5.ØU W=15.ØU M7 1 11 6 1 PMOS L=5.ØU W=15.ØU M1 1 3 Ø Ø NMOS L=5.ØU W=15.ØU M1 1 3 0 1 PMOS L=5.ØU W=15.ØU M1 1 3 0 1 PMOS L=5.ØU W=15.ØU M1 1 3 0 1 PMOS L=5.ØU W=15.ØU M1 2 16 15 Ø Ø NMOS L=5.ØU W=15.ØU M1 2 16 15 Ø Ø NMOS L=5.ØU W=10.ØU M1 4 2 10 17 Ø HMOS L=5.ØU W=10.ØU M1 4 2 10 17 Ø HMOS L=5.ØU W=10.ØU M1 5 1 15 2 1 PHOS L=5.ØU W=10.ØU M1 6 1 9 15 1 PHOS L=5.ØU W=15.ØU M1 7 2 Ø 17 1 PMOS L=5.ØU W=15.ØU M1 8 Ø 6 13 Ø NMOS L=5.ØU W=15.ØU M2 21 10 15 Ø NMOS L=5.ØU W=15.ØU M2 21 10 15 Ø NMOS L=5.ØU W=15.ØU M2 21 10 15 Ø NMOS L=5.ØU W=15.ØU M2 1 6 10 1 FHOS L=5.ØU W=20.ØU M2 1 5 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU SEND M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU SEND M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU W=20.ØU M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS ØNS 1ØNS) M2 1 0 10 1 PHOS L=5.ØU SENS ØNS ØNS ØNS ØNS ØNS ØNS 1ØNS) M2 1 0 2 PULCE (SV ØV ØNS ØNS ØNS ØNS 1ØNS) M2 1 0 2 PULCE (SV ØV ØNS ØNS ØNS ØNS ØNS 0NS 1ØNS) M2 1 0 2 PULCE (SV ØV ØNS ØNS ØNS ØNS ØNS ØNS 0NS 0NS 0NS 0NS) M2 10 2 Ø PULCE (SV ØV ØNS ØNS ØNS ØNS ØNS 0NS 0NS) M2 10 2 Ø PULCE (SV ØV ØNS ØNS ØNS ØNS ØNS ØNS 0NS 0NS) M2 10 2 Ø PULCE (SV ØV ØNS ØNS ØNS Ø ENC

1********12/@1/34 ******* SPICE 2G.1 (150CT8Ø) *******19:47:07****

• ÷., . • .

-

•••••

C-98

•



Figure C-50. SPICE Plot Section 3 Input 101 One Node Precharge with Widths x2.

.WIDTH OUT=8Ø .OPTIONS ITL1=5ℓØ ITL5=Ø MODEL NMOS HMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PHOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=15.ØU M2 2 5 1 1 PMOS L=5.ØU W=3Ø.ØU 176 1 PMOS L=5.00 W=10.00 ΜЗ 1 7 6 1 PMOS L=5.ØU W=IØ.ØU 1 9 3 1 PMOS L=5.ØU W=1Ø.ØU 12 11 Ø Ĵ NHOS L=5.ØU W=15.ØU 12 7 6 Ĵ NHOS L=5.ØU W=15.ØU 1 11 6 1 PHOS L=5.ØU W=1Ø.ØU 14 13 Ø Ĵ NHOS L=5.ØU W=15.ØU 14 9 8 Ĵ NHOS L=5.ØU W=15.ØU Ø 1 13 3 1 PHOS L=5.ØU W=1Ø.ØU 1 1 3 2 1 PHOS L=5.ØU W=1Ø.ØU 2 16 15 Ø Ø NHOS L=5.ØU W=15.ØU M4 Μ5 МБ М7 MЗ 119 MIØ M11 1 2 2 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M12 16 15 ϑ ϑ NMOS L=5. ϑ U W=15. ϑ U M13 16 3 2 ϑ NMOS L=5. ϑ U W=15. ϑ U M14 2 13 17 ϑ NMOS L=5. ϑ U W=15. ϑ U M15 1 15 2 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M16 1 19 15 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M17 2 2 ϑ 17 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M17 2 2 ϑ 17 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M17 ϑ 5 13 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M19 21 6 ϑ MMOS L=5. ϑ U W=15. ϑ U M20 21 19 15 ϑ NMOS L=5. ϑ U W=15. ϑ U M21 1 5 13 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M22 1 5 15 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M23 ϑ 0 19 ϑ MMOS L=5. ϑ U W=1 ϑ . ϑ U M24 1 3 19 1 PMOS L=5. ϑ U W=15. ϑ U M24 1 3 19 1 PMOS L=5. ϑ U W=15. ϑ U M11 1 0 0.1230PF C25 C26 6 J Ø.316PF C27 9 ມີ Ø.232PF C28 ມີ ມີ.940PF C29 19 5 5.111PF C35 15 5 5.12JPF 13 J J.100PF 2 J J.234PF 17 J J.1PF C01 č32 C33 C33 17 σ σ .1PF C34 8 σ σ .11PF VIN1 9 σ PULCE (σ V 5V 5NS σ NS σ NS 1 σ NS) VP1 3 σ PULCE (σ V 5V 4NS σ NS σ NS 2NS) VP1BAD 5 σ PULCE (5V σ V 4NS σ NS σ NS 2NS) VCAP 11 σ PULCE (5V 5V σ NS σ NS σ NS 1 σ NS) VCAP 11 σ PULCE (σ V σ V σ NS σ NS 1 σ NS) VPH12 13 σ PULCE (σ V σ V σ NS σ NS 1 σ NS) VPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) VPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) VPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) VPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) VPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) VPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) VPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) NPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) VPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) NPH12 13 σ PULSE (σ V σ V σ NS σ NS 2 σ NS) NPH12 13 σ NS σ NS 2 σ NS 2 σ NS)

1******12/Ø1/84 ******* SPICE 2G.1 (150CT80) *******19:48:57***** CMOS/SOS BITSLICE SECTION 3 PRECHARGED AT ONE NODE

TEMPERATURE =

27.ØØØ DEG C

....

ø

INPUT LISTING



Figure C-51. SPICE Plot Section 3 Input 010 One Node Precharge with Widths x3.

******12/Ø1/84 ******** SPICE 2G.1 (150CT8Ø) *******19:48:24***** CMOS/SOS BITSLICE SECTION 3 PRECHARGED AT ONE NODE INPUT LISTING TEMPERATURE = 27.000 DEG C *********** .WIDTH OUT=80 .OPTIONS ITL1=510 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) .MODEL PMOS PMOS (VTO=-1V TOX=75NM +LEVEL=1 VDD 1 ϑ DC 5V M1 1 3 2 ϑ NMOS L=5. ϑ U W=15. ϑ U M2 2 5 1 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M3 1 7 6 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M4 1 9 3 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M5 12 11 ϑ ϑ NMOS L=5. ϑ U W=15. ϑ U M6 12 7 6 ϑ NMOS L=5. ϑ U W=15. ϑ U M7 1 11 6 1 PMOS L=5. ϑ U W=15. ϑ U M3 1 4 13 ϑ ϑ NMOS L=5. ϑ U W=15. ϑ U M1 ϑ 1 3 1 7 PMOS L=5. ϑ U W=15. ϑ U M1 ϑ 1 3 3 1 PMOS L=5. ϑ U W=15. ϑ U M1 ϑ 1 3 3 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M1 ϑ 1 3 3 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M1 ϑ 1 3 3 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M1 ϑ 1 3 3 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M1 ϑ 1 6 3 2 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M12 16 3 2 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M13 16 3 2 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M14 2 13 17 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M15 1 15 2 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M16 1 19 15 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M17 2 2 ϑ 17 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M18 ϑ G 13 ϑ NMOS L=5. ϑ U W=15. ϑ U M20 21 19 15 ϑ NMOS L=5. ϑ U W=15. ϑ U M21 1 6 13 1 PMOS L=5. ϑ U W=15. ϑ U M24 1 9 19 ϑ NMOS L=5. ϑ U W=1 ϑ . ϑ U M24 1 9 19 1 PMOS L=5. ϑ U W=1 ϑ . ϑ U M24 1 9 19 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M24 1 9 19 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M24 1 9 19 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U M24 1 9 19 1 PMOS L=5. ϑ U W=3 ϑ . ϑ U +LEVEL=1 C26 6 Ø Ø.316PF C27 9 Ø Ø.232PF C28 Ø Ø Ø.94ØPF C29 19 Ø Ø.111PF C39 15 Ø Ø.120PF C31 13 Ø Ø.1Ø3PF C32 2 Ø Ø.234PF C33 17 Ø Ø.1PF C34 8 9 0.119PF C34 8 Ø Ø.119PF VIN1 9 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (2V 5V 4NS ØNS ØNS 2NS) VP1BAR 5 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VSEL 7 Ø PULSE (5V ØV ØNS ØNS 1ØNS) VCAR 11 Ø PULSE (5V 5V ØNS ØNS 1ØNS) VPH12 18 Ø PULSE (5V 5V ØNS ØNS 2ØNS) VPH12LAR 20 Ø PULSE (ØV ØV ØNS ØNS ØNS 2ØNS) TRAH Ø.5NS 40NS .PLOT TRAM V(2) (ØV.5V) .EHD



7 7.7.7.

Figure C-52. SPICE Plot Section 3 Input 101 One Node Precharge with Widths x3.



• • •

. .

Figure C-53. Section 3 Four Node Precharge CLL Plot.



Figure C-54. Section 3 Four Node Precharge Node Plot.

C-104

<u>Table</u> <u>C-9</u>

BITSLICE PRECHARGED SECTION 3 NODE REFERENCE LIST

2

. . .

÷.,

. . .

.WIDTH OUT=8Ø .OPTIONS ITL1=5£Ø ITL5=Ø .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PHOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V 3 2 Ø NMOS L=5.ØU W=5.ØU M1 1 NMOS L=5.ØU W=5.ØU NMOS L=5.ØU W=5.ØU 3 5 Ø M2 1 MЗ 1360 1 3 7 ø NMOS L=5.ØU W=5.ØU M4 PMOS L=5.ØU W=1Ø.ØU M5 28 1 1 PMOS L=5.ØU W=10.ØU PMOS L=5.ØU W=10.ØU M6 5811 M7 6 8 1 1 6 8 1 1 PMOS L=5.00 W=10.00 7 8 1 1 PMOS L=5.00 W=10.00 1 9 6 1 PMOS L=5.00 W=10.00 1 1 10 1 PMOS L=5.00 W=10.00 14 13 0 0 NMOS L=5.00 W=15.00 14 9 6 0 NMOS L=5.00 W=15.00 1 13 6 1 PMOS L=5.00 W=10.00 7811 M8 196 M9 MIØ M11 M12 M13 15 5 Ø Ø NMOS L=5.ØU W=15.ØU 15 11 10 Ø NMOS L=5.ØU W=15.ØU M14 M15 1 PMOS L=5.00 W=10.00 1 PMOS L=5.00 W=10.00 1 5 10 M16 M17 1 10 2 17 16 2 1 PHOS L=5.00 W=10.00 17 16 2 0 NMOS L=5.00 W=15.00 17 10 2 0 NMOS L=5.00 W=15.00 2 19 13 0 NMOS L=5.00 W=5.00 1 16 2 1 PHOS L=5.00 W=10.00 M13 M19 M2.Ø M20 2 19 13 Ø NMOS L=5.00 W=5.00 M21 1 16 2 1 PMOS L=5.00 W=10.00 M22 1 7 16 1 PMOS L=5.00 W=10.00 M23 2 20 13 1 PMOS L=5.00 W=10.00 M24 0 6 5 0 NMOS L=5.00 W=15.00 M25 21 6 0 0 NMOS L=5.00 W=15.00 M26 21 7 16 0 NMOS L=5.00 W=15.00 M27 1 6 5 1 PMOS L=5.00 W=30.00 M28 1 6 16 1 PMOS L=5.00 W=10.00 M28 1 G 16 1 PMOS L=5.ØU W=1Ø.ØU M29 Ø 11 7 Ø NMOS L=5.ØU W=15.ØU МЗØ 1 11 7 1 PMOS L=5.00 W=30.00 C31 1 J Ø.1416PF C32 6 J Ø.433PF C33 7 J Ø.359PF C34 11 Ø Ø.232PF C35 Ø Ø Ø.940PF C36 16 Ø Ø.12ØPF 5 Ø Ø.266PF 2 Ø Ø.234PF C37 C33 0 Ø.234PF COD 18 0 0.1PF 10 0 0.119PF C4Ø C41 3 Ø Ø. 5ØPF VIN1 11 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VCAR 9 Ø PULSE (ØV ØY ØNS ØNS ØNS 1ØNS) VSEL 13 Ø PULSE (5V 5V ØNS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 3 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VPHI2 19 Ø PULSE (ØV 5Y 5NS ØNS ØNS 2ØNS)

*****12/Ø1/84 ******* SPICE 2G.1 (150CT8Ø) ******19:42:58**** CMOS/SOS PRECHARGED BITSLICE SECTION 3 TRANSIENT ANALYSIS

INPUT LISTING

Į

.

TEMPERATURE =

27.888 DEG C

VPHIZIAR 23 Ø PULSE (5V ØV 5NS ØNS ØNS 2ØNS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(2) (ØV,5V) .END



Figure C-55. SPICE Plot Section 3 Input 010 Four Node Precharge with Basic Widths.

C-107

•

.WIDTH OUT=3Ø .OPTIONS ITL1=540 ITL5=0 .HODEL NMOS NHOS (VTO=IV TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL = 1.MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=290 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V NMOS L=5.ØU W=5.ØU NMOS L=5.ØU W=5.ØU NMOS L=5.ØU W=5.ØU NMOS L=5.ØU W=5.ØU NMOS L=5.ØU W=5.ØU ้ 3 2 ต 3 5 ต M1 1 M2 1 36Ø 37Ø M3 1 MA 1 PMOS L=5.ØU W=1Ø.ØU PMOS L=5.ØU W=1Ø.ØU M5 2811 Μ6 58 1 1 M7 68 1 1 PMOS L=5.ØU W=1Ø.ØU M8 7 8 1 1 PMOS L=5.ØU W=10.ØU PMOS L=5.ØU W=10.ØU Μ9 196 1 1 11 1Ø 1 PMOS L=5.ØU W=1Ø.ØU 14 13 Ø Ø NMOS L=5.ØU W=15.ØU 14 9 6 Ø NMOS L=5.ØU W=15.ØU 1 13 6 1 PMOS L=5.ØU W=1Ø.ØU 1 5 5 Ø Ø NMOS L=5.ØU W=1Ø.ØU MIØ M11 M12 M13 M14 M15 15 11 10 Ø NMOS L=5.ØU W=15.ØU 1 5 1ø 1 1ø 2 1 PMOS L=5.00 W=10.00 1 PMOS L=5.00 W=10.00 M16 M17 1 10 2 1 PHOS L=5.00 W=10.00 17 16 \emptyset \emptyset NHOS L=5.00 W=15.00 17 1 \emptyset 2 \emptyset NHOS L=5.00 W=15.00 2 19 13 \emptyset NHOS L=5.00 W=5.00 1 i6 2 1 PHOS L=5.00 W=10.00 1 7 16 1 PHOS L=5.00 W=10.00 M18 M19 2 M2Ø M21 M22 M23 2 20 18 1 PMOS L=5.00 W=10.00 M23 2 20 18 1 PMOS L=5.00 W=15.00 M24 0 6 5 0 NMOS L=5.00 W=15.00 M25 21 6 0 0 NMOS L=5.00 W=15.00 M26 21 7 16 0 NMOS L=5.00 W=15.00 M27 1 6 5 1 PMOS L=5.00 W=30.00 M23 1 3 16 1 PMOS L=5.00 W=10.00 M29 Ø 11 7 Ø NMOS L=5.00 W=15.00 M3J 1 i1 7 1 Pi403 L=5.00 W=30.00 1 J J. 1416PF C31 C32 6 J J.433PF C33 7 J J.359PF 11 Ø Ø.232PF C34 C35 Ø Ø Ø.94ØPF 0 0 0.940PF 16 0 0.120PF 5 0 0.266PF 2 0 0.234PF 10 0 0.1PF 11 σ σ 119PF C36 C37 C33 C39 C39 1C σ σ σ .1PF C40 1 σ σ σ .119PF C41 3 σ σ .5 σ PF VIH1 1 0 JULCE (5V σ V 5NS σ NS σ NS 1 σ NS) VCAR 9 σ PULCE (σ V σ V σ NS σ HS σ NS 1 σ NS) VSEL 13 σ PULCE (σ V σ V σ NS σ HS σ NS 1 σ NS) VP1 3 σ PULCE (σ V 5V 4NS σ NS σ NS 2NS) VP1SAR 8 σ PULCE (σ V 5V 4NS σ NS σ NS 2NS) VP12 19 σ PULCE (σ V 5V 5NS σ HS σ NS σ SS) VPH2 19 σ PULCE (σ V 5V 5NS σ HS σ NS σ SS) VPHI2BAR 20 Ø PULSE (5V ØV 5NS ØNS ØNS 2ØNS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(2) (ØV.5V) .END

C-108



Figure C-56. SPICE Plot Section 3 Input 101 Four Node precharge with Basic Widths.

4

.WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 MODEL NMOS NHOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 MODEL PMOS PMOS (VTO=-1V TOX=75NM UO≈200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC M1 1 3 2 Ø M2 1 3 5 Ø 5V NMOS L=5.0U W=10.0U NMOS L=5.0U W=10.0U M2 1 3 5 Ø NMOS L=5.ØU W=1Ø.ØU M3 1 3 6 Ø NMOS L=5.ØU W=1Ø.ØU M4 1 3 7 Ø NMOS L=5.ØU W=1Ø.ØU M5 2 8 1 1 PMOS L=5.ØU W=2Ø.ØU M7 6 8 1 1 PMOS L=5.ØU W=2Ø.ØU M7 6 8 1 1 PMOS L=5.ØU W=2Ø.ØU M8 7 8 1 1 PMOS L=5.ØU W=2Ø.ØU M9 1 9 6 1 PMOS L=5.ØU W=2Ø.ØU M10 1 11 1Ø 1 PMOS L=5.ØU W=1Ø.ØU M12 14 13 ØØ NMOS L=5.ØU W=15.ØU M13 1 13 6 1 PMOS L=5.ØU W=15.ØU M14 15 5 ØØ NMOS L=5.ØU W=15.ØU M15 15 11 1ØØ NMOS L=5.ØU W=15.ØU M16 1 5 1Ø 1 PMOS L=5.ØU W=15.ØU M17 1 IØ 2 1 PMOS L=5.ØU W=15.ØUM17 1 IØ 2 1 PMOS L=5.ØU W=15.ØUM19 17 <math>IØ 2 0 NMOS L=5.ØU W=15.ØUM19 17 IØ 2 0 NMOS L=5.ØU W=15.ØUM22 2 19 13 Ø NMOS L=5.ØU W=15.ØU M22 1 7 16 1 PMOS L=5.ØU W=15.ØU M22 1 7 16 1 PMOS L=5.ØU W=10.ØU M22 1 7 16 1 PMOS L=5.ØU W=10.ØU M22 1 7 16 1 PMOS L=5.ØU W=15.ØU M22 1 7 16 1 PMOS L=5.ØU W=15.ØU M23 2 2Ø 13 1 PMOS L=5.ØU W=15.ØU M24 Ø 6 5 Ø NMOS L=5.ØU W=15.ØU M25 21 5 ØØ NMOS L=5.ØU W=15.ØU M26 21 7 15 Ø NMOS L=5.ØU W=15.ØU M26 21 7 15 Ø NMOS L=5.ØU W=15.ØU M26 21 7 15 Ø NMOS L=5.ØU W=15.ØU M27 1 6 5 1 PMOS L=5.ØU W=15.ØU M26 21 7 15 Ø NMOS L=5.ØU W=15.ØU M27 1 6 5 1 PMOS L=5.ØU W=15.ØU M28 1 6 16 1 PMOS L=5.ØU W=10.ØU M23 Ø 16 1 PMOS L=5.ØU W=10.ØU M24 Ø 6 5.ØU W=10.ØU M25 Ø 16 1 PMOS L=5.ØU W=10.ØU M26 0 1 7 0 M40S L=5.ØU W=10.ØU M27 0 0 11 7 0 M40S L=5.0U W=10.ØU M28 1 6 16 1 PMOS L=5.0U W=10.ØU M29 Ø 11 7 Ø M40S L=5.0U W=10.ØU M20 Ø 11 7 Ø M40S L=5.0U W=10.ØU M20 Ø 11 7 Ø M40S L=5.0U W=10.ØU M20 Ø 11 7 Ø M40S L=5.0U W=10.ØU M20 Ø 11 7 Ø M40S L=5.0U W=10.ØU M20 Ø 11 7 Ø M40S L=5.0U W=10.ØU M20 Ø 11 7 Ø M40S L=5.0U W=10.0U NMOS L=5.00 W=10.00 36 M27 1 G 5 1 PMOS L=5.00 W=15.0 M28 1 5 1G 1 PMOS L=5.0U W=30.0U M29 0 11 7 0 NMOS L=5.0U W=18.0U M30 1 11 7 1 PHOS L=5.0U W=30.0U C31 1 J J.1416PF 6 มี มี.433PF 7 ป มี.359PF C32 C33 C34 11 Ø Ø.232PF C35 Ø Ø Ø.94ØPF C36 16 Ø Ø.129PF C37 5 Ø Ø.266PF C33 2 J J.234PF 18 Ø Ø.1PF 15 J J.119PF C39 c ;g C41 3 J Ø. 5JPF C41 3 # 0.50PF VIN1 11 0 PULCE (ØV 5V 5NS ØNS ØNS 1ØNS) VCAR 9 0 PULCE (ØV ØV ØNS ØNS ØNS 1ØNS) VSEL 13 0 PULCE (5V 5V ØNS ØNS ØNS 1ØNS) VP1 3 # PULCE (ØV 5V 4NS ØNS ØNS 2NS) VP15AR 3 # PULCE (ØV 5V 4NS ØNS ØNS 2NS) VP112 19 0 PULCE (ØV 5V 5NS ØNS ØNS 20S) VPHI2BAR 20 0 PULSE (5V ØV 5NS ØNS ØNS 20NS) .TRAN U.5NS 40NS .PLOT TRAN V(2) (ØV.5V) . END



- -

4

1

Figure C-57. SPICE Plot Section 3 Input 010 Four Node Precharge with Widths x2.

CMOS/SOS PRECHARGED BITSLICE SECTION 3 TRANSIENT ANALYSIS INPUT LISTING TEMPERATURE = 27.000 DEG C ----.WIDTH OUT=8Ø .OPTIONS ITL1=540 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL=1 .MODEL PMOS PHOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL≠1 VDD 1 Ø DC 5V ้ 3 2 ้ Ø 3 5 Ø M1 1 NMOS L=5.ØU W=1Ø.ØU M2 ī NMOS L=5.00 W=10.00 NMOS L=5.ØU W=10.ØU М3 36Ø 1 NMOS L=5.ØU W=1Ø.ØU 37Ø MΔ 1 PMOS L=5.ØU W=2Ø.ØU Μ5 2811 PMOS L=5.ØU W=20.ØU PMOS L=5.ØU W=20.ØU 5811 Μ6 MŻ 6811 7 8 1 1 PMOS L=5.00 W=20.00 MR M14 15 5 \emptyset \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M15 15 11 1 \emptyset \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M16 1 5 1 \emptyset 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M17 1 1 \emptyset 2 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M18 17 16 \emptyset \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M19 17 1 \emptyset 2 \emptyset NMOS L=5. \emptyset U W=15. \emptyset U M2 \emptyset 2 19 18 \emptyset NMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M21 1 16 2 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M22 1 7 16 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M23 2 2 \emptyset 18 1 PMOS L=5. \emptyset U W=1 \emptyset . \emptyset U M24 \emptyset 6 5 \emptyset NMOS L=5. \emptyset U W=1 \emptyset . \emptyset U C31 1 Ø Ø.1416PF 6 0 Ø.433PF 7 Ø Ø.359PF C32 C33 C34 11 Ø Ø.232PF C35 Ø Ø Ø.940PF 16 0 Ø.120PF C36 C37 5 Ø Ø.266PF C38 2 Ø Ø.234PF C39 18 Ø Ø.1PF C 4 Ø 10 J Ø.119PF C40 10 J 0.119PF C41 3 σ J.50PF VIN1 1 σ PULSE (5V σ V 5NS σ NS σ NS 1 σ NS) VCAR 9 σ PULSE (σ V σ V σ NS σ HS σ NS 1 σ NS) VSEL 13 σ PULSE (σ V σ V σ NS σ HS σ NS 1 σ NS) VPI 3 σ PULSE (σ V 5V 4NS σ NS σ NS 2NS) VP1BAR 3 σ PULSE (σ V σ V 4NS σ NS σ NS 2NS) VP12 19 σ PULSE (σ V 5V 5NS σ HS σ NS σ SNS) VPHI2 19 σ PULSE (σ V 5V 5NS σ HS σ NS σ SNS) VPHI2BAR 20 0 PULSE (5V 0V 5NS 0NS 0NS 20NS) .TRAN 0.5NS 40NS .PLOT TRAN V(2) (0V,5V) .END

1******12/31/84 ******* SPICF 2G.1 (150CT80) *******19:44:18*****

Į.

۰.

.

Ċ

~____

CMOS/SOS PRECHARGED BITSLICE SECTION 3 TRANSIENT ANALYSIS X2 . TEMPERATURE = 27.888 DEG C *a* • TRANSIENT ANALYSIS TIME V(Z) d+## **s**. 1.258d+## 2.5##d+## 3.75#d+## 5.###d+## 8#7d-#7 .8£7d-87 .8£7d-87 .8£7d-87 .8£7d-87

12/#1/84 ****** SPICE 2G.1 (150CT8#) *******19:44:18

1**

•

Figure C-58. SPICE Plot Section 3 Input 101 Four Node Precharge with Widths x2. The second second second second second second second second second second second second second second second s

.WIDTH OUT=8Ø .OPTIONS ITL1=500 ITL5=0 MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL= .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) ⊦LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø M2 1 3 5 Ø NMOS L=5.ØU W=15.ØU NMOS L=5.ØU W=15.ØU 1 3 6 Ø NMOS L=5.ØU W=15.ØU MЗ MA 1 3 7 ø NMOS L=5.00 W=15.00 M5 28 1 1 PMOS L=5.00 W=30.00 PMOS L=5.00 W=30.00 PMOS L=5.00 W=30.00 М6 581 1 M7 6 8 1 1 M8 7 8 1 1 PMOS L=5.80 W=38.80 M9 1 9 6 1 PMOS L=5.80 W=18.80 M18 1 11 18 1 PMOS L=5.80 W=18.80 M18 1 11 18 1 PMOS L=5.80 W=18.80 M11 14 13 8 8 NMOS L=5.80 W=15.80 M11 14 13 Ø Ø NMOS L=5.ØU W=15.ØU M12 14 9 6 Ø NMOS L=5.ØU W=15.ØU M13 1 13 6 1 PMOS L=5.ØU W=10.ØU M14 15 5 Ø Ø NMOS L=5.ØU W=15.ØU M15 15 11 1Ø Ø NMOS L=5.ØU W=15.ØU M16 1 5 1Ø 1 PMOS L=5.ØU W=10.ØU M17 1 1Ø 2 1 PMOS L=5.ØU W=10.ØU 1 18 2 1 PMOS L=5.80 W=18.80 17 16 8 8 NMOS L=5.80 W=15.80 17 18 2 8 NMOS L=5.80 W=15.80 M18 M19 2 19 18 Ø NMOS L=5.ØU W=1Ø.ØU 1 16 2 1 PMOS L=5.ØU W=1Ø.ØU 1 7 16 1 PMOS L=5.ØU W=1Ø.ØU M2Ø M21 1 16 M22 M23 2 28 18 1 PMOS L=5.80 W=18. M24 8 6 5 8 NMOS L=5.80 W=15.80 L=5.ØU W=1Ø.ØU 21 6 8 8 MMOS L=5.80 W=15.80 21 7 16 8 MMOS L=5.80 W=15.80 1 6 5 1 PMOS L=5.80 W=38.80 1 6 5 1 PMOS L=5.80 W=38.80 M25 M26 M27 1 6 5 1 PMOS L=5.00 W=30.00 M28 1 6 16 1 PMOS L=5.00 W=10.00 M29 0 11 7 0 NMOS L=5.00 W=15.00 M30 1 11 7 1 PMOS L=5.00 W=30.00 C31 1 Ø Ø.1416PF 6 Ø Ø.433PF 7 Ø Ø.359PF 11 Ø Ø.232PF Ø Ø Ø.94ØPF C32 C33 C34 Č35 16 Ø Ø.12ØPF 5 Ø Ø.266PF C36 C37 5 Ø Ø.266PF C38 2 Ø Ø.234PF C39 18 Ø Ø.1PF C40 10 Ø Ø.119PF C41 3 Ø Ø.50PF VIN1 11 Ø PULSE (ØV 5V 5NS ØNS ØNS 1ØNS) VCAR 9 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VSEL 13 Ø PULSE (ØV ØV ØNS ØNS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 8 Ø PULSE (5V ØV 4NS ØNS ØNS 2NS) VP12 19 Ø PULSE (ØV 5V 5NS ØNS ØNS 2NS) C37 VPHI2BAR 20 0 PULSE (5V 0V SNS 0NS 0NS 20NS) .TRAN 0.5NS 40NS .PLOT TRAN V(2) (0V,5V) . END

*****12/81/84 ******** SPICE 2G.1 (150CT88) ********19:44:55 CHOS/SOS PRECHARGED BITSLICE SECTION 3 TRANSIENT ANALYSIS TRANSIENT ANALYSIS TEMPERATURE = 27.888 DEG C TIME V(2) . 1.258d+## 2.5##d+## 3.758d+## 5.###d+## #. d+## 5. d+## 5. d+## 5. d+## 5. d+## 5. d+## 5. d+## 5. d+# 5. d

· .

ľ

Figure C-59. SPICE Plot Section 3 Input 010 Four Node Precharge with Widths x3.

CMOS/SOS PRECHARGED BITSLICE SECTION 3 TRANSIENT ANALYSIS Ø INPUT LISTING TEMPERATURE = 27.000 DEG C .WIDTH OUT=8Ø .OPTIONS ITL1=500 ITL5=0 .MODEL NMOS NMOS (VTO=1V TOX=75NM UO=400 NSUB=2.5E16 LD=0.7UM) +LEVEL±1 .MODEL PMOS PMOS (VTO=-1V TOX=75NM UO=200 NSUB=3E15 LD=0.7UM) +LEVEL=1 VDD 1 Ø DC 5V M1 1 3 2 Ø NMOS L=5.ØU W=15.ØU 3 5 Ø NMOS L=5.0U W=15.0U M2 1 NMOS L=5.ØU W=15.ØU NMOS L=5.ØU W=15.ØU M3 1 3 6 Ø 1 3 7 Ø M4 C31 1 Ø Ø.1416PF C32 6 Ø Ø.433PF C33 7 Ø Ø.359PF 11 Ø Ø.232PF C34 C35 Ø Ø Ø.94ØPF 16 Ø Ø.12ØPF C36 C37 5 Ø Ø.266PF C38 2 Ø Ø.234PF C39 18 Ø Ø.1PF C4Ø 10 Ø Ø.119PF C41 3 Ø Ø.5ØPF VIN1 11 Ø PULSE (5V ØV 5NS ØNS ØNS 1ØNS) VCAR 9 Ø PULSE (ØV ØV ØNS ØHS ØNS 1ØNS) VSEL 13 Ø PULSE (ØV ØV ØNS ØHS ØNS 1ØNS) VP1 3 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VP1BAR 8 Ø PULSE (ØV 5V 4NS ØNS ØNS 2NS) VPHI2 19 Ø PULSE (ØV 5V 5NS ØNS ØNS 2ØNS) VPHI2BAR 2Ø Ø PULSE (5V ØV 5NS ØNS ØNS 2ØNS) VFHI2BAR 2Ø Ø PULSE (5V ØV 5NS ØNS ØNS 2ØNS) .TRAN Ø.5NS 4ØNS .PLOT TRAN V(2) (ØV,5V) . FND

C-116

.

.

• •



.....

-

Figure C-60. SPICE Plot Section 3 Input 101 Four Node Precharge with Widths x3.

C-117

Appendix D

This appendix is intended to serve as a practical guide for the individual who wishes to use the SPICE2 simulators available at AFIT. Much time was wasted in trying to determine how to correctly use the simulator due to lack of documentation and knowledgeable people. Much of the data was so spread throughout the users guide that it took a great amount of time to get comfortable with what it all meant. A procedure for establishing input files is presented in this appendix to reduce the confusion.

This appendix refers only to transient analysis use of the SPICE simulator.

The following information outlines the barriers encountered and overcome by this author to perform the simulations required for this thesis project. Hopefully this will benefit future users of SPICE.

The reader is expected to be familiar with the switch level simulator ESIM, since some similiar procedures are followed and the reader should refer to the SPICE users guide by Vladimirescu et al provided as reference material in the EE695 VLSI Notebook provided as a class text.

SPICE was originally intended for use with batch processing, so all the documentation makes "card" references. This is easily translated to lines of instructions in input files for an interactive system such as the VAX 11/780.

To run a SPICE simulation on a circuit, the following must be done:

1. Design a circuit using CLL or some other layout language.

2. Run cll -C filename.cll to obtain the .cif file.

3. Change all 125/1 to 125 1 in the .cif file. (this allows you to run mextra)

4. Run mextra on the .cif file.

(This results in a .nodes and a .sim file)

5. Change the first line of the .sim file from tech=nmos to tech=cmos.

6. Generate a definitions file for NMOS and PMOS substrate connection nodes. This is required for sim2spice, the SPICE preprocessor which converts .sim format to .spice format. In CMOS, the depletion mode device is the PMOS device and the pull-down device is the NMOS device. Sim2spice defaults to plain NMOS technology with ENMOS and DNMOS descriptions. These must be changed in the definitions file. This file can also establish the ground and V_{dd} nodes for substrate connections. The PMOS substrate is connected to V_{dd} and the NMOS substrate is connected to ground. The file would look like this: set GND 0 cmos set Vdd 1 cmos set PMOS 1 cmos set NMOS 0 cmos def PMOS DNMOS cmos def NMOS ENMOS cmos

I tried changing the .cadrc file, as indicated in the sim2spice literature, but it didn't work.

7. Run sim2spice -d definitions filename filename.sim This results in a filename.spice file and a filename.names file. The .names file is a cross reference list between mextra and SPICE node names. The SPICE nodes are sequentially numbered and the mextra nodes appear at random. The SPICE filename.spice input file must refer to only SPICE nodes, not mextra nodes since the SPICE simulator will only recognize its sequential numbering of the respective nodes prepared by sim2spice.

8. The filename.spice is the input file required to run the simulation, but it does not include the command "cards" required for program control. It just contains the MOS, four node descriptions of each device in the circuit to be simulated. There should be an NMOS and PMOS device for each CMOS device. For small circuits these can be counted to verify that sim2spice has run correctly. Larger circuits are hard to verify since they contain so many devices.

This file includes all (four) the node connections for an individual NMOS or PMOS device.

However, all the nodes are SPICE nodes not mextra

nodes.

9. The next step is to modify the filename.spice file to account for the differences in the assignment of nodes between mextra and SPICE. A numbered node cifplot of the filename.cif and the filename.nodes files is required, and a copy of the filename.names file that was generated by sim2spice.

The ground node must be determined from the mextra based node plot. SPICE requires that the ground node be 0. Mextra does not use 0 as a node number. Therefore the mextra node for ground will be a number other thn 0 and in the mextra/SPICE node reference list a number other than 0 will have been assigned to the ground node by SPICE. This node must be determined and chnaged to 0 everywhere it occurs in the filename.spice file. For V_{dd} , the same procedure must be followed, but the node number is optional. However, it must be consistent with the node number assigned to the V_{dd} node when the independent DC supply voltage source is entered as a command line (card).

For all other inputs to the circuits, the input node must be selected on the mextra node cifplot, cross referenced to its respective SPICE node number and that SPICE number used in the filename.spice file when a model input voltage is applied to it for simulation.

10. The next step is to input all the required/desired command (cards) lines to run the

simulation using the node numbers from SPICE node reference list but based on the nodes assigned by mextra on the node cifplot. The required command line (card) inputs and the format rules required by SPICE are presented below.

The following command line (card) inputs must be included in the filename.spice file and the associated rules presented must be adhered to:

a. All command lines (cards) must begin with a period.

b. The first line of the input file must be a title:

i.e. CMOS/SOS TRANSIENT ANALYSIS

This will be printed on each page of the results output. A "dot" command on the first line will cause the run to abort, since the simulator will think it is the title and skip it as a desird command.

c. The .WIDTH command line is used next to establish the last column read on each line. It is normally established at 80.

d. The .OPTIONS command line is then used to establish program control as desired. The documentation on pages 20-21 of the original users guide is adequate for explaining the options available.

e. The user is allowed to select models of the devices that are to be simulated using the .MODEL command line. Since MOS models were used for this thesis, three possible level models were available, (see page 16 in users guide). The level 1 model was selected.

A diode, two bipolar and MOSFET and JFET models are available (see pages 13-18 in the users guide). The user can select specific parameters per device model. Since CMOS/SOS was used for this thesis, two MOSFET models were used, PMOS and NMOS.

f. After selecting models, the V_{dd} supply voltage must be established. Voltage sources are established by using ELEMENT command lines (cards). Resistors, capacitors, inductors, transmission lines can also be set-up using this ELEMENT command line model. An independent 5 volt DC voltage source was established (see pages 8-9 in users guide) as the supply voltage. The node to which the voltage is applied is left up to the users discretion.

g. The sim2spice file data is input after the supply voltage command line is established.

h. Input voltages to the circuit are modeled the same way. However, the PULSE option was used in this thesis to provide a varying pulsed input for transient delay analysis. This is described on page 9 of the users guide.

i. The .TRAN command line is used to establish the

time interval of the simulation, (40 nsec was used for all simulations in this thesis) and the points at which the transient waveform is traced on the output plot (0.5 nsec intervals were used for all simulations in this thesis. See page 25 of the users gyuide for further details.

alian kaning terdina terdina terdina dari berdari dalam berakan kaning meneralah sebelah sebelah sebelah sebela

j. The .PLOT comand line estbalishes that a waveform output plot is desired. The TRAN option of this command indicates that a transient plot is desired. The nodes to be monitored and their respective values plotted over the selected simulation time interval are selectable by the user.

k. The input file must be concluded with a .END command line.

11. The only other modification that may be required is a modification of the capacitance values generated by sim2spice when the filename.spice file is generated form the filename.sim file. The values are given in femtofarads and these sometimes are so small that plotted results are worhtless. Some factor of ten may be required to achieve good output results.

Some of these procedures are tedious, however no shorter, more convenient way could be determined to prepare for and perfrom the simulations using the tools that were

available at AFIT.

Example input files and simulation data are provided in previous appendices for reference.

Hopefully this user guide addendum will aid future users of SPICE2.

Appendix E

This appendix is intended to provide information on the Memory/ALU Bitslice circuit chip that was designed as part of this thesis.

The chip contains the 2x4 precharged decoder, the precharged A and B registers, the ALU input controller, the read/write controller, the 4x4 static RAM and the ALU first bitslice along with input, output, ground and V_{dd} pads.

The chip files are included under bkelley/thesis in this authors directory. The directory listing is presented below:

> README chip.cll ramcell.cll 4bitword.cll 4wordmem.cll Aregpre.cll Bregpre.cll alucntrl.cll RWcontrol.cll bslast.cll - bitslice precharged at 4 nodes decpre4a.cll padamp.cif padVdd.cif padblank.cif padgrnd.cif padout.cif padin.cif paddrve.cif

> > E-1

AD-R152 466	SPEED-UP TECHNIQUES FOR COMPLEMENTARY METAL OXIDE SEMICONDUCTOR VERY LARG. (U) AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGI. B T KELLEY 14 DEC 84 AFIT/GE/EMG/84D-41 F/G 20/12	5/ 5
		NL
1		
1		
i		



MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A

Brian T. Kelley was born on 6 June 1957 in Framingham, Massachusetts. He graduated from Medway Junior-Senior High School in 1975 and attended Worcestor Polytechnic Institute in Worcestor, Massachusetts, where he received the degree of Bachelor of Science in Electrical Engineering in October Upon graduation, he was commissioned a scond 1979. lieutenant in the USAF through the ROTC program. From June 1979 through May 1983 he was stationed at McClellan AFB, California where he served as a Communications Systems Engineer for the Satellite, Surveillance and Electronic Warefare Reliability Engineering branch of the Material Management Acquisition Division of Sacramento Air Logistics Center. In May 1983, he entered the School of Engineering, Air Force Institute of Technology to pursue a graduate degree in digital engineering and Very Large Scale Integration (VLSI) electronics design.

Permanent Address: 240 Main Street

Medway, Massachusetts 02053

Unclassified

F

SECURITY CLASSIFICATION OF THIS PAGE

			REPORT DUCUM	ENTATION PAGE	:		
Unclas	SECURITY CL	ASSIFICATION		16. RESTRICTIVE M	ARKINGS		
28. SECURITY CLASSIFICATION AUTHORITY				3. DISTRIBUTION/AVAILABILITY OF REPORT			
20. DECLASSIFICATION/DOWNGRADING SCHEDULE				Approved for public release; distribution unlimited.			
A PERFORM	E/ENG/84D	ZATION REPORT NU 1-41	MBER(S)	5. MONITORING OR	GANIZATION R	EPORT NUMBER(S)
6. NAME OF PERFORMING ORGANIZATION Bb. OFFICE SYMBOL School of Engineering (If applicable)			7. NAME OF MONIT	ORING ORGAN	ZATION		
			AFIT/ENG				
Air For	S(City, State a) rce Insti	nd ZIP Code) tute of Techn	010gy	76. ADDRESS (City, S	State and 7.1P Cod	ie)	•
Wright	-Patterso	n Air Force I	lase, OH 45433				
B. NAME OF FUNDING/SPONSORING ORGANIZATION (1/ applicable)				9. PROCUREMENT I	NSTRUMENT ID	ENTIFICATION N	IUMBER
	S (City, State a	nd ZIP Code)		10. SOURCE OF FUN	DING NOS.		
				PROGRAM	PROJECT	TASK	WORK UNIT
				ELEMENT NO.	NU.	NU.	NO.
See Blo	iclude Security Ock 19	Classification)		1			
PERSONA Brian	AL AUTHORIS	i) Cant USAF				l	; ;
IJA. TYPE OF	F REPORT	136. TIME	COVERED	14. DATE OF REPOR	IT (Yr , Mo., Day)	15. PAGE (COUNT
MS The	sis	FROM_	to	14 Dec 1984 386			
Prepare Science	ed in Par e	tial Fulfillm	ent of the Requi	rements for the	e Degree of	f Master of	<u>-</u>
Prepare Science	ed in Par e <u>cosatic</u> group	codes sub. GR.	ent of the Requi	rements for the	e Degree o:	f Master of	
Prepare Science 7 FIELD 09	ed in Par e cosatic group 01	tial Fulfilln ODES SUB. GR.	ent of the Requi	continue on reverse if ne Devices, Swite	e Degree of centery of a identiti ching Logic	f Master of 17 by block number c, Speed-Up	rr, Techn Lque
Prepare Science 7 FIELD 09	ed in Par e COSATIC GROUP 01 CT (Continue of	DOES SUB. GR.	ent of the Requi	continue on reverse if ne Devices, Swite	e Degree of centery of a identit ching Logic	f Master of //y by block number c, Speed-Up	ir, Techn ‡que
Preparo Scienco FIELO 09 Title: Thesis	ed in Par e COSATIC GROUP 01 CT (Continue d : Speed- Large s Advisor	Up Techniques Scale Integra	ent of the Requi	rements for the Continue on reverse if ne Devices, Swite er) ry Metal Oxide	e Degree of centery of a identit ching Logic Semiconduc	f Master of //y by block number c, Speed-Up ctor Very	r) Techn įque ,
Prepare Science FIELD 09 Title: Thesis	ed in Par e COSATIC GROUP 01 CT (Continue o : Speed- Large s Advisor	tial Fulfilln SODES SUB. GR. Up Techniques Scale Integra : Major W. S	ent of the Requi	rements for the Continue on reverse if ne Devices, Switt er) ry Metal Oxide	e Degree of centry of a identic ching Logic Semiconduc	f Master of // by block number c, Speed-Up ctor Very	Techn įquer
Prepare Science 17. FIELD 09 19. ABSTRAC Title: Thesis	ed in Par e COSATIC GROUP 01 CT (Continue of CT (Continue of CT (Continue of CT (Continue of CT (CT (CT (CT (CT (CT (CT (CT (CT (CT	tial Fulfillm SODES SUB. GR. Up Techniques Scale Integra : Major W. S	ent of the Requi	rements for the Continue on reverse if ne Devices, Swite er) ry Metal Oxide	e Degree of centery of a (denti- ching Logic Semiconduc Semiconduc	f Master of Ity by block number c, Speed-Up ctor Very release 1AW AF	Techn Lque R 190 De
Prepare Science 17. FIELD 09 19 ABSTRAC Title: Thesis	ed in Par e COSATIC GROUP 01 CT (Continue of CT (Continue of Large s Advisor	tial Fulfillm SUB. GR. Up Techniques Scale Integra : Major W. S	ent of the Requi	rements for the Continue on reverse if ne Devices, Swite er) ry Metal Oxide	e Degree of centry of a identic ching Logic Semiconduc Semiconduc	f Master of fy by block number c, Speed-Up ctor Very release IAW AF Laboration D charactery IAU Laboration D	Techn Lque
Prepare Science 17. FIELD 09 19. ABSTRAC Title: Thesis	ed in Par e COSATIC GROUP 01 CT (Continue of : Speed- Large s Advisor	tial Fulfillm SODES SUB. GR. Up Techniques Scale Integra Major W. S	ent of the Requi	rements for the Continue on reverse if ne Devices, Swite er) ry Metal Oxide	e Degree of censory of a identific ching Logic Semiconduc Semiconduc Semiconduc Semiconduc Semiconduc Semiconduc Semiconduc Semiconduc Semiconduc Semiconduc	f Master of ly by block number c, Speed-Up ctor Very release IAW AF Le taxanal D Labourgy IALC about 4587 bile releaser IAW VER Alfel ch and Professional (J Alfel ch and Professional (J)	AF: 190-17.
Prepare Science 17. FIELD 09 19. ABSTRAC Title: Thesis	ed in Par e COSATIC GROUP 01 CT (Continue d : Speed- Large s Advisor	tial Fulfillm SUB. GR. SUB. GR. Up Techniques Scale Integra : Major W. S ABILITY OF ABSTR	ACT	continue on reverse i/ ne Devices, Swite r) ry Metal Oxide	e Degree of censory of a identific ching Logic Semiconduc Semiconduc sproved for Fublic rain for Seconduc for Fublic for for Seconduc for Fublic for Fubli	f Master of (y by block number c, Speed-Up ctor Very release 1AW AF La taxanal D connectory 1AW release 1AW to taxanal D connectory 1AW release 1AW AFB Cit 442 CATION	Techn Lquer R 190-IZe, AF: 190-IZ. Development TC:
Prepare Science 17. FIELD 09 19. ABSTRAC Title: Thesis	ed in Par e COSATIC GROUP 01 CT (Continue d : Speed- Large s Advisor	tial Fulfillm SUB. GR. SUB. GR. Dy Techniques Scale Integra : Major W. S ABILITY OF ABSTR ED 23 SAME AS RP	ACT	rements for the Continue on reverse if ne Devices, Swite er) ry Metal Oxide	e Degree of centery of a identific ching Logic Semiconduc Semiconduc sproved for public of for the F Kongon for public for the	f Master of If y by block number c, Speed-Up ctor Very release IAW AF is take-and D is tak	Techn i quer
Prepare Science 17. FIELD 09 19. ABSTRAC Title: Thesis Thesis 20. NAME 0	ed in Par e COSATIC GROUP 01 CT (Continue o : Speed- Large s Advisor ution/Avail ed/UNLIMITI	tial Fulfillm SUB. GR. SUB. GR. Up Techniques Scale Integra : Major W. S ABILITY OF ABSTR ED SAME AS RP BLE INDIVIDUAL	ACT	21. ABSTRACT SECU Unclassific 22b. TELEPHONE NU	e Degree of censory of a (denti- ching Logic Semiconduc Semiconduc sproved for public ran for the T Superior for public for for the T Superior for public for for the Superior for public set for the Superior for the Supe	f Master of Ity by block number c, Speed-Up ctor Very relater IAW AF Laborandi D Consology (AAC AD ADA Second Second Laborandi D Consology (AAC AD ADA Second Second AD ADA AD A	Techn Lquer R 190-16, AF: 190-17, Development NC:
Prepare Science 17. FIELD 09 19. ABSTRAC Title: Thesis Thesis SwcLassifi 22. NAME O Major	ed in Par e COSATIC GROUP 01 CT (Continue of Speed- Large S Advisor UTION/AVAIL ED/UNLIMITIC F RESPONSIE W. Sutto	tial Fulfillm SUB. GR. SUB. GR. Up Techniques Scale Integra : Major W. S ABILITY OF ABSTR ED S SAME AS RP BLE INDIVIDUAL n, USAF	ACT	rements for the Continue on reverse if ne Devices, Swite rry Metal Oxide 21. ABSTRACT SECL Unclassifie 22b. TELEPHONE MI (Include Area Co 513-255-691.	e Degree of censory of a identiti ching Logic Semiconduc Semiconduc sproved for Fublic ran for Seconduc for F second for Seconduc for F second for Seconduc for F second for Seconduc for F second for Seconduc for F seconduc for Seconduc for	f Master of (y by block number c, Speed-Up ctor Very retriee IAW AF Listenand D connoisey IAU to taxenand D connoisey IAU to taxenand D connoisey IAU to taxenand D connoisey IAU to taxenand D connoisey IAU 22c. Office SyM AFIT/EN	Techn Lque R 190-IZ, AF. 190-IZ, Development AC.

_ - -1 • . ۰. : ·. ·. ÷

- 1
Unclassified

6

ĺ

SECURITY CLASSIFICATION OF THIS PAGE

Methods of increasing the operating speed of integrated circuits were investigated and a reference to speed-up techniques was generated. Precharging, a specific technique, was applied to existing and newly designed CMOS/SOS circuit elements and evaluated. SPICE was used for transient signal analysis.

Precharging was applied to a test circuit, the first bitslice of a previously designed ALU circuit and three newly designed chip circuit elements, to determine its effect on the operating speed of each circuit. Precharge configurations of each circuit were then simulated with SPICE.

The results of this thesis research indicate that precharging can be applied to both existing and newly designed circuits, that it significantly reduces low-to-high signaltransition delays, if applied correctly, and in general, it has a detrimental effect on high-to-low signal transitions, and increases the associated delays. In addition, the effectiveness of precharging is dependent on the amount of current applied to the precharged nodes.

SECURITY OLASSIFICATION OF THIS PAGE



END

5-85

DTIC